UNITED STATES SECURITIES AND EXCHANGE COMMISSION

WASHINGTON, D.C. 20549

FORM 8-K/A

(Amendment No. 1)

CURRENT REPORT

Pursuant to Section 13 or 15(d) of the Securities Exchange Act of 1934

Date of Report (Date of earliest event reported): August 15, 2023

Precision BioSciences, Inc.

(Exact name of Registrant as Specified in Its Charter)

Delaware (State or Other Jurisdiction of Incorporation) 001-38841 (Commission File Number) 20-4206017 (IRS Employer Identification No.)

302 East Pettigrew St.
Suite A-100
Durham, North Carolina
(Address of Principal Executive Offices)

27701 (Zip Code)

Registrant's Telephone Number, Including Area Code: 919 314-5512

	(Former Name or Former Address, if Changed Since Last Report)						
	Check the appropriate box below if the Form 8-K filing is intended to simultaneously satisfy the filing obligation of the registrant under any of the following provisions:						
	Written communications pursuant to Rule 425 under the Securities Act (17 CFR 230.425)						
	Soliciting material pursuant to Rule 14a-12 under the Ex	change Act (17 CFR 24	0.14a-12)				
	Pre-commencement communications pursuant to Rule 14d-2(b) under the Exchange Act (17 CFR 240.14d-2(b))						
	Pre-commencement communications pursuant to Rule 13e-4(c) under the Exchange Act (17 CFR 240.13e-4(c))						
Securities registered pursuant to Section 12(b) of the Act:							
		Trading					
	Title of each class	Symbol(s)	Name of each exchange on which registered				
	Common Stock, par value \$0.000005 per share	DTIL	The Nasdaq Global Select Market				
	cate by check mark whether the registrant is an emerging poter) or Rule 12b-2 of the Securities Exchange Act of 1934	0 1 5	ined in Rule 405 of the Securities Act of 1933 (§ 230.405 of this napter).				

Emerging growth company ⊠

If an emerging growth company, indicate by check mark if the registrant has elected not to use the extended transition period for complying with any new or revised financial accounting standards provided pursuant to Section 13(a) of the Exchange Act. \Box

Explanatory Note

On August 15, 2023, Precision BioSciences, Inc. (the "Company") filed a Current Report on Form 8-K (the "Original Form 8-K") to report that (i) the Company entered into an Asset Purchase Agreement (the "Purchase Agreement") with Imugene Limited, an Australian corporation, and its wholly owned subsidiary Imugene (USA) Inc. ("Imugene US"), a Nevada corporation, pursuant to which Imugene US acquired the Company's manufacturing infrastructure used in the development and manufacture of azercabtagene zapreleucel ("azer-cel"), including the lease to the Company's manufacturing facility and certain contracts of the Company with respect to the Company's manufacturing facility, and related equipment, supplies, azer-cel clinical trial inventory and other assets related to the Company's CAR T cell therapy platform (the "Acquisition"); and (ii) in connection with the Purchase Agreement, the Company and Imugene US entered into a License Agreement (the "License Agreement"), pursuant to which the Company granted Imugene US certain exclusive and non-exclusive license rights to develop, manufacture, and commercialize oncological applications of the Company's allogeneic CAR T therapy azer-cel and up to three additional research product candidates directed to targets that Imugene US may nominate, pursuant to the terms of the License Agreement.

This Amendment No. 1 on Form 8-K/A (the "Amended 8-K") is being filed by the Company to amend Item 9.01 of the Original Form 8-K solely to (i) file the Purchase Agreement and the License Agreement as Exhibits 2.1 and 10.1, respectively, hereto, and (ii) provide pro forma financial information in connection with the Acquisition. Except as set forth herein, no modifications have been made to the information contained in the Original Form 8-K.

Item 9.01 Financial Statements and Exhibits.					
(b)	Pro Forma Financial Information . The Company's unaudited pro forma condensed consolidated financial statements as of and for the six months ended June 30, 2023 and for the twelve months ended December 31, 2022 and 2021, and the accompanying notes, are included as Exhibit 99.1 hereto and are incorporated by reference in this Item 9.01(b).				
(d)	Exhibits				
Exhibit No.	Description				
2.1†	Asset Purchase Agreement, dated as of August 15, 2023, by and among Precision BioSciences, Inc., Imugene (USA) Inc. and Imugene Limited.				
10.1#	License Agreement, effective as of August 15, 2023 by and between Precision BioSciences, Inc. and Imagene (USA) Inc.				
99.1	Unaudited Pro Forma Condensed Consolidated Financial Statements.				
104	Cover Page Interactive Data File (embedded within the Inline XBRL document).				

 \dagger Portions of this exhibit have been omitted pursuant to Item 601(b)(2)(ii) of Regulation S-K. # Portions of this exhibit have been omitted pursuant to Item 601(b)(10)(iv) of Regulation S-K.

SIGNATURES

Pursuant to the requirements of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned hereunto duly authorized.

Precision BioSciences, Inc.

Date: August 21, 2023 By: /s/ John Alexander Kelly

John Alexander Kelly Chief Financial Officer

Execution Version

Certain information marked as [***] has been excluded from this exhibit because it is both not material and is the type that the registrant treats as private or confidential.

ASSET PURCHASE AGREEMENT

BY AND AMONG

PRECISION BIOSCIENCES, INC.,

IMUGENE (USA) INC.

and

IMUGENE LIMITED

August 15, 2023

TABLE OF CONTENTS

ARTICLE	I PURCHASE AND SALE	
THEFTOLL	TOROM DE TIND OF DE	1
1.1	Purchase and Sale of Acquired Assets; Assumption of Assumed Liabilities.	1
1.2	Purchase Price and Related Matters.	5
1.3	The Closing.	6
1.4	Consents to Assignment	7
1.5	Further Assurances; Delivery of Assets	7
1.6	Withholding	7
ARTICLE	II REPRESENTATIONS AND WARRANTIES OF SELLER	O
2.1	Organization, Qualification and Corporate Power	8
2.1	Authority	9
2.3	Noncontravention	9
2.4	Undisclosed Liabilities	10
2.5	Title to Acquired Assets; Sufficiency of Assets.	10
2.6	Real Property	11
2.7	Contracts.	11
2.8	Litigation	13
2.9	Labor Matters.	13
2.10	Employee Benefits.	14
2.11	Compliance with Laws and Regulatory Matters.	16
2.12	Environmental Matters.	17
2.13	Taxes.	18
2.14	Data Privacy and Protection.	19
2.15	Insurance	19
2.16	Solvency	20
2.17	Brokers' Fees	20
ARTICLE	III REPRESENTATIONS AND WARRANTIES OF THE BUYER PARTIES	
		20
3.1	Organization	20
3.2	Authority	20
3.3	Noncontravention	21
3.4	Capitalization.	21
3.5	Broker's Fees	22
3.6	Litigation	22
3.7	Buyer Material Adverse Effect.	22
3.8	Solvency	22
3.9	No Other Representations or Warranties	23

ARTICLE IV INDEMNIFICATION

THUTCLL	TV INDEMINITION	23
4.1	Indemnification by Seller	23
4.2	Indemnification by the Buyer Parties	24
4.3	Claims for Indemnification; Mitigation.	24
4.4	Survival.	26
4.5	Limitations.	26
4.6	Treatment of Indemnification Payments	28
4.7	Security Agreement.	28
ARTICLE	V TAX MATTERS	
		30
5.1	Apportionment of Certain Taxes	30
5.2	Transfer Taxes	30
5.3	Cooperation	31
ARTICLE	VI OTHER AGREEMENTS	7.1
6.1	Access to Information and Descards, Descard Detention, Cooperation	31 31
6.2	Access to Information and Records; Record Retention; Cooperation. Retained Marks	33
6.3	Payment of Certain Monies	33
6.4	Employee Matters.	33
6.5	GMP Facility Usage	36
6.6	Deferred Assets.	37
0.0	Deterred Assets.	37
ARTICLE	VII . MISCELLANEOUS	
= 4		37
7.1	Press Releases and Announcements	37
7.2	No Third-Party Beneficiaries	38
7.3	Action to be Taken by Affiliates	38
7.4	Entire Agreement	38
7.5	Successors and Assigns	38
7.6	Notices	38
7.7	Amendments and Waivers	39
7.8	Severability	39
7.9	Expenses Pulls Transford Cours	40
7.10	Bulk Transfer Laws	40
7.11 7.12	Specific Performance	40 40
7.12 7.13	Governing Law Submission to Jurisdiction	40
	WAIVER OF JURY TRIAL	
7.14 7.15	Waiver of Conflicts; Privilege.	41 41
7.15 7.16	Construction; Certain Definitions.	41 42
7.16 7.17	·	50
	Counterparts; Electronic Signatures	
7.18	Time is of the Essence	50

Disclosure Schedule

Other Schedules

Schedule 1.1(a)(i) Acquired Contracts

Schedule 1.1(a)(ii) GMP Assets

Schedule 1.1(a)(iii) Laboratory Assets

Schedule 1.1(a)(iv) Inventory

Schedule 1.1(a)(vii) Other Acquired Assets
Schedule 1.1(b)(xvi) Other Excluded Assets
Schedule 1.2(b) Allocation Methodology
Schedule 3.4 Buyer Capitalization
Schedule 7.4(a) Target Employees
Schedule 6.5 GMP Facility Usage
Schedule 7.16(b)-1 Seller Knowledge Persons

Schedule 7.16(b)-2 Shared Contracts

Exhibits

Exhibit A Form of Bill of Sale and Assignment and Assumption Agreement

Exhibit B-1 Form of Lease Assignment and Assumption Agreement

Exhibit B-2 Form of Sublease Agreement

Exhibit C Form of License Agreement

Exhibit D Form of Transition Services Agreement

Exhibit E Form of Registration Rights Agreement

Exhibit F Form of Convertible Note Subscription Deed

Exhibit G Form of Guaranty Agreement

ASSET PURCHASE AGREEMENT

This ASSET PURCHASE AGREEMENT (this "<u>Agreement</u>") is entered into as of August 15, 2023 by and between Precision BioSciences, Inc., a Delaware corporation ("<u>Seller</u>"), Imugene (USA) Inc., a Nevada corporation ("<u>Buyer</u>"), and Imugene Limited, an Australian corporation ("<u>Buyer Parent</u>," and together with Buyer, the "<u>Buyer Parties</u>"). Seller and the Buyer Parties are sometimes referred to herein individually as a "<u>Party</u>" and together as the "<u>Parties</u>."

RECITALS

WHEREAS, Seller desires to sell or cause to be sold to Buyer, and Buyer desires to purchase from Seller, all of the Acquired Assets, and Buyer desires to assume the Assumed Liabilities, in each case upon the terms and subject to the conditions set forth in this Agreement.

NOW, THEREFORE, in consideration of the foregoing and the representations and warranties, covenants and agreements set forth herein, and for other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the Parties, intending to be legally bound, agree as follows:

ARTICLE I PURCHASE AND SALE

- 1.1 Purchase and Sale of Acquired Assets; Assumption of Assumed Liabilities.
- (a) <u>Purchase and Sale of Acquired Assets</u>. On the terms and subject to the conditions of this Agreement, at the Closing, Seller shall sell to Buyer, and Buyer shall purchase from Seller, free and clear of any Security Interest, all of Seller's right, title and interest in, to, and under the following assets (collectively, the "<u>Acquired Assets</u>"):
 - (i) the agreements set forth on <u>Schedule 1.1(a)(i)</u> (the "<u>Acquired Contracts</u>");
 - (ii) the tangible assets including machinery, manufacturing and other equipment, tools and tooling, instruments, furniture, fixtures and leasehold improvements, used or held for use at the GMP Facility, including all such items set forth on Schedule 1.1(a)(ii) (the "GMP Assets"), and any documents or manuals in Seller's possession required to operate the GMP Assets, except for such items set forth on Schedule 1.1(b)(xvi);
 - (iii) the machinery, equipment, computer hardware (but not, for the avoidance of doubt, the software or information contained therein, which Seller shall have a right to remove therefrom), tools, instruments, furniture, fixtures and leasehold improvements used or held for use at the Laboratory Facilities set forth on Schedule 1.1(a)(iii) (the "Laboratory Assets," and together with the GMP Assets, the "Equipment"), and any documents or manuals in Seller's possession required to operate the Laboratory Assets;
 - (iv) the clinical trial materials and supplies manufactured at the GMP Facility or purchased or otherwise held for use exclusively in connection with the development of azer-cel for the treatment of cancer (the "Inventory"), which includes the Inventory set forth on Schedule 1.1(a)(iv);

- (v) copies of personnel and employment records for Continuing Employees (subject to compliance with applicable Laws restricting disclosure); provided that Seller may retain the original records;
- (vi) all of Seller's rights under warranties, guaranties, indemnities, and all similar rights against third parties to the extent related to any Acquired Asset; and
 - (vii) any other assets listed on Schedule 1.1(a)(vii).
- (b) <u>Excluded Assets</u>. It is expressly understood and agreed that, notwithstanding anything to the contrary set forth herein, all right, title and interest of Seller or any of its Affiliates in, to or under the following shall constitute "<u>Excluded Assets</u>":
 - (i) any asset, right or property of Seller or any of its Affiliates that is not expressly identified in clauses (i) through (viii) of Section 1.1(a);
 - (ii) all cash and cash equivalents or similar type investments, bank accounts, certificates of deposit, security deposits, Treasury bills and other marketable securities, other than the security deposit with respect to the GMP Facility Lease;
 - (iii) all insurance policies and all rights to insurance claims, related refunds and proceeds thereunder;
 - (iv) all Contracts that are not Acquired Contracts;
 - (v) all Intellectual Property;
 - (vi) all refunds of, or credits for, Taxes relating to all periods ending on or prior to the Closing Date, whether or not arising out of the Acquired Assets;
 - (vii) the sponsorship of, and all assets maintained pursuant to or in connection with, any Employee Benefit Plans;
 - (viii) all taxpayer and other identification numbers;
 - (ix) all seals, minute books, stock transfer books, blank stock certificates, and other documents relating to the organization, maintenance and existence of Seller as a legal entity;
 - (x) the books, records, laboratory books, batch records and stability studies of Seller (the "Laboratory Records");
 - (xi) original personnel and employment records of Target Employees;
 - (xii) any rights whatsoever in the name "Precision Biosciences, Inc." in any form, formulation or presentation whatsoever or any other trademark, service mark, trade dress, logo or associated goodwill therein;
 - (xiii) all Regulatory Approvals;

- (xiv) all actions, claims, causes of action, rights of recovery, choses in action and rights of setoff of any kind arising before, at or after the Closing relating to the items set forth in this <u>Section 1.1(b)</u> or to any Excluded Assets or Excluded Liabilities (whether known or unknown, matured or unmatured, accrued or contingent);
 - (xv) all rights of Seller or any of its Affiliates under this Agreement and the other Transaction Documents; and
 - (xvi) the assets, rights and property set forth on <u>Schedule 1.1(b)(xvi)</u>.

Notwithstanding anything to the contrary herein, the Acquired Assets shall not include, and, at the Closing, Seller or its applicable Affiliate shall retain, all of the Excluded Assets; <u>provided</u>, <u>however</u>, that Buyer shall also not be restricted under, and shall not be in breach of, this Agreement from using, in connection with the use and operation of the Acquired Assets (including, for the avoidance of doubt, the operation of the GMP Facility and the research and development of azer-cel), any knowledge, skill, and expertise previously acquired by the Continuing Employees in the course of the performance of their duties at Seller ("<u>Residuals</u>") solely to the extent such Residuals shall have been retained in the unaided memory of such employees and without use of tangible copies of any confidential information of Seller; provided that use of such Residuals is on an "as is, where is" basis at Buyer's sole risk.

- (c) <u>Assumed Liabilities</u>. At the Closing, Buyer shall assume and agree to pay, perform and discharge when due all Assumed Liabilities. As used in this Agreement, "<u>Assumed Liabilities</u>" means, collectively, all of the following Liabilities (other than Excluded Liabilities) relating to the Acquired Assets, whether arising out of Law, contract or otherwise:
 - (i) all Liabilities arising out of, relating to or in connection with the Acquired Contracts from and after the Closing Date (provided that such Liabilities do not arise out of a pre-Closing breach of such contract by Seller);
 - (ii) all Liabilities relating to Hazardous Materials or any other environmental matters in connection with the Acquired Assets or the Leased Real Property arising or occurring on or after the Closing Date;
 - (iii) all Liabilities relating to violation or breach (or alleged violation or breach) of any Law (including any Environmental Law) in connection with the ownership, control or use of the Acquired Assets occurring from and after the Closing; and
 - (iv) all other Liabilities to the extent arising out of or relating to the ownership, control or use of the Acquired Assets by Buyer from and after the Closing.
- (d) <u>Excluded Liabilities</u>. It is expressly understood and agreed that, except for the Assumed Liabilities, and without limiting Buyer's or any of its Affiliates' obligations under any other Transaction Document, Buyer shall not assume, shall not take subject to and shall not be liable for any other Liabilities of Seller (the "<u>Excluded Liabilities</u>"). For clarity, the Excluded Liabilities shall include, among others, the following Liabilities of Seller:

- (i) subject to <u>Section 1.6</u> and <u>Article V</u>, any Liability for any Taxes, including (A) any Tax of Seller with respect to any Taxable period (or portion thereof), whether before or after the Closing Date, other than Taxes relating to the ownership or operation of the Acquired Assets from and after the Closing, (B) any Tax of Seller resulting from or attributable to the consummation of the Transactions, (C) any Taxes relating to the ownership or operation of the Acquired Assets for the Taxable periods (or portions thereof) ending on the Closing Date (including any amounts required to be withheld from payments to employees or independent contractors), or (D) any Taxes of any person other than Seller under Treasury Regulation Section 1.1502-6 (or any similar provision of state, local, or foreign Laws), as a transferee or successor, by contract or otherwise;
- (ii) all trade accounts payable accrued and unpaid or other current liabilities of Seller as of immediately prior to the Closing;
- (iii) all Liabilities relating to Hazardous Materials or any other environmental matters in connection with the Acquired Assets or the Leased Real Property to the extent existing as of immediately prior to the Closing;
- (iv) all arbitrations, claims, actions or proceedings arising out of or relating to the manufacture, production or Exploitation of any products or product candidates within the Acquired Assets or the ownership, sale, lease or use of any of the Acquired Assets, in each case, solely to the extent based on actions, events or circumstances arising prior to the Closing;
 - (v) all Indebtedness of Seller;
 - (vi) all Liabilities to the extent arising out of or relating to the Excluded Assets;
- (vii) all Liabilities to the extent resulting from or arising out of Seller's ownership, operation or control of the Acquired Assets prior to the Closing;
 - (viii) all Liabilities to any stockholders of Seller in their capacity as such;
 - (ix) all Liabilities for wrongful conduct occurring prior to the Closing Date of employees of Seller;
 - (x) all Liabilities in respect of Transaction Costs; and
- (xi) all Liabilities to third parties in respect of breaches or other disputes under the Acquired Contracts or the Shared Contracts to the extent existing prior to the Closing.
- 1.2 Purchase Price and Related Matters.
- (a) <u>Purchase Price</u>. In consideration for the sale and transfer of the Acquired Assets, and subject to the terms and conditions of this Agreement and the other Transaction Documents, Buyer shall (i) at the Closing (A) assume the Assumed Liabilities as provided in <u>Section 1.1(c)</u>, (B) pay to Seller, in cash, by wire transfer of immediately available funds, an amount equal to \$8,304,000 (the "<u>Closing Cash Consideration</u>"), and (C) issue to Seller convertible notes (the

"Convertible Notes"), pursuant to the terms and conditions set forth in the Convertible Note Subscription Deed in the form set forth on Exhibit F hereto (the "Convertible Note Subscription Deed") and the Note Deed Poll (the "Note Deed Poll") in the form set forth on Schedule 4 thereto (collectively, the "Purchase Price").

(b) Allocation. Not later than sixty (60) days following the Closing, Buyer shall prepare and deliver to Seller a statement of allocation which shall provide for the allocation of the Purchase Price (inclusive of the Assumed Liabilities, treated as consideration paid to Seller pursuant to Section 1060 of the Internal Revenue Code of 1986, as amended (the "Code")) among the Acquired Assets (the "Allocation Schedule"). Such Allocation Schedule shall be prepared in accordance with the provisions of Code Section 1060 and the Treasury Regulations promulgated thereunder and applying the methodology set forth on Schedule 1.2(b) to determine the agreed fair market value of the Acquired Assets as contemplated in Treasury Regulation Section 1.1060-1(c)(4), (the "Allocation Methodology"). Within thirty (30) days after the receipt of such Allocation Schedule, Seller will propose to Buyer in writing any reasonable changes to such Allocation Schedule together with reasonable support for such changes (and in the event that no such changes are proposed in writing to Buyer within such time period, Seller will be deemed to have agreed to, and accepted, the Allocation Schedule). Buyer and Seller will attempt in good faith to resolve any differences with respect to the Allocation Schedule, in accordance with the Allocation Methodology, within fifteen (15) days after Buyer's receipt of a timely written notice of objection from Seller. If Buyer and Seller are unable to resolve such differences within such time period, then any remaining disputed matters will be submitted to the Independent Accountant for resolution, in accordance with the Allocation Methodology. Promptly, but not later than fifteen (15) days after such matters are submitted to it for resolution hereunder, the Independent Accountant will determine those matters in dispute and will render a written report as to the disputed matters and the resulting allocation of such Purchase Price, which report shall, absent manifest error, be conclusive and binding upon Buyer and Seller. The fees and expenses of the Independent Accountant in respect of such report shall be paid one-half by Buyer and one-half by Seller. Buyer and Seller shall each file or cause to be filed IRS Form 8594 for its taxable year that includes the Closing Date in a manner consistent with the allocation set forth on the Allocation Schedule as so finalized, and (except as set forth below relating to a revised Allocation Schedule) shall not take any position on any Tax Return or in the course of any Tax audit, review, or litigation inconsistent with the allocation provided in the Allocation Schedule. In the event that any adjustment is required to be made to the Allocation Schedule as a result of any adjustment to the Purchase Price pursuant to Article IV of this Agreement or otherwise, Buyer shall prepare or cause to be prepared, and shall provide to Seller, a revised Allocation Schedule reflecting such adjustment. Such revised Allocation Schedule shall be prepared consistent with the Allocation Methodology and subject to review and resolution of timely raised disputes in the same manner as the initial Allocation Schedule. Each of Buyer and Seller shall file or cause to be filed a revised IRS Form 8594 reflecting such adjustments as so finalized for its Taxable year that includes the event or events giving rise to such adjustment, and (except as required by any future revised Allocation Schedule) shall not take any position on any Tax Return or in the course of any Tax audit, review, or litigation inconsistent with the allocation provided in the revised Allocation Schedule.

1.3 The Closing.

(a) <u>Time and Location</u>. The closing of the Transactions (the "<u>Closing</u>") shall take place remotely by exchange of documents and signatures (or their electronic counterparts) simultaneously with the execution of this Agreement, or such other date or time as the Parties may mutually determine (the "<u>Closing Date</u>").

(b) Actions at the Closing. At the Closing:

- (i) Seller and Buyer shall execute and deliver to each other a bill of sale and assignment and assumption agreement (the "Bill of Sale and Assignment and Assumption Agreement") in the form attached hereto as Exhibit A;
- (ii) Seller and Buyer shall execute and deliver to each other (A) a lease assignment and assumption agreement (the "Lease Assignment and Assumption Agreement") in the form attached hereto as Exhibit B-1 assigning a leasehold interest in the leased real property described therein and in any structures, improvements, buildings and facilities located on such leased real property (the "GMP Facility") and (B) a sublease agreement (the "Sublease Agreement") in the form attached hereto as Exhibit B-2 assigning a sublease interest in the leased real property described therein and in any structures, improvements, buildings and facilities located thereon (the "Laboratory Facilities," and collectively with the GMP Facility, the "Leased Real Property");
- (iii) Seller and Buyer shall execute and deliver to each other a license agreement (the "<u>License Agreement</u>") in the form attached hereto as Exhibit C;
- (<u>iv</u>)_Seller and Buyer shall execute and deliver to each other a transition services agreement (the "<u>Transition Services Agreement</u>") in the form attached hereto as <u>Exhibit D</u>;
- (v) Seller and the Buyer Parties shall execute and deliver to each other the Convertible Note Subscription Deed and the Note Deed Poll, and Buyer shall issue Seller the Convertible Notes in accordance therewith;
- (vi) Seller and Buyer Parent shall execute and deliver to each other a registration rights agreement (the "Registration Rights Agreement") in the form attached hereto as Exhibit E;
- (vii) Seller and Buyer Parent shall execute and deliver to each other the guaranty agreement (the "<u>Guaranty</u> <u>Agreement</u>") in the form attached hereto as <u>Exhibit G</u>; and
 - (viii) Seller shall deliver to Buyer a properly completed and duly executed IRS Form W-9.

1.4 Consents to Assignment

. Anything in this Agreement to the contrary notwithstanding, this Agreement shall not constitute an agreement to assign or transfer any Acquired Asset if an attempted assignment or transfer thereof, without the consent of a third party thereto or of the issuing Governmental Entity,

as the case may be, would constitute a breach or default thereof, would result in a violation of the rights of any such third party, would be ineffective, or would in any way adversely affect the rights of Seller or Buyer thereunder and such consent is not obtained at or prior to the Closing (a "Deferred Consent"). With respect to each Deferred Consent, (a) the Acquired Asset to which such Deferred Consent relates (a "Deferred Item") shall be withheld from sale pursuant to this Agreement without any reduction in the Purchase Price, (b) for a period of twelve (12) months after the Closing, Seller and Buyer shall, at Buyer's expense, cooperate, in all reasonable respects, to obtain such Deferred Consent as soon as practicable after the Closing, provided that Seller shall not be required to make any payments that are not reimbursed by Buyer or agree to any material undertakings in connection therewith; and (c) until such Deferred Consent is obtained or a replacement Contract is entered into by Buyer or an Affiliate of Buyer, Seller and Buyer shall, at Buyer's expense, cooperate, in all reasonable respects, in any lawful and commercially reasonable arrangement reasonably proposed by Buyer under which (i) Buyer would obtain (without infringing upon the legal rights of any third party) the economic claims, rights and benefits (net of the amount of any related Tax costs and any other Liabilities imposed on Seller or any of its Affiliates) with respect to the Deferred Item.

1.5 Further Assurances; Delivery of Assets

. At any time and from time to time after the Closing Date, as and when requested by any Party and at the requesting Party's expense, the other Parties shall promptly execute and deliver, or cause to be executed and delivered, all such documents, instruments and certificates and shall take, or cause to be taken, all such further or other actions as are necessary to evidence and effectuate the Transactions. Subject to the terms and conditions of this Agreement, Seller shall move or otherwise transport the tangible Acquired Assets (except for Acquired Assets that will remain at the Leased Real Property) to the Leased Real Property within twenty (20) Business Days after the Closing Date, as applicable. The Seller shall pay the costs of and bear the risks associated with the physical removal, shipping and transfer of any Acquired Assets to the Buyer; provided that Buyer agrees to cooperate with the Seller in a commercially reasonable manner, at the Seller's expense, with respect to such removal, shipping, transfer and move.

1.6 Withholding

. Buyer shall be entitled to deduct and withhold from any amounts otherwise payable pursuant to this Agreement (or, as the case may be, be promptly reimbursed therefor) as Buyer reasonably determines that it is required to deduct and withhold with respect to the making of such payment under the Code or any provision of state, local, or non-U.S. Tax Law; provided, however, if the Buyer Parties determine that any withholding or deduction is required under applicable Tax Law with respect to any payment under this Agreement, then the Buyer shall use commercially reasonable efforts to provide notice to Seller as promptly as reasonably practicable prior to making such payment, and the Buyer shall reasonably cooperate to reduce or eliminate such withholding or deduction, including by providing any applicable forms or exemption certificates, and the Buyer shall provide to Seller any Tax receipts or other evidence of payment in respect of any deductions or withholdings made. The Buyer Parties acknowledge and agree that, if any Tax, including any Australian Royalty Withholding Tax, but excluding any such Tax resulting from failure of the

Seller to provide the form described in <u>Section 1.3(b)(viii)</u>, is required to be deducted or withheld from any payment to Seller hereunder, the Buyer Parties shall pay an additional amount together with the payment so that, after making any deduction or withholding on account of such Tax (including any deduction or withholding on the additional payment), Seller receives an amount equal to the payment which would have been due if no such deduction or withholding had been required.

ARTICLE II REPRESENTATIONS AND WARRANTIES OF SELLER

Seller represents and warrants to Buyer as of the date hereof that the following representations and warranties are true and correct, except (a) as set forth in the disclosure schedule provided by Seller to Buyer (the "<u>Disclosure Schedule</u>") or (b) as otherwise contemplated by this Agreement.

2.1 Organization, Qualification and Corporate Power

. Seller is a corporation duly organized, validly existing and in good standing under the laws of Delaware and is duly qualified to conduct business under the laws of each other jurisdiction where the character of the properties owned, leased or operated by it or the nature of its activities makes such qualification necessary, except for any such failure to be qualified that would not reasonably be expected to result in a Target Material Adverse Effect. Seller has all requisite corporate power and authority to carry on the business in which it is now engaged and to own and use the properties now owned and used by it. For purposes of this Agreement, "Target Material Adverse Effect" means any event, change, effect, development or circumstance that has a material adverse effect on the Acquired Assets, taken as a whole; provided, however, that a "Target Material Adverse Effect" shall not include any adverse event, change, effect, development or circumstance directly or indirectly resulting from, arising out of or attributable to (a) actions taken by the Parties pursuant to this Agreement or at the request or with the consent of the Buyer Parties, or the failure to take any action prohibited by this Agreement; (b) the negotiation, execution, announcement, pendency or performance of this Agreement or the Transactions, the consummation of the Transactions, the identity of (or facts or circumstances related to) the Buyer Parties, or any communications or disclosure by or on behalf of any Party regarding this Agreement or the Transactions, including, in any such case, the impact thereof on relationships, contractual or otherwise, with customers, licensors, licensees, partners, suppliers, vendors or employees; (c) changes in Seller's industry or in markets generally and not specifically relating to the Acquired Assets (including any such changes arising out of the Pandemic); (d) changes in economic conditions or financial markets in any country or region or globally, including changes in interest or exchange rates and changes in currency and credit markets; (e) changes in general legal, Tax, regulatory, political or business conditions in any country or region; (f) national or international political, regulatory or social conditions, including acts of war, armed hostilities, sabotage or terrorism, or any escalation or worsening of any such acts of war, armed hostilities, sabotage or terrorism threatened or underway as of the date of this Agreement; (g) earthquakes, hurricanes, tsunamis, tornadoes, floods, mudslides, wild fires or other natural disasters, weather conditions, outbreak of any disease or other public health event (including, for the avoidance of doubt, the Pandemic and responses thereto) and other force majeure events in any country or region; (h) changes in Law (including the Pandemic Measures) or other legal or regulatory conditions (or the

interpretation thereof) or changes in accounting standards, rules or principles (or the interpretation thereof); and (i) any change, development or state of facts relating to (A) the products or product candidates of any third party, including the entry into the market of products competitive with any product or product candidate of the Acquired Assets, (B) any failure to receive any Regulatory Approvals for any product or product candidate, any failure to conduct successful clinical trials for any product or product candidate, or any other regulatory development affecting any product candidate, or (C) the results of, or any announcement or publication related to, any clinical trials or studies undertaken by any third party and any negative publicity or unfavorable media attention resulting therefrom.

2.2 Authority

. Seller has all requisite corporate power and authority to execute and deliver this Agreement and the other Transaction Documents and perform its obligations hereunder and thereunder. The execution and delivery of this Agreement and the other Transaction Documents by Seller, the performance by Seller of its obligations hereunder and thereunder and the consummation by Seller of the Transactions have been duly and validly authorized by all necessary corporate and other action on the part of Seller. This Agreement has been, and the other Transaction Documents will be, upon execution thereof, duly and validly executed and delivered by Seller and, assuming this Agreement and the other applicable Transaction Documents constitute or will, upon execution, constitute the valid and binding agreement of Buyer or Buyer Parent, as applicable, constitutes or will, upon execution, constitute a valid and binding obligation of Seller, enforceable against Seller in accordance with their respective terms, except as enforceability may be limited by bankruptcy, insolvency, fraudulent transfer, reorganization, moratorium or other similar laws relating to or affecting the rights of creditors generally and by equitable principles, including those limiting the availability of specific performance, injunctive relief and other equitable remedies and those providing for equitable defenses (the "Enforceability Exceptions").

2.3 Noncontravention

- . Neither the execution and delivery of this Agreement or the Transaction Documents by Seller, nor the consummation by Seller of the Transactions, will:
 - (a) conflict with or violate any provision of the Governing Documents of Seller;
- (b) require on the part of Seller any Regulatory Approval filing with, or any Permit, authorization, consent or approval of, any court, arbitrational tribunal, administrative agency or commission or other governmental authority or agency (a "Governmental Entity"), except for any filing, Permit, authorization, consent or approval that would not reasonably be expected to be material to the ownership and use of the Acquired Assets;
- (c) conflict with, result in a breach of, constitute (with or without due notice or lapse of time or both) a default under, result in the acceleration of, create in any party the right to accelerate, terminate, modify or cancel, or require any consent or waiver under, any Acquired Contract, except for those Acquired Contracts set forth on Section 2.3(c) of the Disclosure Schedule, in each case, except as would not reasonably be expected to be material to the ownership and use of the Acquired Assets;

- (d) result in the imposition of any Security Interests upon any of the Acquired Assets; or
- (e) violate any order, writ, injunction or decree specifically naming, or statute, rule or regulation applicable to, Seller or any of or its properties or assets, except for any violation that would not reasonably be expected to be material to the ownership and use of the Acquired Assets.

2.4 Undisclosed Liabilities

. To Seller's Knowledge, Seller (solely with respect to the Acquired Assets) does not have any Liability that would be required to be included on a balance sheet in compliance with GAAP, except for (a) Liabilities constituting future performance obligations under the Acquired Contracts, (b) Liabilities incurred in the ordinary course of business (none of which relate to a breach of any Acquired Contract, breach of warranty, tort, infringement or violation of Law), (c) Liabilities disclosed on the Disclosure Schedule, (d) Excluded Liabilities and (e) Liabilities that would not reasonably be expected to have a Target Material Adverse Effect.

2.5 <u>Title to Acquired Assets; Sufficiency of Assets.</u>

- (a) Seller has good and marketable title to, a valid leasehold interest in or a valid license to use all of the material tangible personal property included in the Acquired Assets, free and clear of any Security Interests. For purposes of this Agreement, "Security Interest" means any mortgage, pledge, security interest, encumbrance, charge or other lien (whether arising by contract or by operation of law), other than (a) mechanic's, materialmen's, landlord's and similar liens, (b) liens arising under worker's compensation, unemployment insurance, social security, retirement and similar legislation, (c) liens on goods in transit incurred pursuant to documentary letters of credit, in each case arising in the ordinary course of business, (d) liens for Taxes not yet due and payable, (e) liens for Taxes which are being contested in good faith and by appropriate proceedings, (f) liens relating to capitalized lease financings or purchase money financings that have been entered into in the ordinary course of business, (g) liens under Intellectual Property licenses, (h) liens arising solely by action of any Buyer Party (including, for the avoidance of doubt, under any Indebtedness or other secured obligations of Buyer or any Affiliate of Buyer), (i) Security Interests granted to Seller under this Agreement and (j) liens which do not materially and adversely impair the use of the Acquired Assets, taken as a whole.
- (b) As a result of the consummation of the Transactions and the execution of the instruments of transfer contemplated by this Agreement, assuming all Transaction Documents are fully executed, the accuracy of Buyer's representations and warranties herein, and the conditions to Closing occur or are waived in writing, Buyer will, upon the Closing, hold good, valid and marketable title to, a valid leasehold interest in or a valid license to use, the Acquired Assets, free and clear of all Security Interests.
- (c) Assuming (i) receipt of the consents applicable to the Acquired Contracts indicated in <u>Schedule 1.1(a)(i)</u>, (ii) the transfer or replacement of the Permits set forth on <u>Section 2.11(f)</u> of the Disclosure Schedule, and (iii) the replacement of the Shared Contracts, the Acquired Assets, together with the services, assets, and Intellectual Property and other rights granted to the Buyer Parties under the Transaction Documents, are sufficient to enable the Buyer Parties, immediately

following the Closing, to conduct the research of azer-cel for the treatment of cancer at its present stage of clinical development in all material respects as conducted by Seller as of the date hereof, except as set forth in $\underline{\text{Section 2.5(c)}}$ of the Disclosure Schedule.

2.6 Real Property

. Seller does not own any real property that relates to the Acquired Assets. Section 2.6 of the Disclosure Schedule includes a complete and accurate list of all the real property leases included in the Acquired Assets (the "Leases"). Seller has made available to Buyer correct and complete copies of the Leases, each as amended to date. To the Knowledge of Seller, as of the date hereof, the buildings, plants, facilities, installations, fixtures and other structures or improvements themselves included as part of, or located on or at, the Leases and Seller's activities at the Leases, are not in material violation of, or in material conflict with, any applicable zoning regulations or ordinances. Seller has the right to quiet enjoyment of the real property subject to each Lease for the full term of such Lease. As of the date hereof, there exist no defaults on the part of Seller under the Leases nor, to the Knowledge of Seller, any state of facts which, upon notice or lapse of time, or both, would constitute a default under the Leases.

2.7 Contracts.

- (a) As of the date of this Agreement, except as set forth in <u>Section 2.7</u> of the Disclosure Schedule, the Acquired Contracts do not include, and, except for the Shared Contracts, there are no Contracts of Seller in effect on the date hereof that relate primarily to the Acquired Assets and to which, effective as of the Closing, the Acquired Assets will be subject that include:
 - (i) any agreement entered into other than in the ordinary course of business pursuant to which Seller received more than \$[***] in the year ended December 31, 2022 or would reasonably be expected to receive more than such amount in the year ending December 31, 2023;
 - (ii) any lease or other agreement under which Seller is lessee of, or holds or operates, any machinery, equipment or other tangible personal property owned by a third party that is material to the ownership, operation and control of the Acquired Assets, taken as a whole.
 - (iii) any agreement entered into other than in the ordinary course of business for the purchase of services, materials, supplies or equipment which involved the payment by Seller of more than \$[***] in the year ended December 31, 2022 or would reasonably be expected to involve the payment of more than such amount in the year ending December 31, 2023;
 - (iv) any agreement establishing a partnership or joint venture or any research, collaboration or development agreement, other than any such agreement entered into in the ordinary course of business that is terminable by Seller without penalty with 60 days' or less notice;
 - (v) any agreement evidencing indebtedness for borrowed money or any guaranty thereof or any Security Interest in the Acquired Assets;

- (vi) any agreement between Seller and any director, officer or employee of Seller, other than offer letters for atwill employment that do not provide for any severance benefit upon termination of employment;
- (vii) any contract, agreement or other arrangement granting a right of first offer, right of first refusal, or any purchase rights to any Acquired Asset that is material to the ownership and use of the Acquired Assets;
- (viii) any contract, agreement or other arrangement imposing a non-competition or non-solicitation obligation, including exclusive dealing arrangements, that, following the Closing, would be binding on Buyer, except for any such contract, agreement or arrangement that is terminable by Seller without penalty or payment upon 60 days' or less notice;
- (ix) any licenses of any material Intellectual Property owned by a third party, other than "off-the-shelf" or "shrink wrap" licenses entered in the ordinary course of business; or
 - (x) any settlement agreements related to the Acquired Assets.
- (b) Seller has made available to Buyer a correct and complete copy of each Acquired Contract and Shared Contract. With respect to each Acquired Contract and Shared Contract:
 - (i) such Contract is a legal, valid, binding and enforceable obligation of Seller and, to Seller's Knowledge, of each other party thereto (except as enforceability may be limited by the Enforceability Exceptions);
 - (ii) there exist no material breaches or defaults of Seller thereunder;
 - (iii) to Seller's Knowledge, no event has occurred and no condition exists that constitutes, or which, with notice and/or the passage of time would constitute, a material breach, a default or event of default by Seller under such Contract; and
 - (iv) Seller has not made a claim with respect to, or to Seller's Knowledge, received notice of, any material breach or default by any party thereto.
- (c) <u>Section 2.7(c)</u> of the Disclosure Schedule sets forth all of the Milestone Payments (as defined in the [***] License) previously paid by Seller under the [***] License and all of the credits that have accrued to either Seller or [***] ("[***]") under the [***] Agreements as of the Closing Date. There are no payments currently due and payable to [***] under or in connection with the [***] Agreements.

2.8 Litigation

. Except as set forth on <u>Section 2.8</u> of the Disclosure Schedule, there are no and has not been for the last [***] years, any (a) judgment, order, decree, stipulation or injunction specifically naming Seller relating to the Acquired Assets or (b) claim, complaint, action, suit, proceeding, hearing or investigation of or in any Governmental Entity relating to the Acquired Assets to which

Seller is a party or, to Seller's Knowledge, which has been threatened in writing against Seller that, in the case of either clause (a) or (b), would reasonably be expected to be material to the ownership, operation and control of the Acquired Assets.

2.9 Labor Matters.

- (a) <u>Section 2.9</u> of the Disclosure Schedule lists, as of the date of this Agreement, each collective bargaining or works council agreement related to the Continuing Employees to which Seller is a party or is bound. Seller (solely with respect to the Continuing Employees) has not experienced since [***], any material strikes, grievances, claims of unfair labor practices or other collective bargaining or works council disputes.
- (b) <u>Section 2.9(b)(i)</u> of the Disclosure Schedule sets forth a complete and correct list of all Target Employees (including names, title/positions, dates of hire, status (active or on leave of absence), and annual compensation or rates of pay). Except as set forth on <u>Section 2.9(b)(ii)</u> of the Disclosure Schedule, no Target Employee has given as of the date hereof written notice of an intention to leave Seller's employ before or after the Closing. Subject to applicable Laws, upon termination of the employment or engagement of any employees of Seller, Buyer shall not be liable, by reason of the purchase of the Acquired Assets or anything done prior to the Closing Date, to any Target Employees for severance, retention pay or any other compensation payments (including accrued salary, vacation or sick pay in accordance with normal policies) with respect to pre-Closing periods.
- (c) Seller is, and has been for the last [***] years, with respect to the Continuing Employees, in compliance in all material respects with all applicable Laws respecting employment and employment practices, terms and conditions of employment, wages and hours, training, and recordkeeping, including any provisions relating to (i) wages, hours, meal and rest periods, bonuses, overtime pay, commissions, termination pay, vacation pay, sick pay, any other form of compensation, and classification of employees or contractors for purposes of compensation or otherwise; (ii) unlawful, wrongful, retaliatory, harassing, or discriminatory employment or labor practices; (iii) occupational health and safety standards; (iv) leaves of absence and accommodation of disability or religion; and (v) immigration and records supporting authorization to work in the United States, workers' compensation, disability, unemployment compensation, employee privacy rights, whistleblower laws, and all other federal, state and local employment Laws. Seller is not, and has not been for the last [***] years, engaged in any unfair labor or unlawful employment practice with respect to the Continuing Employees. With respect to the Continuing Employees, there is no (i) wage complaint or investigation pending with the United States Department of Labor's Wage and Hour Division, the North Carolina Department of Labor, the North Carolina Attorney General, or, to the Knowledge of Seller, threatened in writing, against or involving or affecting Seller; (ii) unlawful employment practice discrimination charge pending before the Equal Employment Opportunity Commission (the "EEOC"), any EEOC recognized state "referral agency," any state or federal court, or, to the Knowledge of Seller, threatened in writing, against or involving or affecting the Continuing Employees; or (iii) unfair labor practice charge or complaint against Seller involving the Continuing Employees pending before the National Labor Relations Board, or to the Knowledge of Seller, threatened in writing, against or involving or affecting Seller involving the Continuing Employees.

(d) There are no actions, suits, claims or legal, administrative or arbitratorial proceedings pending or, to the Knowledge of Seller, threatened in writing, against Seller (whether under federal or state law, under any employment agreement or otherwise) asserted by any Continuing Employee or by any Governmental Entity with respect to any Continuing Employee. Seller is not subject to any actions, suits, claims or legal, administrative or arbitratorial proceedings or Governmental Entity inquiry or investigation relating to any Continuing Employee. No Governmental Entity has issued any finding of any violation of any Law regarding the Continuing Employees in their capacity as employees of Seller.

2.10 Employee Benefits.

- (a) Section 2.10(a) of the Disclosure Schedule contains a complete and accurate list of all Employee Benefit Plans currently maintained, or contributed to, by Seller or any ERISA Affiliate for the benefit of Target Employees (and their beneficiaries) (the "Target Benefit Plans"). For purposes of this Agreement, "Employee Benefit Plan" means any "employee pension benefit plan" (as defined in Section 3(2) of the Employee Retirement Income Security Act of 1974 ("ERISA")), any "employee welfare benefit plan" (as defined in Section 3(1) of ERISA), whether or not subject to ERISA, and any other written plan, agreement or arrangement involving direct or indirect compensation, including insurance coverage, severance benefits, disability benefits, deferred compensation, bonuses, stock options, stock purchase, phantom stock, stock appreciation or other forms of incentive compensation or post-retirement compensation, retention bonuses, fringe benefits, flexible spending accounts, insurance premium reimbursements or other benefits, but excluding any Employee Benefit Plan maintained or contributed to under foreign law. For purposes of this Agreement, "ERISA Affiliate" means any entity which is a member of (i) a controlled group of corporations (as defined in Section 414(b) of the Code), (ii) a group of trades or businesses under common control (as defined in Section 414(c) of the Code), or (iii) an affiliated service group (as defined under Section 414(m) of the Code or the regulations under Section 414(o) of the Code), any of which includes Seller. Complete and accurate copies of all Target Benefit Plans and all material related trust agreements, insurance contracts and summary plan descriptions have been made available to Buyer. Each Target Benefit Plan has been administered in accordance with its terms and Seller has met its obligations with respect to such Target Benefit Plan, except for any failure to so administer or so meet its obligations that would not reasonably be expected to be material to the ownership and use of the Acquired Assets. Seller and the Target Benefit Plans are in compliance with the currently applicable provisions of ERISA and the Code, except for any failure to so comply that would not reasonably be expected to be material to the ownership and use of the Acquired Assets.
- (b) There are no claims (except claims for benefits payable in the normal operation of the Target Benefit Plans and proceedings with respect to qualified domestic relations orders), suits or proceedings against or involving any Target Benefit Plan or asserting any rights or claims to benefits under any Target Benefit Plan, or, to Seller's Knowledge, investigations by any Governmental Entity involving any Target Benefit Plan, except for any such claims, suits, proceedings or investigations that would not reasonably be expected to be material to the ownership and use of the Acquired Assets.
- (c) The Target Benefit Plans that are intended to be qualified under Section 401(a) of the Code have received determination, opinion or advisory letters from the Internal Revenue

Service to the effect that such Target Benefit Plans are qualified and the plans and the trusts related thereto are exempt from federal income Taxes under Sections 401(a) and 501(a), respectively, of the Code, and, to the Knowledge of Seller, nothing has occurred that would reasonably be expected to adversely affect the qualification of such Target Benefit Plans.

- (d) None of Seller or its ERISA Affiliates has within the past [***] years (i) maintained an Employee Benefit Plan that was subject to Section 412 of the Code or Title IV of ERISA or (ii) been obligated to contribute to a "multiemployer plan" (as described in Section 4001(a)(3) or Section 3(37) of ERISA), in each case of clauses (i) and (ii) solely with respect to the ownership and use of the Acquired Assets.
- (e) There are no unfunded obligations under any Target Benefit Plan providing welfare benefits after termination of employment to any Target Employee (or to any beneficiary of any such Target Employee), excluding continuation of health coverage required to be continued under Section 4980B of the Code or other applicable laws and the extension of coverage during severance periods.
- (f) No act or omission has occurred and no condition exists with respect to any Target Benefit Plan maintained by Seller or any ERISA Affiliate that would subject Seller or any ERISA Affiliate to any fine, penalty, Tax or Liability of any kind imposed under ERISA or the Code (other than Liabilities for benefits accrued under Target Benefit Plans for employees of Seller and each of their beneficiaries), except for any fine, penalty, Tax or Liability that would not reasonably be expected to be material to the ownership and use of the Acquired Assets.
- (g) The Parties agree that the only representations and warranties of Seller in this Agreement as to any matters relating to employee benefits are those contained in this <u>Section 2.10</u>.

2.11 Compliance with Laws and Regulatory Matters.

- (a) Seller (solely with respect to the Acquired Assets) is in compliance with all applicable Laws of any Governmental Entity currently in effect with respect to the Acquired Assets, including the FDCA and applicable implementing regulations and guidance issued by the FDA thereunder, except in each case where the failure to comply therewith would not reasonably be expected to be material to the ownership and use of the Acquired Assets. As of the date of this Agreement, Seller has not received written notice of any pending action, suit, proceeding, hearing, investigation, claim, demand or notice relating to the Acquired Assets alleging any failure to so comply, other than those that would not reasonably be expected to be material to the ownership and use of the Acquired Assets.
- (b) All material filings, declarations, listings, registrations, reports or submissions, including adverse event and other safety reports, required to be filed by Seller with any Regulatory Authority with respect to the Acquired Assets have been filed. All such filings, declarations, listings, registrations, reports or submissions were in compliance with applicable laws when filed (except for any noncompliance that would not reasonably be expected to be material to the ownership and use of the Acquired Assets), and, as of the date of this Agreement, no material deficiencies have been asserted in writing by any applicable Regulatory Authority with respect to any such filings, declarations, listing, registrations, reports or submissions.

- (c) All preclinical and clinical investigations with respect to any product or product candidate sponsored by or on behalf of Seller are being conducted in compliance with applicable laws, including Good Clinical Practices (as defined in the FDCA) and the Health Insurance Portability and Accountability Act of 1996, as amended by the Health Information Technology for Economic and Clinical Health Act of 2009, and the regulations promulgated thereunder, and any applicable state privacy laws, except in each case for any noncompliance that would not reasonably be expected to be material to the ownership and use of the Acquired Assets. As of the date of this Agreement, Seller has not received any written notice from any Regulatory Authority with respect to any clinical or pre-clinical studies or tests concerning the Acquired Assets and requiring the termination, suspension or material modification of such studies or tests. To Seller's Knowledge, no person involved in any preclinical and clinical investigations concerning the Acquired Assets and sponsored by Seller has been convicted of any crime or engaged in any conduct for which debarment is mandated by 21 U.S.C. § 335a(a) or any similar law or authorized by 21 U.S.C. § 335a(b) or any similar law.
- (d) Seller has made available to Buyer all applications, registrations, licenses, authorizations, approvals and correspondence submitted to or received from any Regulatory Authority (including minutes and official contact reports relating to any communications with any Regulatory Authority) and all supporting documents and all preclinical studies and other data, relating to the Acquired Assets in Seller's possession or control, in each case to the extent requested by Buyer and material to the ownership, control and use of the Acquired Assets.
- (e) Seller has not made an untrue statement of a material fact to any Regulatory Authority, nor failed to disclose to any Regulatory Authority any material fact required to be disclosed to such Regulatory Authority, in violation of the "Fraud, Untrue Statements of Material Facts, Bribery, and Illegal Gratuities" set forth in 56 Fed. Reg. 46191 (September 10, 1991). To Seller's Knowledge, as of the date of this Agreement, Seller (solely with respect to the Acquired Assets) (A) is not the subject of any pending or threatened investigation by any Regulatory Authority pursuant to its "Fraud, Untrue Statements of Material Facts, Bribery, and Illegal Gratuities" policy, or similar policy, and (B) has not received any notification of any such potential investigation. Seller (solely with respect to the Acquired Assets) has not been convicted of any crime or engaged in any conduct for which debarment is mandated by 21 U.S.C. § 335a(a) or any similar law or authorized by 21 U.S.C. Section 335a(b) or any similar law.
- (f) Seller currently has all Permits that are required for the operation of the Acquired Assets as presently conducted by Seller, a list of which is set forth on Section 2.11(f) of the Disclosure Schedule. As of the date hereof, Seller is not in default or violation (and, to Seller's Knowledge, no event has occurred that, with notice or the lapse of time or both, would constitute a default or violation) of any term, condition or provision of any Permit held by Seller required for the operation of the Acquired Assets, except where such default or violation would not reasonably be expected to be material to the ownership, operation and control of the Acquired Assets.

2.12 Environmental Matters.

(a) Seller owns, holds or is in possession of all material Permits issued under Environmental Laws that are required for the operation of the Acquired Assets as presently conducted, except where the absence of which would not reasonably be expected to result in

material liability to Seller ("Environmental Permits"). As of the date hereof, Seller is not in default or violation (and, to Seller's Knowledge, no event has occurred that, with notice or the lapse of time or both, would constitute a default or violation) of any term, condition or provision of any Environmental Permit, except where such default or violation would not reasonably be expected to be material to the ownership, operation and control of the Acquired Assets.

- (b) As of the date of this Agreement, Seller has not received any written notice or written report regarding any actual or alleged releases, discharges or spills of Hazardous Materials by Seller (solely with respect to the Acquired Assets) in violation of any Environmental Laws, except for any such violation that would not reasonably be expected to be material to the ownership and use of the Acquired Assets. As of the date of this Agreement, Seller has not received written notice of any pending action, suit, proceeding, hearing, investigation, claim, demand or notice relating to the Acquired Assets alleging any failure to so comply, other than those that would not reasonably be expected to be material to the ownership and use of the Acquired Assets.
- (c) The Parties agree that the only representations and warranties of Seller in this Agreement as to any Environmental Laws, Hazardous Materials or any other environmental matters are those contained in this <u>Section 2.12</u>.

2.13 Taxes.

- (a) All Tax Returns (including any consolidated, combined, unitary, or other similar Tax Return that includes or is required to include Seller) required to be filed by Seller or otherwise with respect to the Acquired Assets have been timely filed. All such Tax Returns are true, correct and complete in all material respects, and all Taxes owed or required to be paid by Seller or otherwise with respect to the Acquired Assets, whether or not shown or required to be shown on any Tax Return, have been paid. Seller has not received written notice of any claim by any taxing authority in any other jurisdiction that it or the Acquired Assets are or may be subject to taxation by that jurisdiction.
- (b) No Tax liens have been filed and no claims have been asserted or threatened in writing with respect to any Taxes due with respect to the Acquired Assets that have not been paid in full or otherwise settled.
- (c) No audit or administrative or judicial Tax examination or proceeding is pending or is being conducted with respect to Seller or otherwise related to the Acquired Assets. Seller has not received any communication from any taxing authority that such an audit or proceeding is forthcoming. No deficiency for any Taxes has been proposed, or is expected to be proposed, against Seller or otherwise with respect to the Acquired Assets, which deficiency has not been paid in full, and Seller does not know of any basis for the assertion of such a Tax deficiency.
- (d) There are no outstanding agreements, waivers or arrangements extending (or requesting the extension of) the statutory period of limitation applicable to any claim for, or the period for the collection or assessment of, Taxes due from or with respect to Seller or otherwise with respect to the Acquired Assets for any Taxable period. No written rulings or agreements in respect of any Tax are pending or have been issued by or entered into with any relevant Governmental Entity with respect to Seller, or otherwise with respect to the Acquired Assets.

- (e) Seller has never been a member of any affiliated group filing or required to file a consolidated, combined, unitary or other similar Tax Return, in each case, except any such group with respect to which Seller is or was the common parent. Seller (i) is not and never has been a party to or bound by, nor does it have or has it ever had any obligation under, any Tax sharing agreement or similar contract or arrangement and (ii) does not have any Liability for the Taxes of any other person under Treasury Regulation Section 1.1502-6 (or any corresponding provision of state, local or non-U.S. Tax law), as a transferee or successor, by contract, or otherwise.
- (f) Seller has not been nor is it currently in violation (or, with or without notice or lapse of time or both, would be in violation) of any applicable Law or regulation relating to the payment, collection or withholding of Taxes, or the remittance thereof, and all such withholding and payroll Tax requirements required to be complied with by such person up to and including the date hereof have been satisfied.
- (g) None of the Acquired Assets include any stock or other ownership interest in any U.S. or non-U.S. corporation, joint venture, partnership, limited liability company, business trusts, or other entities, any stock of a "controlled foreign corporation" as defined in Section 957 of the Code (or any similar provision of state, local or non-U.S. Law) or "passive foreign investment company" within the meaning of Section 1297 of the Code. None of the Assumed Liabilities is an obligation to make a payment that will not be deductible under Section 280G of the Code.
- (h) For purposes of this <u>Section 2.13</u>, and <u>Article V</u>, any reference to Seller shall be deemed to include any subsidiary of Seller and any person that merged with, was liquidated into, or otherwise was a predecessor of Seller or any such subsidiary.

2.14 Data Privacy and Protection.

- (a) Seller is, and has been in the [***] years prior to the Closing Date, in compliance in all material respects with all Privacy Requirements affecting the Acquired Assets.
- (b) Seller has adopted a written information security program ("<u>WISP</u>") in compliance in all material respects with Privacy Requirements to govern the protection of all Personal Information processed by or on behalf of Seller included in the Acquired Assets. Seller has made available to Buyer (a) Seller's WISP and (b) other applicable security program documents, including Seller's incident response policies, encryption standards and/or other computer security protection policies or procedures, that constitute compliance with Privacy Requirements.
- (c) During the past [***] years, Seller has not received notice in writing regarding any actual, alleged or asserted violation of any Privacy Requirement relating to the Acquired Assets.
- (d) During the past [***] years, to the Knowledge of Seller, no Personal Data relating to the Acquired Assets has been processed by or on behalf of Seller, or transferred by or on behalf of Seller to any third party, in violation of any Privacy Requirement. There are no notices, claims, investigations or proceedings pending, or, to the Knowledge of Seller, threatened, by state or federal agencies, or private parties involving notice or information to individuals that Personal Information processed by or on behalf of Seller relating to the Acquired Assets has been compromised, lost, taken, accessed or misused. Seller has not received any notice regarding any violation of any Privacy Requirement relating to the Acquired Assets.

(e) During the last [***] years, Seller has no Knowledge of any act or attempt, successful or unsuccessful, to gain unauthorized access to, disrupt or misuse any information system of Seller included in the Acquired Assets or information stored on such information system included in the Acquired Assets that required or would be reasonable likely to require notification to any person or Governmental Entity pursuant to any Privacy Requirement, or that has caused or would be reasonably expected to cause a material disruption to the ownership, operation and control of the Acquired Assets.

2.15 Insurance

. Section 2.15(a) of the Disclosure Schedule sets forth a true and correct list of all current policies or binders of fire, theft, casualty, comprehensive general liability, workers compensation and employers liability, directors' and officers' liability, business interruption, environmental, products and professional liability and automobile insurance providing coverage with respect to the Acquired Assets or the Target Employees. Such policies and binders are in full force and effect and are, to the Knowledge of Seller, reasonably adequate for the ownership and operation of the Acquired Assets and are in conformity with the requirements of all Acquired Contracts. Seller is not in material default with respect to any provision contained in any such policy or binder nor has Seller failed to give any notice or present any claim under any such policy or binder in due and timely fashion. All material injuries, incidents or claims relating to participants who are Target Employees in the workers' compensation insurance, medical expense plans and health care plans utilized by Seller that occurred prior to Closing have been reported to the respective insurer in accordance with the respective policy or plan. For the last [***] years, no insurer with respect to Seller's workers compensation insurance has denied Seller coverage on any claims made with respect to Target Employees with respect to such insurance. Section 2.15(b) of the Disclosure Schedule lists any outstanding claims with respect to Seller's workers compensation insurance for any Target Employees who are participants in the workers' compensation insurance utilized by Seller.

2.16 Solvency

Immediately after giving effect to the consummation of the Transactions, Seller shall be solvent and shall (a) be able to pay its debts as they become due; (b) own property that has a fair saleable value greater than the amounts required to pay its debts (including a reasonable estimate of the amount of all contingent liabilities); and (c) have adequate capital to carry on its business (including the performance of its obligations under the Transaction Documents). No transfer of property is being made and no obligation is being incurred in connection with the Transactions with the intent to hinder, delay or defraud either present or future creditors of Seller or the Buyer Parties. In connection with the Transactions, Seller has not incurred, nor plans to incur, debts beyond its ability to pay as they become absolute and matured.

2.17 Brokers' Fees

. Seller has no Liability or obligation to pay any fees or commissions to any broker, finder or agent with respect to the Transactions, other than any such fees or commissions that constitute Excluded Liabilities.

ARTICLE III REPRESENTATIONS AND WARRANTIES OF THE BUYER PARTIES

The Buyer Parties jointly and severally represent and warrant to Seller as of the date hereof that the following representations and warranties are true and correct:

3.1 Organization

. Each Buyer Party is a corporation duly organized, validly existing and in good standing under the laws of the jurisdiction of its formation or organization, as applicable.

3.2 Authority

. Each Buyer Party has all requisite corporate power and authority to execute and deliver this Agreement and the other Transaction Documents to which it is or will be a party and to perform its obligations hereunder and thereunder. The execution and delivery of this Agreement and the other Transaction Documents by the Buyer Parties, the performance by the Buyer Parties of their respective obligations hereunder and thereunder and the consummation by the Buyer Parties of the Transactions have been duly and validly authorized by all necessary corporate and other action on the part of each Buyer Party. This Agreement has been, and the other Transaction Documents will be, upon execution thereof, duly and validly executed and delivered by the applicable Buyer Parties and, assuming this Agreement and the other Transaction Documents constitute or will, upon execution, constitute the valid and binding obligation of Seller, constitutes or will, upon execution, constitute a valid and binding obligation of the Buyer Parties, enforceable against the Buyer Parties in accordance with its terms, except as enforceability may be limited by the Enforceability Exceptions.

3.3 Noncontravention

- . Subject to compliance with applicable Antitrust Laws, neither the execution and delivery of this Agreement by Buyer, nor the consummation by Buyer of the Transactions, will:
- (a) conflict with or violate any provision of the Governing Documents of Buyer or require approval by the Buyer Shareholders pursuant to the terms of the Governing Documents of Buyer or any Law;
- (b) require on the part of Buyer any filing with, or any Permit, authorization, consent or approval of, any Governmental Entity, except for any filing, Permit, authorization, consent or approval which if not obtained or made would not reasonably be expected to result in a Buyer Material Adverse Effect;
- (c) conflict with, result in a breach of, constitute (with or without due notice or lapse of time or both) a default under, result in the acceleration of, create in any party any right to accelerate, terminate, modify or cancel, or require any consent or waiver under, any Contract to which Buyer is a party or by which Buyer is bound or to which any of its assets are subject, except for any conflict, breach, default, acceleration, right to accelerate, termination, modification, cancellation, consent or waiver which would not be material to the business and operations of

Buyer, taken as a whole, and would not be material to the Buyer's ability to consummate the Transactions without delay; or

(d) violate any order, writ, injunction or decree specifically naming, or statute, rule or regulation applicable to, Buyer or any of its properties or assets, except for any violation that would not reasonably be expected to result in a Buyer Material Adverse Effect.

3.4 Capitalization.

- (a) Buyer Parent has (a) 6,423,039,111 ordinary shares in its capital ("Ordinary Shares") and (b) no other shares, including preference shares, issued and outstanding. All of the issued and outstanding Ordinary Shares and other equity interests of Buyer Parent are set forth on Schedule 3.4(a). All of the issued and outstanding shares and other equity interests of Buyer Parent have been duly authorized, are fully paid and non-assessable and not in violation of any purchase option, right of first refusal, preemptive right, subscription right or any similar right under any provision of any applicable Law, the Governing Documents of Buyer or any contract or agreement to which Buyer is a party or by which Buyer or its securities are bound. Buyer does not hold, directly or indirectly, any of its shares or other equity interests in its treasury.
- (b) Schedule 3.4(b) sets forth the beneficial and record owners of all outstanding securities convertible into, or exercisable or exchangeable for, the securities of Buyer Parent (the "Buyer Convertible Securities") (including in each case the issuance date, number and type of shares issuable thereunder, the exercise price, the expiration date and any vesting schedule). Other than as set forth on Schedule 3.4(b), there are no Buyer Convertible Securities, preemptive rights or rights of first refusal or first offer, nor are there any contracts, commitments, arrangements or restrictions to which Buyer Parent or, to the knowledge of Buyer Parent, any of its security holders is a party or otherwise subject or relating to any equity securities of Buyer Parent, whether or not outstanding. There are no outstanding or authorized equity appreciation, phantom equity or similar rights with respect to Buyer Parent. There are no voting trusts, proxies, shareholder agreements or any other agreements or understandings with respect to the voting of Buyer Parent's securities. There are no outstanding contractual obligations of Buyer Parent to repurchase, redeem or otherwise acquire any of its equity interests or securities, nor has Buyer Parent granted any registration rights to any person with respect to its equity securities. All of Buyer Parent's issued and outstanding securities have been granted, offered, sold and issued in compliance with all applicable securities Laws. Except as set forth on Schedule 3.4(b), as a result of the consummation of the Transactions, no equity interests of Buyer Parent are issuable, and no rights in connection with any interests, warrants, rights, options or other securities of Buyer Parent will accelerate or otherwise become triggered (whether as to vesting, exercisability, convertibility or otherwise).

3.5 Broker's Fees

. Buyer has no Liability or obligation to pay any fees or commissions to any broker, finder or agent with respect to the Transactions.

3.6 Litigation

. There are no actions, suits, claims or legal, administrative or arbitratorial proceedings pending against, or, to Buyer's knowledge, threatened against, Buyer that would reasonably be expected to result in a Buyer Material Adverse Effect.

3.7 <u>Buyer Material Adverse Effect.</u>

Since January 1, 2023, there has been no Buyer Material Adverse Effect.

3.8 Solvency

Immediately after giving effect to the consummation of the Transactions, each Buyer Party shall be solvent and shall (a) be able to pay its debts as they become due; (b) own property that has a fair saleable value greater than the amounts required to pay its debts (including a reasonable estimate of the amount of all contingent liabilities); and (c) have adequate capital to carry on its business (including the performance of its obligations under the Transaction Documents). No transfer of property is being made and no obligation is being incurred in connection with the Transactions with the intent to hinder, delay or defraud either present or future creditors of the Buyer Parties or Seller. In connection with the Transactions, the Buyer Parties have not incurred, nor plan to incur, debts beyond their ability to pay as they become absolute and matured.

3.9 No Other Representations or Warranties

. Buyer hereby acknowledges and agrees that, except for the representations and warranties set forth in Article II (in each case as qualified and limited by the Disclosure Schedule), (i) none of Seller or any of its Affiliates, stockholders, directors, officers, employees, agents, representatives or advisors, or any other person, has made or is making (and Seller hereby expressly disclaims) any express or implied representation or warranty (including any warranty of habitability, merchantability or fitness for a particular purpose) with respect to the Acquired Assets or the Assumed Liabilities, including with respect to any information provided or made available to Buyer or any of its Affiliates, shareholders, directors, officers, employees, agents, representatives or advisors, or any other person, or, except as otherwise expressly set forth in this Agreement, had or has any duty or obligation to provide any information to Buyer or any of its Affiliates, shareholders, directors, officers, employees, agents, representatives or advisors, or any other person in connection with this Agreement, the Transactions or otherwise, and (ii) none of Seller or any of its Affiliates, stockholders, directors, officers, employees, agents, representatives or advisors, or any other person, will have or be subject to any Liability or indemnification or other obligation of any kind or nature to Buyer or any of its Affiliates, shareholders, directors, officers, employees, agents, representatives or advisors, or any other person, resulting from the delivery, dissemination or any other distribution to Buyer or any of its Affiliates, shareholders, directors, officers, employees, agents, representatives or advisors, or any other person, or the use by Buyer or any of its Affiliates, shareholders, directors, officers, employees, agents, representatives or advisors, or any other person, of any such information provided or made available to any of them by Seller or any of its Affiliates, stockholders, directors, officers, employees, agents, representatives or advisors, or any other person, including any information, documents, estimates, projections, forecasts or other forward-looking information, business plans or other material provided or made available to Buyer or any of its Affiliates, shareholders, directors, officers, employees, agents, representatives or advisors, or any other person, in "data rooms," confidential

information memoranda, management presentations or otherwise in anticipation or contemplation of the Transactions, and (subject to the express representations and warranties of Seller set forth in <u>Article II</u> (in each case as qualified and limited by the Disclosure Schedule)) none of Buyer or any of its Affiliates, shareholders, directors, officers, employees, agents, representatives or advisors, or any other person, has relied on any such information (including the accuracy or completeness thereof).

ARTICLE IV INDEMNIFICATION

4.1 <u>Indemnification by Seller</u>

- . Subject to the terms and conditions of this <u>Article IV</u>, from and after the Closing, Seller shall indemnify Buyer in respect of, and hold Buyer harmless against, and pay any and all Damages incurred or suffered by Buyer or any Affiliate thereof arising out of, caused by or resulting from, directly or indirectly, any of the following:
 - (a) any inaccuracy in or breach of any representation or warranty of Seller contained in Article II;
- (b) failure to perform any covenant or agreement of Seller contained in this Agreement, the Bill of Sale, the Lease Assignment and Assumption Agreement or the Sublease Agreement; or
 - (c) the Excluded Assets or the Excluded Liabilities.

4.2 Indemnification by the Buyer Parties

- . Subject to the terms and conditions of this <u>Article IV</u>, from and after the Closing, the Buyer Parties shall jointly and severally indemnify Seller in respect of, and hold Seller harmless against, and pay any and all Damages incurred or suffered by Seller or any Affiliate thereof arising out of, caused by or resulting from, directly or indirectly, any of the following:
 - (a) any inaccuracy in or breach of any representation or warranty of the Buyer Parties contained in Article III;
- (b) failure to perform any covenant or agreement of any Buyer Party contained in this Agreement, the Bill of Sale, the Lease Assignment and Assumption Agreement or the Sublease Agreement;
- (c) all Taxes (i) relating to the ownership or operation of the Acquired Assets for any taxable period (or portions thereof) beginning after the Closing and (ii) any Tax for which Buyer is responsible pursuant to <u>Article V</u>;
- (d) any Liabilities arising from the employment of the Continuing Employees by the Buyer Parties or any of their respective Affiliates; or

- (e) (i) the Acquired Assets, to the extent such Damages arise from and after the Closing, and (ii) the Assumed Liabilities.
 - 4.3 <u>Claims for Indemnification; Mitigation</u>.
- (a) Third-Party Claims. All claims for indemnification made under this Agreement resulting from, related to or arising out of a third-party claim (a "Third-Party Claim") against an Indemnified Party shall be made in accordance with the following procedures. A Party or its Affiliate entitled to indemnification under this Article IV (an "Indemnified Party") shall give prompt written notice (the "Claim Notice") to the Party from whom indemnification is sought (the "Indemnifying Party") of the commencement of any action, suit or proceeding relating to a Third-Party Claim for which indemnification may be sought or, if earlier, upon the assertion of any such Third-Party Claim; provided, however, that any failure to give prompt notice shall not relieve an Indemnifying Party of its obligations unless, and then only to the extent that, the Indemnifying Party is prejudiced thereby. The Claim Notice shall contain (i) a reasonably detailed description and, if ascertainable, an estimate of the amount of any Damages incurred by the Indemnified Party and (ii) a statement that the Indemnified Party is entitled to indemnification under this <u>Article IV</u> and an explanation of the basis therefor. In connection with any claim giving rise to indemnity hereunder resulting from or arising out of any Third-Party Claim, the Indemnifying Party, at its sole cost and expense and upon written notice to the Indemnified Party, delivered within ten (10) Business Days of receipt of a Claim Notice, may assume the defense of any such proceeding with counsel reasonably satisfactory to the Indemnified Party. The Party not controlling such defense may participate therein at its own expense; provided that if the Indemnifying Party controls such defense and the Indemnified Party reasonably concludes, based on advice from counsel, that the Indemnifying Party and the Indemnified Party have conflicting interests with respect to such action, suit, proceeding or claim, the reasonable fees and expenses of counsel to the Indemnified Party solely in connection therewith shall be considered "Damages" for purposes of this Agreement; provided, however, that in no event shall the Indemnifying Party be responsible for the fees and expenses of more than one counsel for all Indemnified Parties. The Party controlling such defense shall keep the other Parties advised of the status of such action, suit, proceeding or claim and the defense thereof and shall consider recommendations made by the other Parties with respect thereto. Indemnified Party shall not agree to any settlement of such action, suit, proceeding or claim without the prior written consent of the Indemnifying Party. The Indemnifying Party shall not agree to any settlement of such action, suit, proceeding or claim that does not include a complete release of the Indemnified Party from all Liability with respect thereto without the prior written consent of the Indemnified Party.
- (b) <u>Procedure for Other Claims</u>. An Indemnified Party wishing to assert a claim for indemnification under this <u>Article IV</u> which is not subject to <u>Section 4.3(a)</u> shall deliver to the Indemnifying Party a written notice which contains (i) a reasonably detailed description of, and an estimate of the amount of any Damages incurred by the Indemnified Party (to the extent ascertainable), and (ii) a statement that the Indemnified Party is entitled to indemnification under this <u>Article IV</u> and an explanation of the basis therefor. If such dispute is not resolved within 60 days following the delivery by the Indemnified Party of such notice, the Indemnifying Party and the Indemnified Party shall each have the right to submit such dispute to a court of competent jurisdiction in accordance with the provisions of Section 7.13.

- (c) <u>Determination of Claims</u>. For the purpose of determining the Damages with respect to any breach of a representation, warranty or covenant, but not for determining whether there has been a breach of a representation, warranty or covenant, the representations and warranties set forth in this Agreement shall be considered without regard to any materiality or material adverse effect qualification set forth therein.
- (d) <u>Mitigation</u>. An Indemnified Party shall, and shall cause its Affiliates to, use commercially reasonable efforts to mitigate any Damages upon becoming aware of any event which would reasonably be expected to, or does give rise to, such Damages.

4.4 Survival.

- (a) The representations and warranties of Seller and Buyer set forth in this Agreement shall survive the Closing and the consummation of the Transactions and continue until the [***] month anniversary of the Closing Date, at which time they shall expire. Notwithstanding the foregoing, the representations and warranties of Seller contained in Sections 2.1, 2.2, 2.3(a), 2.5(a), 2.13 and 2.17 (the "Seller Fundamental Representations") and of Buyer contained in Sections 3.1, 3.2, 3.3(a), 3.4 and 3.5 (the "Buyer Fundamental Representations," and together with the Seller Fundamental Representations, the "Fundamental Representations") shall survive the Closing and the consummation of the Transactions until the [***] anniversary of the Closing Date, at which time they shall expire.
- (b) None of the covenants or other agreements contained in this Agreement shall survive the Closing Date other than those which by their terms contemplate performance after the Closing Date, and each such surviving covenant and agreement shall survive the Closing only until the expiration of the term of the undertaking (if any) set forth in such agreement and covenant, at which time they shall expire.
- (c) No Party shall have any Liability of any nature with respect to any representation, warranty, agreement or covenant after the expiration thereof; <u>provided</u>, <u>however</u>, that any valid claim that is properly asserted in writing pursuant to <u>Section 4.3</u> prior to the expiration as provided in <u>Section 4.4(a)</u> or <u>Section 4.4(b)</u> of the representation, warranty, covenant or agreement that is the basis for such claim shall survive until such claim is finally resolved and satisfied but solely for purposes of the resolution thereof.
- (d) It is the express intent of the Parties that, if the applicable survival period for an item as contemplated by this Section 4.4 is shorter than the statute of limitations that would otherwise have been applicable to such item, then, by contract, the applicable statute of limitations with respect to such item shall be reduced to the shortened survival period contemplated hereby. The Parties further acknowledge that the time periods set forth in this Section 4.4 for the assertion of claims under this Agreement are the result of arms'-length negotiation between the Parties and that they intend for the time periods to be enforced as agreed by the Parties.

4.5 Limitations.

(a) Except in cases of Fraud on the part of any Party, from and after the Closing, the remedies of the Indemnified Parties under this <u>Article IV</u> and the other remedies expressly set forth in this Agreement shall be the sole and exclusive remedies of the Indemnified Parties and their

respective Affiliates with respect to claims resulting from any breach of representation or warranty or failure to perform any covenant or agreement contained in this Agreement; provided, however, that this Article IV shall not limit the rights or remedies of any Indemnified Party under, or with respect to the matters contemplated by, any other Transaction Document. Without limiting the generality of the foregoing, in no event shall Buyer or its successors or permitted assigns be entitled to claim or seek recission of the Transactions. Notwithstanding anything to the contrary herein, the representations and warranties of Seller shall not be construed to relate to, and Buyer shall not have any rights to indemnification under this Agreement with respect to claims or Damages arising out of or with respect to, Intellectual Property matters, which such claims or Damages shall be governed exclusively by the License Agreement.

- (b) Notwithstanding anything to the contrary contained in this Agreement, each of the following limitations shall apply to claims for indemnification pursuant to this Agreement; <u>provided</u>, <u>however</u>, that the following limitations shall not apply with respect to Fraud:
 - (i) the aggregate Liability of an Indemnifying Party for all Damages under <u>Section 4.1(a)</u> or <u>4.2(a)</u> (other than for breaches of Fundamental Representations), as applicable, shall not exceed an amount equal to \$[***];
 - (ii) the Indemnifying Party shall be liable under <u>Section 4.1(a)</u> or <u>4.2(a)</u> (other than for breaches of Fundamental Representations) only in the event that the aggregate Damages resulting from claims under <u>Section 4.1(a)</u> or <u>4.2(a)</u>, as applicable, exceed $\{****\}$; and
 - (iii) the aggregate Liability of an Indemnifying Party for all Damages under <u>Section 4.1(a)</u> or <u>4.2(a)</u>, as applicable, shall not exceed an amount equal to \$[***].
- (c) For any Damages for which indemnification under Section 4.1(a) is finally determined and payable by Seller and such claim is not otherwise satisfied by Seller within ten (10) Business Days of the final determination of such claim, such Damages may, at Buyer's election, be satisfied by set-off of such amounts against any unpaid amounts under the Convertible Notes or any other amounts payable to Seller under the License Agreement; provided, that this clause (c) shall in no way be construed to limit Seller's obligation to pay Damages to Buyer in cash. Notwithstanding the anything to the contrary herein, (i) in the event that Seller's aggregate Liability under Section 4.1(a) with respect to breaches of Seller Non-fundamental Representations exceeds \$[***], any Liability of Seller under Section 4.1(a) with respect to such breaches shall be satisfied only by set-off of such amounts against any unpaid amounts under the Convertible Notes, and (ii) in the event that Seller's aggregate Liability under Section 4.1(a) with respect to such breaches of Seller Fundamental Representations exceeds \$[***], any Liability of Seller under Section 4.1(a) with respect to such breaches shall be satisfied only by set-off of such amounts against any unpaid amounts under the Convertible Notes; provided, however, that in the event that the Convertible Notes have been paid or converted into Ordinary Shares, Seller's maximum Liability under Section 4.1(a) shall not exceed (i) [****] plus (ii) the amount of [****], but in no event greater than \$[****].
- (d) The amount of any Damages for which indemnification is provided under this Agreement shall be reduced by any related recoveries under insurance policies or other related third-party payments received by the Indemnified Party or any of its Affiliates, in each case net of

the Indemnified Party's reasonable out-of-pocket costs and expenses of collection and any increases in insurance premiums incurred by the Indemnified Party that result from the applicable insurance claim. An Indemnified Party shall use commercially reasonable efforts to pursue, and to cause its Affiliates to pursue, all insurance or other Third-Party Claims to which it may be entitled in connection with any Damages it incurs. If an Indemnified Party (or an Affiliate) receives any insurance or other third-party payment in connection with any claim for Damages for which it has already received an indemnification payment from the Indemnifying Party, it shall pay to the Indemnifying Party, within ten (10) Business Days of receiving such insurance or other third-party payment, an amount equal to the excess of (i) the amount previously received by the Indemnified Party under this Agreement with respect to such claim plus the amount of the insurance or other third-party payments (in each case net of the Indemnified Party's reasonable out-of-pocket costs and expenses of collection and any increases in insurance premiums incurred by the Indemnified Party that result from the applicable insurance claim), over (ii) the amount of Damages with respect to such claim which the Indemnified Party has become entitled to receive under this Agreement.

4.6 <u>Treatment of Indemnification Payments</u>

. All indemnification payments made under this Agreement shall be treated by the Parties as an adjustment to the Purchase Price for Tax purposes to the extent permitted by Law.

4.7 Security Agreement.

- (a) Each Buyer Party grants and pledges to Seller a continuing security interest in all of the Equipment owned by such Buyer Party and any and all claims, rights and interests in any of such Equipment, all substitutions for, additions, attachments, accessories, accessions and improvements to and replacements, products, proceeds and insurance proceeds of such Equipment used or held for use at the GMP Facility (collectively, the "Collateral") to secure prompt payment and performance by the Buyer Parties of their obligations under this Article IV arising out of, caused by or resulting from, directly or indirectly, the Lease Assignment and Assumption Agreement, the GMP Facility Lease or otherwise with respect to the GMP Facility, including all Damages incurred or suffered by Seller or its Affiliates related thereto (such amounts, the "Secured Damages"). Such security interest constitutes a valid, first-priority security interest in the Collateral and shall remain in effect as such until, and shall terminate automatically on, the date on which any and all Liabilities of Seller under the GMP Facility Lease are fully paid or otherwise terminated as a result of the expiration of the term of such lease or the entry by Buyer into a direct lease with the landlord of the GMP Facility in replacement of the GMP Facility Lease (with a termination of the GMP Facility Lease).
- (b) Each Buyer Party authorizes Seller (or its agents or attorneys) to file at any time financing statements, continuation statements and amendments thereto describing the Collateral, which statements may list specific items of Collateral, contain any other information required by the Uniform Commercial Code as in effect in the State of Nevada for the sufficiency of filing office acceptance of any financing statement, continuation statement or amendment, including whether such Buyer Party is an organization, the type of organization and any organizational

identification number issued to such Buyer Party, if applicable. Any financing statements filed before the date of this Agreement to perfect Seller's security interest in the Collateral were authorized by the Buyer Parties and are hereby ratified. Where Collateral is in possession of a third-party bailee (excluding equipment out for repair or with an employee in the ordinary course of business or in transit between locations), each Buyer Party shall take such steps as Seller reasonably requests for Seller to obtain an acknowledgment, in form and substance reasonably satisfactory to Seller. Each Buyer Party shall take such other actions as Seller in its reasonable discretion may deem necessary or advisable from time to time in order to preserve, perfect and protect the liens and security interests granted in this Agreement and to exercise and enforce its rights and remedies thereunder with respect to the Collateral.

- (c) For any Secured Damages for which indemnification under <u>Section 4.2</u> is finally determined and payable by the Buyer Parties and such claim is not otherwise satisfied by the Buyer Parties within ten (10) Business Days of the final determination of such claim, Seller may, at its election, do any one or more of the following, all of which are authorized by the Buyer Parties:
 - (i) Make such payments and do such acts as Seller considers necessary or reasonable to protect its security interest in the Collateral. Each Buyer Party agrees to assemble the Collateral if Seller so requires and to make the Collateral available to Seller as Seller may designate. Each Buyer Party authorizes Seller to enter the premises where the Collateral is located, to take and maintain possession of the Collateral, or any part of it, and to pay, purchase, contest or compromise any encumbrance, charge or lien which in Seller's determination appears to be prior or superior to its security interest and to pay all expenses incurred in connection therewith.
 - (ii) Ship, reclaim, recover, store, finish, maintain, repair, prepare for sale, advertise for sale and sell (in the manner provided for herein) the Collateral;
 - (iii) Sell the Collateral at either a public or private sale, or both, by way of one or more contracts or transactions, for cash or on terms, in such manner and at such places (including any Buyer Party's premises) as Seller determines is commercially reasonable, and apply any proceeds to the payment of the Buyer Parties' indemnification obligations in whatever manner or order Seller deems appropriate. If notice of intended disposition of any Collateral is required by applicable law, it is agreed that ten (10) days' notice shall constitute reasonable notification. Seller may sell the Collateral without giving any warranties as to the Collateral. Seller may specifically disclaim any warranties of title or the like. This procedure will not be considered adversely to affect the commercial reasonableness of any sale of the Collateral. If Seller sells any of the Collateral upon credit, the Buyer Parties will be credited only with payments actually made by the purchaser, received by Seller and applied to the indebtedness of the purchaser. If the purchaser fails to pay for the Collateral, Seller may resell the Collateral and the Buyer Parties shall be credited with the proceeds of the sale;
 - (iv) credit bid and purchase at any public sale;
 - (v) apply for the appointment of a receiver, trustee, liquidator or conservator of the Collateral, without notice and without regard to the adequacy of the security for the

Buyer Parties' indemnification obligations and without regard to the solvency of any Buyer Party or any other person;

- (vi) demand and receive possession of any Buyer Party's books and records relating to the Collateral; and
- (vii) comply with any applicable state or federal law requirements in connection with a disposition of the Collateral, and such compliance will not be considered adversely to affect the commercial reasonableness of any sale of the Collateral.
- (d) Any deficiency that exists after disposition of the Collateral as provided above will be paid immediately by the Buyer Parties. Seller has no obligation to clean up or otherwise prepare the Collateral for sale. All risk of loss, damage, or destruction of the Collateral shall be borne by the Buyer Parties.

ARTICLE V

TAX MATTERS

5.1 Apportionment of Certain Taxes

. Any real property, personal property, or other similar ad valorem Taxes levied with respect to any of the Acquired Assets for any taxable period that includes but does not end on the Closing Date, whether paid prior to, on or after the Closing Date, shall be apportioned between Seller and Buyer based on the number of days of such taxable period up to and including the Closing Date and the number of days of such taxable period after the Closing Date. Seller shall be responsible for and shall pay the proportionate amount of such Taxes that is attributable to the portion of the taxable period ending on the Closing Date, and Buyer shall be responsible for and pay the proportionate amount of such Taxes that is attributable to the portion of the taxable period beginning and including the day after the Closing Date. Within ninety (90) days after the Closing, Seller and Buyer shall present a statement to the other setting forth the amount of reimbursement to which each is entitled under this Section 5.1 together with such supporting evidence as is reasonably necessary to calculate the amount of such reimbursement. Thereafter, upon receipt of any bill for such Taxes, Buyer or Seller, as applicable, shall notify the other Parties setting forth the amount of reimbursement to which it shall be entitled under this Section 5.1 upon payment of such bill, together with such supporting evidence as is reasonably necessary to calculate the amount of reimbursement. Payment of such reimbursement amount shall be made by the Party owing reimbursement to the Party to which it owed within ten (10) days after delivery of such statement. In the event that Seller or Buyer shall make any payment for which such party is entitled to reimbursement under this Section 5.1, the other Party shall make such reimbursement promptly, but in no event later than [***] days after the presentation of a statement setting forth the amount of reimbursement to which the presenting Party is entitled, along with such supporting evidence as is reasonably necessary to calculate the amount of reimbursement.

5.2 <u>Transfer Taxes</u>

. Seller shall be responsible for the payment of any transfer (including all personal property transfer), sales, use, stamp, conveyance, value added, recording, registration, documentary, excise,

filing and other similar non-income Taxes and administrative fees (including interest and penalties thereon and notary fees) ("<u>Transfer Taxes</u>") arising in connection with the consummation of the Transactions. Notwithstanding the foregoing or anything to the contrary herein, Buyer shall be responsible for the payment of one hundred percent (100%) of all non-U.S. Transfer Taxes, including any Australian goods and services Tax, arising in connection with the consummation of the Transactions. Seller will file all necessary Tax Returns and other documentation with respect to all U.S. federal, state, and local Transfer Taxes, fees and charges and Buyer will file all necessary Tax Returns or other documentation with respect to all non-U.S. Transfer Taxes, fees and charges. To the extent required by Law, each Party will join the other in the execution of any such Tax Returns and other documentation.

5.3 Cooperation

. The Parties agree to furnish or cause to be furnished to each other, upon request, as promptly as practicable, such information and assistance relating to the Acquired Assets as is reasonably necessary for the filing of all Tax Returns, and making of any election related to Taxes, the preparation for any Tax audit by any Governmental Entity, and the prosecution or defense of any claim, suit or proceeding relating to any Tax Return or Tax, including any audit or Tax working papers (to the extent required to secure access to working papers related to any Taxable period beginning prior to Closing, upon request of any third party responsible for the preparation or production of such work papers, Seller and Buyer shall provide customary releases, exculpation and/or indemnification in favor of such third party). The Parties shall cooperate with each other in the conduct of any audit or other proceeding related to Taxes involving the Acquired Assets and shall deliver such documents as are necessary to carry out the intent of this Section 5.3. In addition, Buyer and Seller, with respect to the documents referred to in this Section 5.3, agree to maintain such records for a period of [***] years from the Closing Date and each such Party agrees to afford the other reasonable access to such records during normal business hours. Each Party will notify the other Parties before the proposed destruction of such records prior to such records destruction and give the other Parties a reasonable opportunity to obtain copies of such records prior to their destruction.

ARTICLE VI OTHER AGREEMENTS

- 6.1 Access to Information and Records; Record Retention; Cooperation.
- (a) Access to Information and Records. Subject to compliance with contractual obligations and applicable Laws, following the Closing, each Party shall afford to the other Parties and to the other Parties' Affiliates, authorized accountants, counsel and other designated representatives reasonable access (including using commercially reasonable efforts to give access to third parties possessing information and providing reasonable access to its own employees who are in possession of relevant information) and duplicating rights during normal business hours in a manner so as to not unreasonably interfere with the conduct of business to all Laboratory Records and all other non-privileged records, books, contracts, instruments, documents, correspondence, computer data and other data and information (collectively, "Information") within the possession of (or readily accessible to) such Party or its Affiliates, exclusively relating to the Acquired Assets prior to the Closing, insofar as such access is reasonably required by any other Party. Information

may be requested under this <u>Section 6.1(a)</u> for research and drug development purposes, financial reporting and accounting matters, preparing financial statements, preparing and filing any Tax Returns, prosecuting any claim for refunds, defending any Tax claims or assessment, preparing securities law or exchange filings, prosecuting, defending or settling any litigation, environmental matter or insurance claim, performing this Agreement and the Transactions, and all other proper business purposes.

(b) <u>Access to Personnel</u>. Following the Closing, each Party shall use commercially reasonable efforts to make available to the other Parties, upon written request, such Party's and its Affiliates' officers, directors, employees and agents to the extent that such persons may reasonably be required in connection with any legal, administrative or other proceedings in which the requesting Party may from time to time be involved relating to the Acquired Assets or the Continuing Employees prior to the Closing or for any other matter referred to in <u>Section 6.1(a)</u>.

(c) Retrieval and Use of Excluded Assets.

- (i) For a period of [***] days following the Closing, Seller shall have reasonable access to the Leased Real Property during normal business hours or, with the prior consent of Buyer (not to be unreasonably withheld, conditioned or delayed), at other times outside of normal business hours, in order to utilize, maintain, service, repair or remove from the Leased Real Property any Excluded Assets. To the extent any Excluded Assets remain on the Leased Real Property during such period following the Closing, Buyer shall not, without the prior written consent of Seller, access, use, dispose of, sell, damage or destroy any such Excluded Assets.
- (ii) The rights set forth in this <u>Section 6.1(c)</u> are in furtherance and not in limitation of Seller's license rights included in the Lease Assignment and Assumption Agreement.
- (d) <u>Reimbursement</u>. A Party providing Information or personnel to another Party under <u>Section 6.1(a)</u> or <u>Section 6.1(b)</u> shall be entitled to receive from the recipient, upon the presentation of invoices therefor, payments for such amounts, relating to supplies, disbursements and other out-of-pocket expenses, as may reasonably be incurred in providing such Information or personnel; <u>provided</u>, <u>however</u>, that no such reimbursements shall be required for the salary or cost of fringe benefits or similar expenses pertaining to employees or directors of the providing Party or its Affiliates.
- (e) <u>Retention of Records</u>. Except as otherwise required by Law or agreed to in writing by the Parties, Buyer and Seller shall each (and each shall cause its Affiliates to) use commercially reasonable efforts to preserve copies of all Information in its possession pertaining to the Acquired Assets prior to the Closing until [***]. Notwithstanding the foregoing, in lieu of retaining any specific Information, any Party may offer in writing to the other Parties to deliver copies of such Information to the other Parties and, if such offer is not accepted within 90 days, the offered Information may be disposed of at any time.
- (f) <u>Preparation of Seller Financial Statements</u>. Following the Closing, Buyer shall prepare and provide to Seller and its subsidiaries (and its and their respective auditors, attorneys, financial

advisors, bankers and other consultants and advisors) all information relating to the Acquired Assets reasonably required for Seller and its subsidiaries for preparing financial statements of Seller and its subsidiaries for all fiscal periods that precede or include the Closing Date. In furtherance thereof, Buyer shall use its commercially reasonable efforts to ensure that Seller and its subsidiaries (and its and their respective auditors, attorneys, financial advisors, bankers and other consultants and advisors) will be provided with full unrestricted access to the Acquired Assets, its financial management, including the financial directors involved with the Acquired Assets and any accountant's work papers, and their books, accounts and records and will be able to review the work being carried out in accordance with this Section 6.1(f).

(g) <u>Retained Records</u>. Seller may retain copies of all books, records and other written materials included in the Acquired Assets and any other books, records and written materials included in any electronic data room or that are otherwise in the possession or under the control of Seller relating to the Acquired Assets or the Assumed Liabilities.

6.2 Retained Marks

. From and after the Closing, neither Buyer nor any Affiliate thereof shall have any right to use any Retained Mark, and Buyer shall (and shall cause each of its Affiliates to) cease all use of each Retained Mark. "Retained Marks" means all business names, trade names, trademarks and domain names of Seller or any Affiliate of Seller, any derivative thereof, or any word that is similar in sound or appearance to any of the foregoing or is otherwise confusingly similar thereto and, for the avoidance of doubt, shall include all business names trade names, trademarks and domain names consisting of, or that include, the name "PRECISION BIOSCIENCES."

6.3 Payment of Certain Monies

. In the event that Seller or Buyer (or an Affiliate thereof) inadvertently pays or discharges, after the Closing, any liability of the other Parties, such Party shall be reimbursed for the amount so paid or discharged within 30 days of being presented with written evidence of such payment or discharge. Each Party shall promptly forward to the other Parties all monies received by it or its Affiliates following the Closing with respect to any asset of the other Parties promptly (and in any event within thirty (30) days) after receipt thereof.

6.4 Employee Matters.

(a) Buyer shall, or shall cause an Affiliate to, make an offer of employment to each Target Employee set forth on Schedule 6.4(a) (including all such Target Employees on leave), with such offered employment to be effective as of immediately following the termination of such Target Employee by Seller. Each such Target Employee who accepts such offer of employment no later than the Employee Termination Date and becomes an employee of Buyer, or one of its Affiliates shall be referred to herein as a "Continuing Employee." The initial terms of employment or continued employment for each Continuing Employee shall provide for (i) a position consistent with such Continuing Employee's position as of immediately prior to Closing and on terms consistent with those set forth in this Section 6.4, (ii) a base salary or wage rate that is not less than the base salary or wage rate in effect for such Continuing Employee immediately prior to the Closing, (iii) annual target bonus opportunity and long-term target incentive opportunities that are,

in the aggregate no less favorable than those provided to such Continuing Employee immediately prior to Closing, (iv) severance benefits that are no less favorable than the greater of (A) the severance benefits that would have been applicable to such Continuing Employee under a Target Benefit Plan in effect as of immediately prior to the Closing or (B) the severance benefits applicable to similarly-situated employees of Buyer or its Affiliates, and (v) other employee benefits that are comparable in the aggregate to those provided by Seller to such Continuing Employee immediately prior to the Closing with coverage under, and participation in, Buyer's benefit plans or programs to commence immediately upon a Continuing Employee's commencement of employment with Buyer or an Affiliate of Buyer. Notwithstanding the immediately foregoing subsection (v), the Parties acknowledge that the Continuing Employees will remain eligible for group health coverages under Seller's group health through August 31, 2023 (the "Run-Out Coverage Period"), and Buyer may accordingly satisfy its employee benefit coverage obligations hereunder provided benefits coverages for Continuing Employees begin under Buyer's benefit plans effective September 1, 2023. Subject to applicable Law, nothing herein shall be deemed to alter the at-will status of any Continuing Employee. Seller shall terminate the employment of each Continuing Employee effective as of 11:59 p.m. on the third calendar day following the Closing Date (the "Employee Termination Date"), and Seller shall be responsible for any severance or separation benefits that become due to any Continuing Employee as a result of his or her termination by Seller on the Employee Termination Date.

- (b) Seller shall take, or shall cause to be taken, all actions required to release or cause to be released, each Continuing Employee, effective at or before the Closing, from any contract with, or other obligation to, Seller to the extent (and only to the extent) such contract or obligation imposed by Seller limits or restricts such Continuing Employee from being employed by or providing services to Buyer or any of its Affiliates, or otherwise engaging in the ownership or use of the Acquired Assets on behalf of Buyer or any of its Affiliates.
- (c) From and after the Employee Termination Date, and where applicable with respect to any Employee Benefit Plan or compensation arrangement maintained by or on behalf of Buyer or any Affiliate of Buyer, with respect to each Continuing Employee, and without limiting the employment obligations or terms set forth above in Section 6.4(a), (a) Buyer shall recognize and honor, or cause to be recognized and honored, any service credit accrued or recognized by Seller or any Affiliate thereof as of such date for purposes of determining eligibility and vesting, as well as for purposes of determining the amount of PTO to which a Continuing Employee is entitled (but not for purposes of determining the amount of benefits under any Employee Benefit Plan or compensation arrangement maintained by or on behalf of Buyer or any Affiliate of Buyer), to the same extent and for the same purpose as such service was credited under the corresponding Target Benefit Plan, except to the extent that such credit would result in the duplication of benefits for the same period of service; (b) Buyer shall recognize and honor, or cause to be recognized and honored, PTO that was accrued or recognized by Seller as of such date with respect to such Continuing Employee (or, to the extent honoring PTO in kind is limited or prohibited by Law after commercially reasonable accommodation by Buyer or an Affiliate of Buyer for employee consent or similar procedures, pay each affected Continuing Employee the accrued value of his or her accrued PTO in cash); and (c) Buyer shall use, or cause to be used, all commercially reasonable efforts to cause the applicable group health plan maintained by or on behalf of Buyer or an applicable Affiliate of Buyer to recognize and credit amounts paid by such employees under an applicable Target Benefit Plan towards satisfying deductible expense requirements and

out-of-pocket expense limits during the portion of the calendar year prior to the Employee Termination Date, for purposes of an analogous welfare benefit plan maintained by or on behalf of Buyer or any Affiliate of Buyer in which an employee participates, as if such amounts had been paid in accordance with such Buyer plan for the same plan year; provided that the applicable plan administrator under each such Target Benefit Plan shall have timely provided Buyer with validated information that allows Buyer to recognize and credit, or cause to be recognized and credited, such amounts with respect to each applicable group health plan maintained by Buyer or its Affiliates (including credit for positive account balances within any health flexible spending accounts (FSAs), as applicable). Buyer and Seller agree that, where applicable with respect to any medical or dental benefit plan of Buyer or an Affiliate of Buyer, Buyer shall use, or cause to be used, all commercially reasonable efforts to cause such Buyer plan to waive, with respect to each Continuing Employee, any eligibility requirements, waiting periods, or pre-existing condition exclusion to the extent such items were inapplicable, waived, or satisfied by such employee under the applicable Target Benefit Plan prior to the Closing Date.

- (d) To the extent permitted by Law and Buyer's benefits plans, Buyer shall provide, or cause to be provided, continuation health care coverage to all Continuing Employees (and their respective qualified beneficiaries) who experience (or would be deemed to experience a "qualifying event" with Buyer while covered under Seller's group health or welfare plans during any Run-Out Coverage Period through August 31, 2023) a "qualifying event" after the Employee Termination Date while employed with Buyer or any of its Affiliates, in accordance with and to the extent required under the continuation health care coverage requirements of COBRA. Seller shall make commercially reasonable efforts to cooperate with Buyer with regard to Buyer's commitments under the prior sentence. Except as required to be provided by Buyer pursuant to the first sentence of this subsection (d), Seller shall be responsible for providing continuation coverage and all related notices to the extent required by applicable Law to any Target Employee (and their respective qualified beneficiaries) who experience a "qualifying event" under COBRA on or before the Closing Date and to whom Seller, as of immediately prior to the Closing Date, is (a) providing COBRA coverage or (b) under an obligation to provide such coverage at the election of such individual (or his or her qualified beneficiary) based on a termination of employment on or prior to the Closing Date.
- (e) With respect to each Continuing Employee who participates in the Precision Biosciences, Inc. 401(k) Plan (the "401(k) Plan"), Buyer shall permit (or shall cause its Affiliates to permit) each such Continuing Employee to participate in and make rollover contributions of "eligible rollover distributions" (within the meaning of Section 401(a)(31) of the Code including loans) in an amount equal to the full account balance (including participant plan loans) distributable to such Continuing Employee from the 401(k) Plan to a 401(k) plan maintained by or on behalf of Buyer or its Affiliates no later than September 1, 2023.
- (f) With respect to Continuing Employees eligible to receive a Form W-2 and pursuant to the "Standard Procedure" provided in Section 4 of Revenue Procedure 2004-53, 2004-2 C.B. 320, (a) Buyer and Seller shall report, or cause to be reported, on a predecessor/successor basis as set forth therein; (b) Seller will not be relieved from filing a Form W-2 with respect to any Continuing Employees; and (c) Buyer will undertake to file (or cause to be filed) a Form W-2 for each such Continuing Employee with respect to the portion of the year during which such Continuing Employee is employed by Buyer or an Affiliate of Buyer, excluding the portion of such year that

such Continuing Employee was employed by Seller. Buyer acknowledges and covenants that Buyer or an Affiliate of Buyer will be able to process payroll and all related functions for Continuing Employees, including payroll for the period worked for Buyer between immediately following Employee Termination Date and August 31, 2023, no later than August 31, 2023.

- (g) Buyer and Seller shall comply, and cause their Affiliates to comply, with all applicable Laws relating to the hiring of Target Employees by Buyer or its Affiliates, including all applicable discrimination and leave Laws. Buyer shall provide, or cause to be provided, any required notice under, and otherwise comply with, the WARN Act with respect to employment losses of Continuing Employees that occur after the Closing Date. In accordance with 29 U.S.C. 2101(b)(1), Seller shall provide any required notice under, and otherwise comply with, the WARN Act with respect to employment losses of Target Employees that occur on or prior to the Closing Date, <u>provided</u> that Seller and Buyer acknowledge that in light of Buyer's obligation to make, or cause to be made, offers of employment to all of the Target Employees, with employment with Buyer to be effective immediately following the Employee Termination Date, it is their intent that no such employee will suffer an "employment loss" within the meaning of the WARN Act.
- (h) Nothing contained in this Agreement, whether express or implied, shall (i) be treated as an amendment or modification of any Employee Benefit Plan; (ii) without limiting the generality of Section 7.2, give any person (including any current or former employee or any other individual associated therewith or any Employee Benefit Plan or trustee thereof) other than a Party any right to enforce the provisions of this Section 6.4; (iii) modify the at-will nature of any at-will employee's employment; or (iv) obligate Seller or Buyer or any of their respective Affiliates to (A) maintain any particular Employee Benefit Plan, (B) refrain from amending or terminating any particular Employee Benefit Plan, (C) retain the employment of any particular employee or (D) refrain from changing the terms and conditions of employment.
- (i) Prior to Closing, Seller shall take such actions as may be necessary or appropriate to extend the vesting terms of all restricted stock units of Seller held by any Continuing Employees so that such awards survive and continue to vest while the Continuing Employee remains an employee of Buyer, as if such Continuing Employee had remained employed by Seller. Buyer agrees to notify Seller within five (5) days of the termination of any Continuing Employee.

6.5 GMP Facility Usage

Without limiting Seller's other rights set forth herein and in the other Transaction Documents, Seller shall have the right to use the GMP Facility following the Closing as set forth in <u>Schedule 6.5</u>; provided, that Seller will use commercially reasonable efforts to ensure that such usage does not unduly interfere with Buyer's operation of the GMP Facility.

6.6 <u>Deferred Assets</u>.

As provided in the Transition Services Agreement, following the Closing, Seller shall assist the Buyer in replacing the Shared Contracts. As of the Closing Date, the [***] Equipment shall constitute an Excluded Asset and the [***] Agreements shall constitute Shared Contracts. The Parties acknowledge and agree that, from and after the Closing, Seller will use commercially reasonable efforts to maintain in effect, and (subject to

compliance with the [***] Agreements) provide Buyer the benefits under, the [***] Agreements until such time as Buyer has either assumed the [***] Agreements or entered into replacement agreements directly with [***], and the Parties shall reasonably cooperate and use commercially reasonable efforts to obtain consent from [***] to assign the [***] Agreements to Buyer, including any amendments thereto on which the Parties mutually agree; provided, that (a) in no event shall the foregoing require the payment of money by Seller to [***] or otherwise except as specifically set forth in this Agreement, and (b) Seller shall retain the right to use the [***] Equipment until (i) such consent has been obtained and (ii) Buyer is fully authorized and able to operate and use the [***] Equipment. Notwithstanding anything to the contrary in this Agreement, following the Closing, Seller shall be responsible for paying, and shall pay, to [***] only those Milestone Payments under the [***] License (as in effect on the date hereof) that become payable for the first Indication of "[***]" (i.e., the \$[***], and the \$[***]). Seller will be entitled to use all credits that have accrued under the [***] License prior to the Closing Date when making any such payment, and the Buyer Parties will not use any credit that accrued under the [***] License prior to the Closing Date, including the credits described on Section 2.7(c) of the Disclosure Schedule. Buyer shall be responsible for all other payments that become payable to [***] under or in connection with the [***] License after the Closing Date, and all such amounts shall constitute Assumed Liabilities hereunder. Capitalized terms used in this Section 6.6 but not defined in this Agreement have the meaning given to them in the [***] License.

ARTICLE VII. MISCELLANEOUS

7.1 Press Releases and Announcements

. No Party shall issue (and each Party shall cause its Affiliates not to issue) any press release or public disclosure relating to the subject matter of this Agreement or the Transactions without the prior written approval of the other Parties (such consent not to be unreasonably withheld or delayed), unless that Party is required by applicable Law, rules or regulations of any applicable Governmental Entity or national securities exchange to issue a press release or public disclosure and is not reasonably able to first consult with and obtain consent from, the other Parties within the required timeframe, in which case, the Party must use commercially reasonable efforts (having regard to its obligations under applicable Law, rules or regulations of any applicable Governmental Entity or national securities exchange) to provide the other Parties with a copy of the press release or public disclosure before it is made. Notwithstanding the foregoing, no such approval or consultation shall be required in connection with any press release or public disclosure by a Party or an Affiliate of a Party if the contents of such press release or public disclosure is limited to information that previously has been publicly disclosed in accordance with the terms of this Agreement or another Transaction Document.

7.2 No Third-Party Beneficiaries

. Except as required by applicable Law or otherwise expressly provided herein (including pursuant to $\underline{\text{Article IV}}$), this Agreement shall not confer any rights or remedies upon any person other than the Parties and their respective successors and permitted assigns.

7.3 Action to be Taken by Affiliates

. The Parties shall cause their respective Affiliates to comply with all of the obligations specified in this Agreement or the other Transaction Documents to be performed by such Affiliates.

7.4 Entire Agreement

. This Agreement, the other Transaction Documents and the Confidentiality Agreement constitute the entire agreement between the Parties and supersede any prior statements, understandings, agreements or representations by or between the Parties, whether written or oral, with respect to the subject matter hereof and thereof, and the Parties specifically disclaim reliance on any such prior statements, understandings, agreements or representations to the extent not expressly embodied in this Agreement or the other Transaction Documents. The Exhibits, Schedules and Disclosure Schedule are incorporated herein by reference and made a part hereof. The Confidentiality Agreement shall survive the execution and delivery of this Agreement and shall terminate in its entirety effective upon the Closing.

7.5 Successors and Assigns

. This Agreement shall be binding upon and inure to the benefit of the Parties and their respective successors and permitted assigns. No Party may assign either this Agreement or any of its rights, interests, or obligations hereunder without the prior written approval of the other Parties; provided, however, that for the avoidance of doubt Seller may transfer the Closing Ordinary Shares and its rights in connection therewith in accordance with the provisions of this Agreement and the other Transaction Documents and Seller may assign its rights and delegate its obligations under this Agreement in the event that Seller is permitted to assign its rights or delegate its obligations under the License Agreement.

7.6 Notices

. All notices, requests, demands and other communications under this Agreement shall be in writing and shall be deemed to have been duly given or made as follows: (a) if sent by registered or certified mail return receipt requested, upon receipt, (b) if sent by nationally or internationally recognized overnight air courier, one (1) Business Day after mailing, (c) if sent by email, on the date sent to such recipient if sent during the normal business hours of the recipient, and on the next Business Day if sent after the normal business hours of the recipient, and (d) if otherwise actually personally delivered, when delivered, provided that such notices, requests, demands and other communications are delivered to the address set forth below, or to such other address as the applicable Party shall provide by like notice to the other Parties:

If to Buyer:

Imugene Limited ABN 99 009 179 551 Suite 12.01, Level 12 4-6 Bligh Street Sydney NSW 2000 Copy to:

Locke Lord LLP 2800 Financial Plaza Providence, Rhode Island 02903 Email: Douglas.Gray@lockelord.com Attention: Douglas G. Gray, Partner

37

AUSTRALIA Email: [***]

Attention: Leslie Chong

If to Seller:

Precision Biosciences, Inc. 302 East Pettigrew St. Durham, NC 27701, U.S.A.

Email: [***]

Attention: Cindy Atwell, Chief Business Officer

Copy to:

Smith, Anderson, Blount, Dorsett, Mitchell & Jernigan, L.L.P. Wells Fargo Capitol Center 150 Fayetteville Street, Suite 2300

150 Fayetteville Street, Suite 2300 Raleigh, North Carolina 27601

Email: jtherien@smithlaw.com; harmstrong@smithlaw.com

Attention: John Therien; Heyward Armstrong

7.7 Amendments and Waivers

. The Parties may mutually amend or waive any provision of this Agreement at any time. No amendment or waiver of any provision of this Agreement shall be valid unless the same shall be in writing and signed by each Party. No waiver by any Party of any default, misrepresentation, or breach of warranty or covenant hereunder, whether intentional or not, shall be deemed to extend to any prior or subsequent default, misrepresentation or breach of warranty or covenant hereunder or affect in any way any rights arising by virtue of any prior or subsequent such occurrence.

7.8 Severability

. If, for any reason, any part of this Agreement is adjudicated invalid, unenforceable, or illegal by a court of competent jurisdiction, (a) such adjudication shall not, to the extent feasible, affect or impair, in whole or in part, the validity, enforceability, or legality of any remaining portions of this Agreement, (b) this Agreement shall be construed and enforced as if such invalid, unenforceable or illegal provision had never comprised a part hereof, (c) all remaining portions will remain in full force and effect and shall not be affected by the invalid, unenforceable or illegal provision or by its severance herefrom, and (d) in lieu of such invalid, unenforceable or illegal provision, the Parties shall use reasonable efforts to seek and agree on an alternative valid and enforceable provision that preserves the original purpose and intent of this Agreement.

7.9 Expenses

. Except as otherwise specifically provided to the contrary in this Agreement or the other Transaction Documents, each Party shall bear its own costs and expenses (including legal fees and expenses) incurred in connection with this Agreement and the Transactions.

7.10 Bulk Transfer Laws

. Buyer acknowledges that Seller will not comply with the provisions of the bulk transfer laws of any jurisdiction in connection with the Transactions, and Buyer agrees that it shall not make or permit to be made any filings under or with respect to any such laws in any jurisdiction.

7.11 Specific Performance

. Each Party acknowledges and agrees that the other Parties would be damaged irreparably in the event any of the provisions of this Agreement are not performed in accordance with their specific terms or otherwise are breached. Accordingly, each Party agrees that the other Parties may be entitled to an injunction or injunctions to prevent breaches of the provisions of this Agreement and to enforce specifically this Agreement and the terms and provisions hereof in any action instituted in any court of the United States or any state or jurisdiction thereof having jurisdiction over the Parties and the matter.

7.12 Governing Law

. This Agreement and the other Transaction Documents (except as may otherwise be specifically set forth therein) shall be interpreted and construed in accordance with the Laws of the State of New York. Any and all claims, controversies, and causes of action arising out of or relating to this Agreement and the Transaction Documents (except as may otherwise be specifically set forth therein), whether sounding in contract, tort, or statute, shall be governed by the Laws of the State of New York, including its statutes of limitations, without giving effect to any conflict-of-laws or other rule that would result in the application of the Laws of a different jurisdiction. The United Nations Convention of International Contracts on the Sale of Goods (the Vienna Convention) does not apply to this Agreement or any other Transaction Document.

7.13 Submission to Jurisdiction

. Each Party hereby (i) submits to the exclusive jurisdiction of the United States District Court for the Southern District of New York or, if such court does not have jurisdiction, any state court sitting in the City of New York, New York in any action or proceeding arising out of or relating to this Agreement or the Transaction Documents (except as may otherwise be specifically set forth therein), (ii) agrees that all claims in respect of such action or proceeding may be heard and determined only in any such court, and (iii) agrees not to bring any action or proceeding arising out of or relating to this Agreement or the Transaction Documents (except as may otherwise be specifically set forth therein) in any other court. Each Party waives any defense of inconvenient forum to the maintenance of any action or proceeding so brought and waives any bond, surety or other security that might be required of the other Parties with respect thereto. Any Party may make service on the other Parties by sending or delivering a copy of the process to the Party to be served at the address and in the manner provided for the giving of notices in Section 7.6. Nothing in this Section 7.13, however, shall affect the right of any Party to serve legal process in any other manner permitted by Law.

7.14 WAIVER OF JURY TRIAL

. TO THE EXTENT PERMITTED BY APPLICABLE LAW, EACH PARTY HEREBY IRREVOCABLY WAIVES ALL RIGHTS TO TRIAL BY JURY IN ANY ACTION, PROCEEDING OR COUNTERCLAIM (WHETHER BASED ON CONTRACT, TORT OR OTHERWISE) ARISING OUT OF OR RELATING TO THIS AGREEMENT OR THE TRANSACTIONS OR THE ACTIONS OF ANY PARTY IN THE NEGOTIATION, ADMINISTRATION, PERFORMANCE AND ENFORCEMENT OF THIS AGREEMENT.

7.15 Waiver of Conflicts; Privilege.

- (a) Each of the Parties acknowledges and agrees that Smith, Anderson, Blount, Dorsett, Mitchell & Jernigan, L.L.P. ("Smith Anderson") has acted as counsel to Seller and its Affiliates in connection with the negotiation of this Agreement and the Transactions. Buyer hereby consents and agrees to Smith Anderson representing Seller and any of its Affiliates (collectively, the "Seller Parties") or any director, shareholder, officer or employee of the Seller Parties after the Closing, including with respect to disputes in which the interests of the Seller Parties may be directly adverse to Buyer and its Affiliates. In connection with the foregoing, Buyer hereby irrevocably waives and agrees not to assert any conflict of interest arising from or in connection with Smith Anderson's representation of the Seller Parties prior to and after the Closing. Buyer represents that Buyer's legal counsel has explained and helped Buyer evaluate the implications and risks of waiving the right to assert a future conflict against Smith Anderson and Buyer's consent with respect to this waiver is fully informed.
- (b) Buyer further agrees that all pre-Closing communications subject to any attorney-client privilege, attorney work-product protection or other similar protection for the benefit of the Seller Parties, in any form or format whatsoever between or among Smith Anderson, on one hand, and Seller, or any of its directors, officers, employees or other representatives, on the other hand, to the extent related in any way to the negotiation, documentation and consummation of the Transactions or any dispute arising under this Agreement (collectively, the "<u>Privileged Deal Communications</u>") shall be deemed to be retained and owned collectively by the Seller Parties, shall be controlled by Seller on behalf of the Seller Parties and shall not pass to or be claimed by Buyer.
- (c) Notwithstanding the foregoing, if a dispute arises between Buyer, on the one hand, and a third party other than a Seller Party on the other hand, Buyer may assert the attorney-client privilege, attorney work-product protection or other similar protection for the benefit of the Seller Parties or Buyer to prevent the disclosure of the Privileged Deal Communications to such third party; provided, however, that Buyer may not waive such privilege without the prior written consent of Seller. If Buyer is legally required by order of any Governmental Entity or otherwise to access or obtain a copy of all or a portion of the Privileged Deal Communications following the Closing, Buyer shall as soon as reasonably practicable notify Seller in writing (including by making specific reference to this Section 7.15) so that Seller can seek a protective order, and Buyer agrees to use commercially reasonable efforts, at the Seller Parties' expense, to assist therewith.
- (d) To the extent that files or other materials maintained by Smith Anderson that contain Privileged Deal Communications constitute property of its clients, only the Seller Parties shall hold such property rights and Smith Anderson shall have no duty to reveal or disclose any Privileged Deal Communications contained in such files or other materials or any other Privileged Deal Communications by reason of any attorney-client relationship between Smith Anderson, on the one hand, and the Buyer, on the other hand.
- (e) Buyer agrees that it will not use any Privileged Deal Communications against the Seller Parties in connection with any dispute between Seller, on the one hand, and Buyer, on the other hand, with respect to this Agreement or the Transactions.

(f) This <u>Section 7.15</u> is intended for the benefit of, and shall be enforceable by, Seller and Smith Anderson. This <u>Section 7.15</u> shall be irrevocable, and no term of this <u>Section 7.15</u> may be amended, waived or modified without the prior written consent of Seller and Smith Anderson.

7.16 Construction; Certain Definitions.

- (a) Except where expressly stated otherwise in this Agreement, the following rules of interpretation apply to this Agreement: (i) "either" and "or" are not exclusive and "include," "includes" and "including" are not limiting; (ii) "hereof," "hereto," "hereby," "herein" and "hereunder" and words of similar import when used in this Agreement refer to this Agreement as a whole and not to any particular provision of this Agreement; (iii) "date hereof" or "date of this Agreement" refers to the date set forth in the initial caption of this Agreement; (iv) "extent" in the phrase "to the extent" means the degree to which a subject or other thing extends, and such phrase does not mean simply "if"; (v) definitions contained in this Agreement are applicable to the singular as well as the plural forms of such terms; (vi) references to a person are also to its successors and permitted assigns; (vii) references to an "Article," "Section," "Exhibit," or "Schedule" refer to an Article or Section of, or an Exhibit or Schedule to, this Agreement; (viii) references to "\$" or otherwise to dollar amounts refer to the lawful currency of the United States; (ix) references to "person" shall be construed broadly and include any individual, corporation, partnership, joint venture, limited liability company, association, joint-stock company, trust, unincorporated organization or Governmental Entity, and (x) except as otherwise expressly set forth herein, references to hours, time of day, days or date mean the local time, days or date in Durham, North Carolina, United States of America. The language used in this Agreement shall be deemed to be the language chosen by the Parties hereto to express their mutual intent, and no rule of strict construction shall be applied against any Party. The section headings contained in this Agreement are inserted for convenience only and shall not affect in any way the meaning or interpretation of this Agreement.
 - (b) As used in this Agreement, the following capitalized terms shall have the respective meanings set forth below:
 - "401(k) Plan" has the meaning set forth in Section 6.4(e).
 - "Acquired Assets" has the meaning set forth in Section 1.1(a).
 - "Acquired Contracts" has the meaning set forth in Section 1.1(a)(i).

"Affiliate" means, with respect to any person, any entity that, at the relevant time (whether as of the Closing Date or thereafter), directly or indirectly through one or more intermediaries, controls, is controlled by, or is under common control with such person. For purposes of this definition of Affiliate, "control" means (i) to possess, directly or indirectly, the power to direct or cause the direction of the management or policies of an entity, whether through ownership of voting securities or by contract relating to voting rights or corporate governance; or (ii) direct or indirect ownership of fifty percent (50%) (or such lesser percentage that is the maximum allowed to be owned by a foreign entity in a particular jurisdiction) or more of the voting share capital or other equity interest in such entity. Notwithstanding anything to the contrary in this Agreement or the other Transaction Document. Buyer and Seller shall not be considered Affiliates of each other.

- "Agreement" has the meaning set forth in the Preamble.
- "Allocation Methodology" has the meaning set forth in Section 1.2(b).
- "Allocation Schedule" has the meaning set forth in Section 1.2(b).
- "Antitrust Division" means the Antitrust Division of the United States Department of Justice.
- "Antitrust Laws" means the Hart-Scott-Rodino Antitrust Improvement Act of 1976 and any applicable foreign or supranational antitrust and competition laws.
 - "Assumed Liabilities" has the meaning set forth in Section 1.1(c).
- "<u>Australian Royalty Withholding Tax</u>" means an amount required to be deducted or withheld from a royalty under Division 11A of Part III of the Income Tax Assessment Act 1936 (Cth) or Subdivision 12-F of Schedule 1 to the Taxation Administration Act 1953 (Cth).
 - "Bill of Sale and Assignment and Assumption Agreement" has the meaning set forth in Section 1.3(b)(i).
- "Business Day" shall be any day other than (i) a Saturday or Sunday or (ii) a day on which banking institutions located in New York, New York or Sydney, Australia are permitted or required by law, executive order or governmental decree to remain closed.
 - "Buyer" has the meaning set forth in the Preamble.
 - "Buyer Convertible Securities" has the meaning set forth in Section 3.4(b).
 - "Buyer Fundamental Representations" has the meaning set forth in Section 4.4(a).
- "<u>Buyer Material Adverse Effect</u>" means any event, change, effect, development or circumstance that (i) has a material adverse effect on the business, financial condition or results of operations of Buyer, taken as a whole, or (ii) would reasonably be expected to prevent, or materially impair or delay, the ability of Buyer to consummate the Transactions.
 - "Buyer Shareholders" means holders of issued and outstanding ordinary shares of Buyer.
 - "Claim Notice" has the meaning set forth in Section 4.3(a).
 - "Closing" has the meaning set forth in Section 1.3(a).
 - "Closing Cash Consideration" has the meaning set forth in Section 1.2(a).
 - "Closing Date" has the meaning set forth in Section 1.3(a).
 - "Code" has the meaning set forth in Section 1.2(b).
 - "Collateral" has the meaning set forth in Section 4.7(a).

- "Confidentiality Agreement" means the Mutual Confidentiality Agreement dated June 19, 2023, previously entered into between Buyer Parent and Seller.
 - "Continuing Employees" has the meaning set forth in Section 6.4(a).
- "Contract" means any written or oral agreement, undertaking, contract, lease, sublease, license, indenture, mortgage, instrument, guaranty, loan or credit agreement, note, bond, or other arrangement, understanding, permission or commitment that, in each case, is legally-binding.
 - "Convertible Note Subscription Deed" has the meaning set forth in Section 1.2(a).
 - "Convertible Notes" has the meaning set forth in Section 1.2(a).
- "<u>Damages</u>" means any debts, losses, costs, damages, liabilities, expenses, fines, fees, penalties, deficiencies, obligations, actions, demands, judgments and settlements, including reasonable and documented out-of-pocket legal fees and expenses, interest, court costs, costs of investigators, reasonable and documented out-of-pocket fees and expenses of accountants, financial advisors and other experts, and other reasonable and documented out-of-pocket expenses of investigation or litigation, in each case whether or not asserted by third parties or incurred or sustained in the absence of Third-Party Claims, but excluding lost profits, diminution in value, speculative, exemplary, special or punitive damages (except, in each case, to the extent paid to a third party in connection with a Third-Party Claim).
 - "<u>Deferred Consent</u>" has the meaning set forth in <u>Section 1.4</u>.
 - "<u>Deferred Item</u>" has the meaning set forth in <u>Section 1.4</u>.
 - "Disclosure Schedule" has the meaning set forth in Article II.
 - "EEOC" has the meaning set forth in Section 2.9(c).
 - "EMA" means the European Medicines Agency or any successor agency thereto.
 - "Employee Benefit Plan" has the meaning set forth in Section 2.10(a).
 - "Employee Termination Date" has the meaning set forth in Section 6.4(a).
 - "Enforceability Exceptions" has the meaning set forth in Section 2.2.
- "Environmental Laws" means all applicable federal, state and local statutes or laws, judgments, orders, regulations, licenses, Permits, rules, ordinances, and principles of common law relating to pollution or the protection of the environment, including the Federal Water Pollution Control Act (33 U.S.C. § 1251 et. seq.), Resource Conservation and Recovery Act (42 U.S.C. § 6901 et. seq.), Safe Drinking Water Act (42 U.S.C. § 3000(f) et. seq.), Toxic Substances Control Act (15 U.S.C. § 2601 et. seq.), Clean Air Act (42 U.S.C. § 7401 et. seq.), Comprehensive Environmental Response, Compensation and Liability Act (42 U.S.C. § 9601 et. seq.), and other similar foreign, state and local statutes and common law.

- "Environmental Permits" has the meaning set forth in Section 2.12(a).
- "Equipment" has the meaning set forth in Section 1.1(a)(iii).
- "ERISA" has the meaning set forth in Section 2.10(a).
- "ERISA Affiliate" has the meaning set forth in Section 2.10(a).
- "European Union" means, collectively, the European Union as a legal entity and the countries that are officially recognized as member states of the European Union, in each case as of the date of this Agreement.
 - "Excluded Assets" has the meaning set forth in Section 1.1(b).
 - "Excluded Liabilities" has the meaning set forth in Section 1.1(d).
 - "FDA" means the United States Food and Drug Administration or any successor agency thereto.
 - "FDCA" means the United States Federal Food, Drug, and Cosmetic Act.
- "Fraud" means actual and intentional fraud with the intent to deceive and mislead, all as defined under the laws of the State of New York, with respect to the making of the representations or warranties contained in Article II and Article III. For the avoidance of doubt, "Fraud" shall not include any claim for equitable fraud, constructive fraud, promissory fraud, unfair dealings fraud, fraud by reckless or negligent misrepresentations or any tort based on negligence or recklessness.
 - "Fundamental Representations" has the meaning set forth in Section 4.4(a).
 - "GAAP" means United States generally accepted accounting principles.
 - "GMP Assets" has the meaning set forth in Section 1.1(a)(ii).
 - "GMP Facility" has the meaning set forth in Section 1.3(b)(ii).
- "GMP Facility Lease" means that certain Lease dated October 2, 2018, as amended by that certain First Amendment to Lease dated December 23, 2019, as further amended by that certain Second Amendment to Lease dated March 13, 2020, and as further amended by that certain Third Amendment to Lease dated June 15, 2020, as further amended, for certain premises containing approximately 33,828 rentable square feet on the first (1st) floor located in the building known as Biopoint Innovation Labs located at 20 TW Alexander Drive, Research Triangle Park, North Carolina 27709, which is included as an Acquired Contract and is being assigned to Buyer pursuant to the Lease Assignment and Assumption Agreement.
- "Governing Documents" means, (a) any certificate or articles of incorporation, constitution, certificate of formation, articles of organization, bylaws or limited liability company agreement, (b) any documents comparable to those described in clause (a) as may be applicable pursuant to any Law and (c) any amendment or modification to any of the foregoing.

"Governmental Entity" has the meaning set forth in Section 2.3(b).

"Guaranty Agreement" has the meaning set forth in Section 1.3(b)(vii).

"Hazardous Materials" means any chemical, substance, material, controlled substance, object, condition, waste, living organisms or combination thereof which is or may be hazardous to the environment due to its radioactivity, ignitability, corrosiveness, reactivity, explosivity, toxicity, carcinogenicity, mutagenicity, phytotoxicity, infectiousness or other harmful or potentially harmful properties or effects, including petroleum hydrocarbons and petroleum products, lead, asbestos, radon, polychlorinated biphenyls, and all of those chemicals, substances, materials, controlled substances, objects, conditions, wastes, living organisms or combinations thereof which are now listed, defined or regulated in any matter by any Environmental Law based upon such properties or effects.

"Indebtedness" means, with respect to any person at any date, any amounts owed, without duplication, in respect of (a) obligations of such person for borrowed money (excluding any trade payables, accounts payable and any other current liabilities), (b) obligations of such person in respect of letters of credit to the extent drawn, (c) obligations of such person evidenced by bonds, debentures, notes or similar instruments (other than performance, surety and appeal bonds in respect of which such person's liability remains contingent), (d) finance lease obligations of such person (as accounted for in accordance with GAAP), and (e) accrued interest, penalties or fees related to any of the foregoing, including any prepayment premiums, penalties and any other fees or expenses paid or payable to satisfy such Indebtedness. For the avoidance of doubt, Indebtedness does not include any Liabilities arising out of, relating to or in connection with the Assigned Contracts, including the GMP Facility Lease.

"Indemnified Party" has the meaning set forth in Section 4.3(a).

"Indemnifying Party" has the meaning set forth in Section 4.3(a).

"Independent Accountant" means PricewaterhouseCoopers LLP, or if such firm does not accept such appointment, such other independent firm mutually agreed upon by Buyer and Seller.

"Information" has the meaning set forth in Section 6.1(a).

"Intellectual Property" means all Intellectual Property rights and other proprietary rights in any jurisdiction throughout the world, including (a) patents; (b) inventions, trade secrets, know-how and other confidential or proprietary Information; (c) copyrights and copyright applications, copyrightable works, moral rights, rights of paternity or integrity or similar rights; (d) trademarks; (e) registrations and applications for any of the foregoing; (f) data, results and reports related to any clinical trial, whether published or unpublished; (g) any additions, advances, changes, derivatives, improvements, enhancements, refinements or modifications made to any of the foregoing; and (h) all other intellectual property, including all applications for (and rights to apply for and be granted) renewals or extensions of, and rights to claim priority from, such rights.

"Inventory" has the meaning set forth in Section 1.1(a)(iv).

"Knowledge of Seller", "Seller's Knowledge" and words or phrases of similar meaning mean the actual knowledge on the date hereof of the persons listed on Schedule 7.16(b)-1, upon reasonable inquiry, and shall not refer to the knowledge of any other person.

"Laboratory Assets" has the meaning set forth in Section 1.1(a)(iii).

"Laboratory Facilities" has the meaning set forth in Section 1.3(b)(ii).

"<u>Law</u>" means any statute, code, directive, ordinance, rule, regulation, award, binding and enforceable guideline, binding and enforceable written policy, or rule of common law, enacted or promulgated, or an order issued or rendered, in each case, by any Governmental Entity.

"Lease Assignment and Assumption Agreement" has the meaning set forth in Section 1.3(b)(ii).

"Leased Real Property" has the meaning set forth in Section 1.3(b)(ii).

"Leases" has the meaning set forth in Section 2.6.

"Liabilities" means liabilities, obligations and commitments.

"License Agreement" has the meaning set forth in Section 1.3(b)(iii).

"[***]" has the meaning set forth in Section 2.7(c).

"[***] Agreements" means the [***] License and any related quality or similar agreements with respect thereto.

"[***] <u>Equipment</u>" means all Equipment licensed to Seller pursuant to the [***] Agreements.

"[***] <u>License</u>" means that certain License Agreement by and between Seller, and [***], dated November 12, 2018, as amended.

"Ordinary Shares" has the meaning set forth in Section 3.4(a).

"<u>Pandemic</u>" means any outbreaks, epidemics or pandemics relating to SARS-CoV-2 or COVID-19, or any evolutions or mutations of thereof, or any other viruses (including influenza), and the governmental and other responses thereto.

"Pandemic Measures" means any quarantine, "shelter in place", "stay at home", workforce reduction, social distancing, shut down, closure, sequester or other directives, guidelines or recommendations promulgated by any Governmental Entity, including the Centers for Disease Control and Prevention and the World Health Organization, in each case, in connection with or in response to the Pandemic.

"Party" and "Parties" has the meaning set forth in the Preamble.

- "<u>Permits</u>" means all authorizations, licenses, permits, franchises, privileges, certificates, Regulatory Approvals and other authorizations of any Governmental Entity.
- "<u>Personal Data</u>" or "<u>Personal Information</u>" shall have the meaning of such term or like terms set forth in each of the applicable U.S. Privacy Laws that describes, covers or defines data that identifies or can be used to identify individuals, including that in the California Consumer Privacy Act of 2018, as amended.
- "<u>Privacy Requirements</u>" shall mean all applicable U.S. Privacy Laws and the requirements of contracts to which Seller is a party, including the Payment Card Industry Data Security Standard.
 - "Privileged Deal Communications" has the meaning set forth in Section 7.15(b).
 - "Purchase Price" has the meaning set forth in Section 1.2(a).
 - "Registration Rights Agreement" has the meaning set forth in Section 1.3(b)(vi).
- "Regulatory Approvals" means all Permits necessary for the manufacturing, use, storage, import, export, transport, marketing, distribution or sale of any Product in a country or regulatory jurisdiction, but excluding any pricing or reimbursement approval.
- "Regulatory Authority" means the FDA or any health regulatory authority in the European Union that is a counterpart to the FDA and holds responsibility for granting Regulatory Approvals for any Product in the European Union, including the EMA.
 - "Residuals" has the meaning set forth in Section 1.1(b).
 - "Retained Marks" has the meaning set forth in Section 6.2.
 - "Run-Out Coverage Period" has the meaning set forth in Section 6.4(a).
 - "Security Interest" has the meaning set forth in Section 2.5.
 - "Seller" has the meaning set forth in the Preamble.
 - "Seller Fundamental Representations" has the meaning set forth in Section 4.4(a).
- "Seller Non-fundamental Representations" means the representations and warranties of Seller set forth in <u>Article II</u>, except for the Seller Fundamental Representations.
 - "Seller Parties" has the meaning set forth in Section 7.15(a).
 - "Shared Contracts" means the Contracts set forth on Schedule 7.16(b)-2.
 - "Smith Anderson" has the meaning set forth in Section 7.15(a).
 - "Sublease Agreement" has the meaning set forth in Section 1.3(b)(ii).

"Target Benefit Plans" has the meaning set forth in Section 2.10(a).

"Target Employees" means the employees of Seller set forth on Schedule 6.4(a).

"Target Material Adverse Effect" has the meaning set forth in Section 2.1.

"<u>Tax Returns</u>" means any return, declaration, report, claim for refund, information return, or statement relating to Taxes, and any schedule, attachment, or amendment thereto, including any consolidated, combined or unitary return or other document (including any related or supporting information), filed or required to be filed by any taxing authority in connection with the determination, assessment, collection, imposition, payment, refund or credit of any federal, state, local or foreign Tax or the administration of the Laws relating to any Tax.

"Taxes" means any and all federal, state, local or non-US taxes, including charges, fees, levies, deficiencies other assessments of whatever kind or nature including all net income, alternative or add-on minimum, gross income, profits, gross receipts, excise, real or personal property, sales, ad valorem, paid-up capital, lease, withholding, social security, retirement, employment, unemployment, estimated, severance, stamp, occupation, environmental, windfall profits, use, service, net worth, premium, escheat, abandoned property, payroll, franchise, license, gains, customs, transfer, recording and other taxes, customs duty, fees assessments, deposits or charges of any kind whatsoever, imposed by any taxing authority, including any Liability for Tax incurred or borne, pursuant to Treasury Regulation Section 1.1502-6 (or any similar provision of state, local or non-United Statement Law), as a transferee or successor, or by contract, or otherwise, together with any interest, penalties or additions to Tax relating thereto.

"Third-Party Claim" has the meaning set forth in Section 4.3(a).

"Transaction Costs" shall mean any costs and expenses incurred by Seller in connection with the sale of the Acquired Assets or the process leading up to such sale, including, the preparation of this Agreement and the Transaction Documents, including all fees and expenses of legal counsel, all compensation in the nature of retention bonuses and any D&O insurance not already paid for as of the date hereof. Notwithstanding anything to the contrary in the foregoing, Transaction Costs do not include any costs or expenses incurred by, or that are otherwise the responsibility of, the Buyer Parties under or in connection with this Agreement or any other Transaction Document.

"<u>Transaction Documents</u>" means this Agreement, the License Agreement, the Lease Assignment and Assumption Agreement, the Convertible Note Subscription Deed, the Note Deed Poll, the Convertible Notes, the Transition Services Agreement, the Bill of Sale and Assignment and Assumption Agreement, the Guaranty Agreement and any other agreement, certificate or instrument executed or delivered in accordance with, in connection with or required by this Agreement.

"Transactions" means the transactions contemplated by this Agreement.

"<u>Transfer Taxes</u>" has the meaning set forth in <u>Section 5.2</u>.

"Transition Services Agreement" has the meaning set forth in Section 1.3(b)(iv).

"<u>U.S. Privacy Laws</u>" shall mean all rules, regulations, codes, orders, decrees, guidelines, and rulings thereunder of any federal, state, regional, county, city, municipal or local government of the U.S. or any department, agency, bureau or other administrative or regulatory body obtaining authority from any of the foregoing that relate to privacy, data protection or data transfer issues, including all implementing laws, ordinances, regulations, or guidelines, including the Health Insurance Portability and Accountability Act (HIPAA) of 1996, as amended; the Health Information Technology for Economic and Clinical Health (HITECH) Act; and all applicable state privacy, security, data protection and destruction, and data breach notification statutes and regulations, including the California Consumer Privacy Act of 2018, as amended.

"WISP" has the meaning set forth in Section 2.14(b).

7.17 Counterparts; Electronic Signatures

. This Agreement may be executed in any number of counterparts, each of which is deemed an original, but all of which together constitute one instrument. This Agreement may be executed and delivered electronically and upon such delivery such electronic signature will be deemed to have the same effect as if the original signature had been delivered to the other Parties.

7.18 Time is of the Essence

. Time is of the essence of this Agreement and the performance of the Parties' respective rights and obligations hereunder.

[Remainder of page intentionally left blank]

IN WITNESS WHEREOF, the Parties have executed this Agreement as of the date first above written.

SELLER:

PRECISION BIOSCIENCES, INC.

By: <u>/s/ Michael Amoroso</u> Print Name: Michael Amoroso

Print Title: President & Chief Executive Officer

BUYER:

IMUGENE (USA) INC.

By: <u>/s/ Paul Hopper</u> Print Name: Paul Hopper Print Title: President

BUYER PARENT:

IMUGENE LIMITED

By: <u>/s/ Leslie Chong</u> Print Name: Leslie Chong

Print Title: Chief Executive Officer

[Signature Page to Asset Purchase Agreement]

IF "DOCVARIABLE "SWDocIDLocation" " = "1" "DOCPROPERTY "SWDocID" 134111405v.14" ""

Certain information marked as [***] has been excluded from this exhibit because it is both not material and is the type
that the registrant treats as private or confidential.

LICENSE AGREEMENT

by and between

IMUGENE (USA) INC.

and

PRECISION BIOSCIENCES, INC.

LICENSE AGREEMENT

This License Agreement ("Agreement") is entered into as of August 15, 2023 (the "Effective Date"), by and between Precision BioSciences, Inc., a corporation organized and existing under the laws of Delaware, having an address at 302 East Pettigrew St., Suite A-100, Durham, North Carolina 27701, U.S.A. ("Precision"), and Imugene (USA) Inc., a corporation organized and existing under the laws of Nevada, with its principal business office located at Suite 200, 701 South Carson St, Carson City, Nevada 89701, U.S.A. ("Imugene"). Imugene and Precision are each hereafter referred to individually as a "Party" and together as the "Parties."

WHEREAS, Precision is a U.S., Nasdaq-listed, genome-editing and cell therapy company, which leverages its proprietary ARCUS Technology (as defined below) that is based on I-CREI derived engineered meganucleases and cell therapy platform to develop, manufacture, and commercialize allogeneic CAR-T (as defined below) products and for *in vivo* gene editing for the treatment of genetic disease;

WHEREAS, Imugene is the U.S. subsidiary of Imugene, Ltd., a corporation organized and existing under the laws of Australia, with its principal business office located at ABN 99 009 179 551, Suite 12.01, Level 12, 4-6 Bligh Street, Sydney NSW 2000, AUSTRALIA ("**Parent**"); Parent is an Australian, ASX-listed, oncology company focused on development of cancer therapeutics

WHEREAS, Imugene desires to obtain from Precision, and Precision desires to grant to Imugene, certain exclusive and non-exclusive license rights to develop, manufacture, and commercialize cell therapy products based on ARCUS Technology for treatment of cancer, including Precision's current clinical candidate product known as "azer-cel," subject to the terms and conditions of this Agreement; and

WHEREAS, in connection with the above and the Parties entering into this Agreement, Precision, Parent and Imugene have entered into that certain Asset Purchase Agreement dated August 15, 2023, pursuant to which Precision is selling and transferring certain cell therapy business assets to Imugene (the "Asset Purchase Agreement"), and as of the date hereof, Precision, Parent and Imugene are entering into the other Transaction Documents (as defined below).

Now, Therefore, in consideration of the mutual covenants contained herein, and for other good and valuable consideration, the receipt and adequacy of which are hereby acknowledged, the Parties hereby agree as follows:

ARTICLE 1

DEFINITIONS

Capitalized terms used in this Agreement and the Schedules and Exhibits hereto shall have the following meanings (or as defined elsewhere in this Agreement):

- 1.1"Acquirer" has the meaning set forth in the definition of "Change of Control."
- 1.2" *Active Ingredient*" means, with respect to a Combination Product, an active therapeutic ingredient having a different Target or mode of action, or which is otherwise treated

1

or designated by the applicable Regulatory Authority as a separate active ingredient, than the applicable Licensed Product.

- 1.3"*Additional Product*" means any CAR-T product, other than an Existing Product, that is Directed to solely an Included Target and created by Imugene through *ex vivo* gene editing using the Licensed ARCUS Nuclease, including any preparation, formulation, dosage, packaging or method of administration thereof.
 - 1.4"ADSs" has the meaning set forth in the Registration Rights Agreement.
- 1.5"Affiliate" means, with respect to any Person, any entity that, at the relevant time (whether as of the Effective Date or thereafter), directly or indirectly through one or more intermediaries, controls, is controlled by, or is under common control with such Person, for so long as such control exists. As used in this Section 1.5, "control" means: (a) to possess, directly or indirectly, the power to direct or cause the direction of the management or policies of an entity, whether through ownership of voting securities or by contract relating to voting rights or corporate governance; or (b) direct or indirect ownership of fifty percent (50%) (or such lesser percentage that is the maximum allowed to be owned by a foreign entity in a particular jurisdiction) or more of the voting share capital or other equity interest in such entity. Notwithstanding anything to the contrary in this Agreement or the other Transaction Documents, Precision, on the one hand, and Imugene and Parent, on the other hand, shall not be considered Affiliates of each other.
 - 1.6" Agreement" has the meaning set forth in the Preamble.
- 1.7"*Applicable Laws*" means the applicable provisions of any and all federal, national, supranational, regional, state and local laws, treaties, statutes, rules, regulations, guidelines or requirements, administrative codes, guidance, ordinances, judgments, decrees, directives, injunctions, orders, or permits of or from any court, arbitrator, Regulatory Authority, Governmental Authority, taxing authority, national securities exchange or exchange listing organization having jurisdiction over or related to the relevant subject item that may be in effect from time to time during the Term.
- 1.8" ARCUS Nuclease" means any fully synthetic nuclease derived from a homing endonuclease and made using the ARCUS Technology.
 - 1.9"ARCUS Regulatory Matters" has the meaning set forth in Section 5.1.4.
 - 1.10"ARCUS Research Tool" means the research tools set forth on Schedule 1.10.
- 1.11"*ARCUS Technology*" means Precision's proprietary genome editing platform known as ARCUS™, relating to the design, creation, selection, development, optimization and delivery of fully synthetic enzymes derived from homing endonucleases, including ARCUS Research Tools, and including any modifications or improvements to the foregoing.
 - 1.12" Asset Purchase Agreement" has the meaning set forth in the Preamble.
- 1.13"ASX" means ASX Limited (ABN 98 008 624 691) or, where the context requires, the securities market which it operates.

- 1.14" *Australian Corporations Act*" means *Corporations Act* 2001 (Cth).
- 1.15"*Australian Ipso Facto Stay*" means any limitation on the enforcement of rights or self-executing provisions prescribed from time to time in the Australian Corporations Act that are exercisable because a counterparty becomes subject to a relevant Insolvency Event.
- 1.16"*Australian Royalty Withholding Tax*" means an amount required to be deducted or withheld from a royalty under Division 11A of Part III of the Income Tax Assessment Act 1936 (Cth) or Subdivision 12-F of Schedule 1 to the Taxation Administration Act 1953 (Cth).
 - 1.17" Background IP" means Imugene Background IP or Precision Background IP, as applicable.
 - 1.18" *Bayh-Dole Act*" has the meaning set forth in Section 11.2.6.
- 1.19"**BLA**" means a Biologic License Application (as more fully described in U.S. 21 C.F.R. Part 601.20 or its successor regulation), as may be amended from time to time, or any analogous application or submission with any Regulatory Authority outside of the United States.
- 1.20"*Business Day*" means any day, other than any Saturday, Sunday, or any day that banks are authorized or required to be closed in Durham, North Carolina, U.S. or Sydney, Australia.
- 1.21"*Calendar Quarter*" means each respective period of three (3) consecutive months ending on March 31, June 30, September 30, and December 31 of any Calendar Year.
- 1.22"*Calendar Year*" means each respective period of twelve (12) consecutive months commencing on January 1 and ending on December 31.
 - 1.23"*CAR-T*" means human T cells genetically engineered *ex vivo* with Chimeric Antigen Receptor(s).
 - 1.24"CD19" means B-lymphocyte antigen CD19.
 - 1.25"*Cellectis Agreement*" has the meaning set forth in Section 8.6.2.
 - 1.26"*Cellectis Patents*" has the meaning set forth in Section 8.6.2.
 - 1.27"*Cellectis S.A.*" has the meaning set forth in Section 8.6.2.
- 1.28"Change of Control" means, with respect to a Person: (a) the acquisition by a person or group (each as used in this definition uncapitalized, such terms have the meanings specified in Section 13(d) of the Securities Exchange Act of 1934, as amended, and Rule 13d-3 thereunder), in one transaction or a series of related transactions, of direct or indirect beneficial ownership of more than fifty percent (50%) of the outstanding voting equity securities of such Person (excluding, for clarity, an acquisition by a person or group where the equity holders of such acquired Person or its parent immediately prior to such transaction hold a majority of the outstanding voting equity securities of the surviving entity or the parent of the surviving entity

immediately following such transaction); (b) a merger, reorganization or consolidation involving such Person as a result of which (1) a person or group acquires direct or indirect beneficial ownership of more than fifty percent (50%) of the voting power of the surviving entity immediately after such merger, reorganization or consolidation and (2) the voting securities of such Person outstanding immediately prior to such merger, reorganization or consolidation, or any securities into which such voting securities have been converted or exchanged, cease to represent more than fifty percent (50%) of the combined voting power of the surviving entity or the parent of the surviving entity immediately following such merger, reorganization or consolidation; or (c) a sale, exclusive license or other transfer of all or a material part of the assets of such Person related to the transactions contemplated by this Agreement or the other Transaction Documents (e.g., in the case of Imugene, Parent or any of their respective Affiliates, the cell therapy business of Imugene, Parent or any of their respective Affiliates) in one transaction or a series of related transactions to a person or group. The acquiring or combining person or group in any of (a), (b) or (c), and any of such person's or group's Affiliates (whether in existence as of or any time following the applicable transaction, but other than such acquired Person and its Affiliates as in existence prior to the applicable transaction or Affiliates it controls after the applicable transaction) are referred to collectively herein as the "Acquirer".

- 1.29"Change of Control Consideration" means [***].
- 1.30"*Chimeric Antigen Receptor*" means a genetically engineered molecule that (a) when present on the surface of human T cells, enables the T cells to recognize and bind to specific antigens that are present on the surface of cells, and (b) comprises a single-chain antibody fragment (scFv), a transmembrane domain, and at least one intracellular signaling domain.
 - 1.31"*Claim*" has the meaning set forth in Section 12.1.1.
- 1.32"Clinical Trial" means a clinical study conducted on certain numbers of human subjects (depending on the phase of the trial) that is designed to (a) establish that a product for the treatment of human diseases and conditions is reasonably safe for continued testing, (b) investigate the safety and efficacy of the product for its intended use, and to define warnings, precautions and adverse reactions that may be associated with the product in the dosage range to be prescribed, or (c) support Marketing Authorization or Pricing and Reimbursement Approval of such product or label expansion of such product.
 - 1.33"CMO" means contract manufacturing organization.
 - 1.34"*Code*" has the meaning set forth in Section 14.7.
 - 1.35" *Combination Product*" has the meaning set forth in the definition of "Net Sales."
 - 1.36"Commercial Milestone Payment" has the meaning set forth in Section 9.2.
- 1.37"*Commercialization*" means any and all activities directed to the commercial exploitation a Licensed Product, including: (a) activities directed to storing, marketing, promoting, detailing, distributing, importing, exporting, selling and offering to sell that Licensed Product; (b) conducting Clinical Trials after Marketing Authorization of a Licensed Product with respect to

such Licensed Product; (c) interacting with Regulatory Authorities regarding the foregoing; and (d) seeking Regulatory Approvals (as applicable) for and registration of that Licensed Product; *provided* that seeking Marketing Authorization constitutes Development and not Commercialization. When used as a verb, "to *Commercialize*" and "*Commercializing*" means to engage in Commercialization and "*Commercialized*" has a corresponding meaning.

1.38"Commercially Reasonable Efforts" means:

- 1.38.1with respect to the obligations of a Party under this Agreement relating to Development activities, the level of efforts and expenditure of resources required to carry out such obligation in a sustained manner consistent with the efforts and resources such Party or its Affiliates typically devotes to a product of similar market potential, resulting from its own research efforts or development and commercialization collaborations for which it is responsible, at a similar stage in its development or product life, taking into account Relevant Factors; or
- 1.38.2 with respect to the level of obligations of a Party under this Agreement relating to Commercialization activities, the level of efforts and expenditure of resources required to carry out such obligation in a sustained manner consistent with the efforts and resources of a typical Third Party biopharmaceutical company of similar size and with similar resources as such Party or its Affiliates typically devotes to a product of similar market potential, at a similar stage in its development or product life, taking into account Relevant Factors;
- 1.38.3with respect to the obligations of a Party under this Agreement relating to any other objective, reasonable, good-faith efforts, taking into account industry practices.

Provided that, if in consideration of the Relevant Factors (or, as it relates to Section 1.38.3, industry practices), Commercially Reasonable Efforts requires any act to be performed, with respect to such performance and for the period of time during which Commercially Reasonable Efforts dictates such performance, Commercially Reasonable Efforts requires that the applicable Party (a) promptly assign responsibility for obligations to specific employee(s) who are held accountable for progress and monitor such act on an on-going basis, (b) set and consistently seek to achieve specific, meaningful and measurable objectives for carrying out such act, and (c) consistently make and implement decisions and allocate resources designed to advance progress with respect to such act.

- 1.39 "Confidential Proprietary Information" has the meaning set forth in Section 13.1.1.
- 1.40" *Confidentiality Agreement*" means that certain Mutual Confidentiality Agreement entered into between the Parties as of June 19, 2023.
- 1.41 "*Control*" or "*Controlled*" means, with respect to any Know-How, Patents, or other intellectual property rights, that a Party has the legal authority or right (whether by ownership, license, or otherwise, but without taking into account any rights granted by one Party to the other Party pursuant to this Agreement) to grant to the other Party a license, covenant not to sue, sublicense, access, or right to use (as applicable) under such Know-How, Patents, or other intellectual property rights, on the terms and conditions set forth herein, in each case without

violating any obligations of the granting Party owed to a Third Party or breaching the terms of any agreement with a Third Party.

- 1.42"*Cover*" means, with respect to a claim of a Patent and given product or other subject matter, that such claim would be infringed, in the absence of a license, or ownership, by the Exploitation of such product or other subject matter (considering claims of patent applications to be issued as then pending).
 - 1.43" **Depositary**" has the meaning set forth in the Registration Rights Agreement.
- 1.44" **Development**" means all activities related to the development of products, including Licensed Products, for the treatment of human diseases and conditions. When used as a verb, "**Develop**" or "**Developing**" means to engage in Development and "**Developed**" has a corresponding meaning.
- 1.45" **Directed to**" means, when used to describe the relationship between an engineered human T cell and a Target, that the T cell (a) is designed or being developed to bind to the Target (or a portion thereof), (b) is designed or being developed to exert its biological effect in whole or in part through binding to such Target (or such portion thereof), and (c) is not designed or developed to bind to or exert its effect on any other Target (or a portion thereof).
 - 1.46" *Disclosing Party*" has the meaning set forth in Section 13.1.2.
 - 1.47" **Dispute**" has the meaning set forth in Section 15.2.
- 1.48" *Distributor*" means, as applicable, with respect to a given Licensed Product, any Person appointed by (a) Imugene, (b) any of Imugene's Affiliates or (c) any of their respective Sublicensees that is not an Affiliate of (a) or (b), to distribute, market and sell the Licensed Product in one or more countries in the Territory, in circumstances where the Person (x) purchases its requirements of the Licensed Product from Imugene for their respective Affiliates or its or their Sublicensees but (y) has no right to conduct any Development or Manufacturing (other than packaging) activities with respect to such Licensed Product.
 - 1.49" *Dollar*" means a U.S. dollar, and "\$" is to be interpreted accordingly.
 - 1.50" **Duke**" has the meaning set forth in the definition of "Duke Agreement".
- 1.51"*Duke Agreement*" means the License Agreement entered into by Precision and Duke University ("*Duke*") on April 17, 2006, as amended by the Amendment, dated May 31, 2007 and as further amended by the Letter Agreements, dated December 10, 2007, February 13, 2009, January 17, 2012, December 6, 2013, December 13, 2013 and February 4, 2014, and as further amended from time to time.
- 1.52"**Duke IP**" means all Patents and Know-How licensed to Precision under the Duke Agreement that constitute Precision Background IP. The patent numbers and patent application numbers of the Patents that are included within the Duke IP as of the Effective Date are set forth in <u>Schedule 1.52</u>.

- 1.53"*Effective Date*" has the meaning set forth in the preamble to this Agreement.
- 1.54" *EMA*" means the European Medicines Agency (or the competent UK authority) or any successor agency thereto.
- 1.55"*E.U.*" means the European Union as constituted on the Effective Date.
- 1.56" *Excluded Target*" means each of the following Targets: [***].
- 1.57" *Executive Officers*" means (a) with respect to Precision, [***], and (b) with respect to Imugene, [***]; or any other person that such person in the foregoing (a) or (b) designates from time to time.
 - 1.58" Existing In-License Agreements" means the Duke Agreement and the Cellectis Agreement.
 - 1.59" Existing Patents" has the meaning set forth in Section 11.2.2.
- 1.60" *Existing Product*" means the CAR-T product Directed to CD19 known as "azer-cel" and having the Precision internal designation PBCAR0191, including any preparation, formulation, dosage, packaging or method of administration thereof.
- 1.61" *Expansion Cohort*" means that certain Phase I expansion cohort to the Ongoing Precision Trial that is ongoing as of the Effective Date.
 - 1.62" Expansion Cohort Modification" means [***].
 - 1.63" Expansion Cohort Success Milestone" has the meaning set forth in Section 9.1.
 - 1.64" *Expansion Cohort Success Milestone Cash*" has the meaning set forth in Section 9.1.2.
 - 1.65" Expansion Cohort Success Milestone Consideration" has the meaning set forth in Section 9.1.2.
 - 1.66" Expansion Cohort Success Milestone Payment Date" has the meaning set forth in Section 9.1.2.
 - 1.67" Expansion Cohort Success Milestone Shares" has the meaning set forth in Section 9.1.2.
- 1.68" *Exploit*" means to Research, Develop, Manufacture, Commercialize and otherwise exploit. "*Exploitation*" has correlating meaning.
 - 1.69"FDA" means the United States Food and Drug Administration or any successor agency thereto.

1.70" First Commercial Sale" means, with respect to a given Licensed Product, the first sale of such Licensed Product by the applicable Selling Party to a Third Party for end use or consumption of such Licensed Product in a given country in the Territory after Marketing Authorization required to market and sell the Licensed Product has been granted with respect to such Licensed Product by the applicable Regulatory Authority in such country in which such Licensed Product is sold.

1.71[***]

- 1.72"Good Clinical Practices" or "cGCP" means all applicable current Good Clinical Practice standards for the design, conduct, performance, monitoring, auditing, recording, analysis and reporting of Clinical Trials, including, as applicable: (a) as set forth in the International Conference on Harmonisation of Technical Requirements for Registration of Pharmaceuticals for Human Use ("ICH") E6 and any other guidelines for good clinical practice for trials on medicinal products in the Territory; (b) the Declaration of Helsinki (2004) as last amended at the 52nd World Medical Association in October 2000 and any further amendments or clarifications thereto; (c) U.S. Code of Federal Regulations Title 21, Parts 50, 54, 56, 312 and 314, as may be amended from time to time; and (d) the equivalent Applicable Laws in any relevant country, each as may be amended and applicable from time to time, and in each case ((a)-(d)), that provide for, among other things, assurance that the clinical data and reported results are credible and accurate and protect the rights, integrity, and confidentiality of trial subjects.
- 1.73"*Good Laboratory Practices*" or "*GLPs*" means all applicable Good Laboratory Practice standards, including, as applicable: (a) as set forth in the then-current good laboratory practice standards promulgated or endorsed by the FDA as defined in 21 C.F.R. Part 58; and (b) the equivalent Applicable Laws in any relevant country, each as may be amended and applicable from time to time.
- 1.74"*Good Manufacturing Practices*" or "*cGMP*" means all applicable current Good Manufacturing Practices including, as applicable: (a) the principles detailed in the US Current Good Manufacturing Practices, 21 C.F.R. Parts 4, 210, 211, 601, 610 and 820; (b) European Directive 2003/94/EC and Eudralex 4; (c) the principles detailed in the WHO TRS 986 Annex 2, TRS 961 Annex 6, TRS 957 Annex 2, and TRS 999 Annex 2; (d) ICH Q7 guidelines; and (e) the equivalent Applicable Laws in any relevant country, each as may be amended and applicable from time to time.
 - 1.75" *Government Official*" has the meaning set forth in Section 11.6.4.
- 1.76"*Governmental Authority*" means any national, international, federal, state, provincial or local government, or political subdivision thereof, or any multinational organization or any authority, agency or commission entitled to exercise any administrative, executive, judicial, legislative, police, regulatory or taxing authority or power, and any court or tribunal (or any department, bureau or division thereof, or any governmental arbitrator or arbitral body).
 - 1.77"*ICH*" has the meaning set forth in the definition of "Good Clinical Practices".
 - 1.78" *Imagene*" has the meaning set forth in the Preamble.

- 1.79" *Imagene Arising IP*" means, individually or collectively, Imagene Sole IP and Imagene's share in Joint IP.
- 1.80"*Imugene Background IP*" means any and all Patents and Know-How that Imugene or any of its Affiliates Controls as of the Effective Date, or discovers, creates or acquires outside the scope of its performance of activities under this Agreement; in each case, that is necessary or reasonably useful for the Exploitation of any Licensed Product.
 - 1.81" *Imagene Indemnitee*" has the meaning set forth in Section 12.1.1.
 - 1.82" Imagene Patent" means any Patent constituting or claiming any Imagene Background IP or Imagene Sole IP.
 - 1.83"*Imagene Sole IP*" has the meaning set forth in Section 10.1.2.
 - 1.84" Imagene Technology" means Imagene Background IP and Imagene Sole IP.
- 1.85" *Included Target*" means any of the up to three (3) Targets that has become an Included Target pursuant to Section 3.1.
- 1.86"*IND*" means an investigational new drug application filed with the FDA or any similar application filed with a Regulatory Authority in a country outside the U.S. required to commence Clinical Trials of a pharmaceutical product.
 - 1.87"*Indemnitee*" has the meaning set forth in Section 12.1.3.
 - 1.88"*Indemnitor*" has the meaning set forth in Section 12.1.3.
 - 1.89"*Infringement*" has the meaning set forth in Section 10.3.1.
 - 1.90" *Initiation*" means with respect to any Clinical Trial, the enrollment of the first human subject in such Clinical Trial.
 - 1.91" *Initiation Milestone*" has the meaning set forth in Section 9.1.
 - 1.92" *Insolvency Event*" means any of the events set out in Section 14.2.3.
- <u>1.93</u>"*Inventions*" means all Know-How and inventions, whether or not patentable, and all rights, title and interest in and to the intellectual property rights (including Patent rights) therein.
 - 1.94" Issue Date" means the date on which the Expansion Cohort Success Milestone Shares are issued.
 - 1.95" *Joint IP*" has the meaning set forth in Section 10.1.2.
 - 1.96" Joint Patents" means any Patent constituting or claiming any Joint IP.

- 1.97 "*JSC*" has the meaning set forth in Section 2.2.1.
- 1.98"*Know-How*" means any proprietary scientific or technical information, inventions, discoveries, results and data of any type whatsoever, in any tangible or intangible form, including databases, safety information, practices, methods, instructions, techniques, processes, drawings, documentation, specifications, formulations, formulae, knowledge, know-how, trade secrets, materials, skill, experience, test data and other information and technology applicable to formulations, compositions or products or to their Exploitation or to methods of assaying or testing them, including pharmacological, pharmaceutical, medicinal chemistry, biological, chemical, biochemical, toxicological and clinical test data, physical and analytical, safety, quality control data, manufacturing, and stability data, studies and procedures, and manufacturing process and development information, results and data.
- 1.99"*Knowledge*" means the actual knowledge of each of Precision's Chief Executive Officer, Chief Research Officer, Chief Financial Officer, Chief Business Officer and Vice President of Intellectual Property after due inquiry.
 - 1.100"Licensed ARCUS Nuclease" means [***].
- 1.101"*Licensed Field*" means the treatment, prevention, cure, mitigation or palliation of any and all cancers (i.e., diseases, conditions or disorders identified in chapter 2 ("Neoplasms") of the 11th revision of the International Classification of Diseases of the World Health Organization (ICD-11) or a successor thereto). [***]
 - 1.102" Licensed Product" means an Existing Product or an Additional Product, as applicable.
 - 1.103" *Licensed Product Trademarks*" has the meaning set forth in Section 10.8.
 - 1.104"Losses" has the meaning set forth in Section 12.1.1.
 - 1.105"*Major Foreign Markets*" means [***].
- 1.106"*Manufacture*" and "*Manufacturing*" means any and all activities related to the production, manufacture, formulation, finishing, packaging, labeling, shipping and holding of a relevant Licensed Product, or other product or therapy, or any component, intermediary or precursor thereof (including, for clarity, [***]), and including process development, process qualification and validation, scale-up, pre-clinical, clinical and commercial manufacture, characterization, quality assurance and quality control (including testing).
 - 1.107"*Manufacturing Technology Transfer*" has the meaning set forth in Section 7.2.
- 1.108"*Marketing Authorization*" means, with respect to a particular Licensed Product in a particular country or regulatory jurisdiction, collectively, all Regulatory Approvals (including any Pricing and Reimbursement Approval or access approvals, if applicable) required by the relevant Regulatory Authority in order to initiate marketing, selling or Commercializing a Licensed Product in such country or jurisdiction.

1.109"*Medical Affairs*" means activities conducted by a Party's medical affairs departments (or, if a Party does not have a medical affairs department, the equivalent function thereof), including communications with key opinion leaders, medical education, symposia, advisory boards (to the extent related to medical affairs or clinical guidance), activities performed in connection with patient registries, and other medical programs and communications, including educational grants, research grants (including conducting investigator-initiated studies), and charitable donations to the extent related to medical affairs related to any Existing Product and not to other activities that do not involve the promotion, marketing, sale, or other Commercialization of any Existing Product and are not conducted by a Party's medical affairs (or equivalent) departments. Medical Affairs excludes any activities directed to Manufacturing, Development, or Commercialization.

1.110"*Medical Affairs Plan*" means, with respect to an Existing Product, the written high-level strategic and tactical plans for the Medical Affairs activities for such Existing Product to be conducted in the Licensed Field in the Territory that will be prepared and updated by Imugene as provided in Section 5.6.

- 1.111" *Milestone Event*" means any milestone event set forth in Section 9.1 or Section 9.2.
- 1.112"Nasdaq" means the Nasdaq Stock Market LLC.
- 1.113"*Net Sales*" means[***]

The foregoing amounts shall be determined from the books and records of the Selling Party, maintained in accordance with U.S. GAAP, consistently applied. [***] In the case of sale or disposal of the applicable Licensed Product for consideration other than exclusively monetary consideration, Net Sales for such Licensed Product shall be the value of the non-cash consideration received, as determined in accordance with U.S. GAAP. In no event will any particular amount identified above be deducted more than once in calculating Net Sales. Sales of the applicable Licensed Product between the individual Selling Parties for such Licensed Product for resale shall be excluded from the computation of Net Sales (unless such Licensed Product is consumed by such Selling Party), but the subsequent resale of such Licensed Product by such Selling Party to a Third Party shall be included within the computation of Net Sales. Licensed Products transferred as part of an expanded access program, compassionate sales or use program, an indigent program, as *bona fide* samples, as donations, or for the performance of clinical trials, shall not be included in Net Sales for such Licensed Product.

For purposes of determining Net Sales of any Licensed Product sold in combination with or as part of a bundle with other products, or in packaged arrangements to customers that include a Licensed Product, in each case other than Combination Products (which are addressed below), [***]

In the event that the Licensed Product is sold as part of a Combination Product (where "Combination Product" means any pharmaceutical product which comprises the Licensed Product and one or more other Active Ingredients that do not constitute the Licensed Product,

whether co-formulated, co-packaged or otherwise sold together for one price), the Net Sales of the Licensed Product, for the purposes of determining royalty payments, shall be determined by [***].

- 1.114"*Ongoing Precision Trial*" means the Clinical Trial sponsored by Precision with respect to the Existing Product that is set forth in <u>Schedule 1.114</u>.
 - 1.115" *Ordinary Shares*" means the fully paid ordinary shares of Parent.
 - 1.116"*Parent*" has the meaning set forth in the Preamble.
 - 1.117" *Party*" and "*Parties*" has the meaning set forth in the Preamble.
- 1.118"**Patent Defense Matters**" means the conduct of interferences, derivation proceedings, *inter partes* review and post-grant review, the defense of oppositions and other similar proceedings with respect to a Patent, excluding any activities associated with claims, including as a counterclaim or declaratory judgment action, of unpatentability, invalidity or unenforceability of such Patent that are brought by a Third Party in connection with an alleged or threatened infringement by a Third Party of a Patent.
- 1.119"**Patents**" mean: (a) pending patent applications, issued patents, utility models and designs; (b) reissues, substitutions, confirmations, registrations, validations, re-examinations, additions, continuations, continued prosecution applications, continuations-in-part, or divisions of or to any of the foregoing; and (c) extensions, renewals or restorations of any of the foregoing by existing or future extension, renewal or restoration mechanisms, including supplementary protection certificates or the equivalent thereof.
- 1.120"*Person*" means any corporation, limited or general partnership, limited liability company, joint venture, trust, unincorporated association, governmental body, authority, bureau or agency, any other entity or body, or an individual.
- 1.121" *Phase I Clinical Trial*" means a Clinical Trial that would satisfy the requirements of 21 C.F.R. § 312.21(a) (or equivalent regulation in countries other than the United States).
- 1.122" *Phase II Clinical Trial*" means a Clinical Trial that would satisfy the requirements of 21 C.F.R. § 312.21(b) (or equivalent regulation in countries other than the United States). [***]
- 1.123"*Phase III Clinical Trial*" means a controlled or uncontrolled human Clinical Trial of a product that would satisfy the requirements of 21 C.F.R. § 312.21(c) (or equivalent regulation in countries other than the United States). [***]
 - 1.124"Pivotal Clinical Trial" means [***].
 - 1.125" **Precision**" has the meaning set forth in the Preamble.
 - 1.126" Precision Arising IP" means, individually or collectively, Precision Sole IP and Precision's share in Joint IP.

- 1.127" **Precision Arising Platform IP**" means any and all Precision Arising IP that is not Precision Arising Product IP.
- 1.128" *Precision Arising Product IP*" means Precision Arising IP that that is necessary or reasonably useful for the Exploitation of an Existing Product.
- 1.129"*Precision Background IP*" means any and all (a) Patents Controlled by Precision or its Affiliates at any time during the Term that Cover the Licensed ARCUS Nuclease or a Licensed Product or any Know-How in the following clause (b); and (b) Know-How Controlled by Precision or its Affiliates (i) as of the Effective Date or (ii) during the Term solely to the extent Precision has disclosed such Know-How to Imugene in writing or during the conduct of a JSC meeting hereunder, in each case (i) and (ii) that is necessary or reasonably useful for the Exploitation of a Licensed Product, including Know-How that is specific to the Licensed ARCUS Nuclease or any ARCUS Research Tool.
- 1.130" *Precision Background Platform IP*" means any and all Precision Background IP that is not Precision Background Product IP, including ARCUS Technology other than ARCUS Research Tools.
- 1.131"**Precision Background Product IP**" means any and all Precision Background IP that (a) is necessary or reasonably useful for the Exploitation of an Existing Product, or (b) Covers (in the case of a Patent) or is specific to (in the case of Know-How) the Licensed ARCUS Nuclease or any ARCUS Research Tool.
 - 1.132"*Precision Indemnitee*" has the meaning set forth in Section 12.1.2.
 - 1.133" **Precision Patent**" means any Patent included in the Precision Technology.
- 1.134" *Precision Platform IP*" means, individually or collectively, the Precision Background Platform IP and the Precision Arising Platform IP.
- 1.135" *Precision Product IP*" means, individually or collectively, the Precision Background Product IP and the Precision Arising Product IP.
 - 1.136" Precision Product Patent" means [***].
 - 1.137" Precision Product-Specific Claim" means [***].
 - 1.138" *Precision Sole IP*" has the meaning set forth in Section 10.1.2(b).
- 1.139"**Precision Technology**" means, individually or collectively, the Precision Background IP and the Precision Arising IP.
- 1.140"*Pricing and Reimbursement Approval*" means, with respect to a particular Licensed Product and a particular country or regulatory jurisdiction, any approval, agreement, determination or decision of any Regulatory Authority establishing the price or level of reimbursement for such Licensed Product, as required in a given country or jurisdiction prior to sale of such Licensed Product in such country or jurisdiction at the relevant time.

- 1.141"*Proceeds*" means all compensation of any kind received by Imugene or any of its Affiliates in consideration for or otherwise in connection with the execution of a Product Partnering Transaction (including [***]).
- 1.142" *Product Partnering Transaction*" means a transaction pursuant to which Imagene or any of its Affiliates grants to a Third Party [***].
- 1.143"*Prosecute and Maintain*" or "*Prosecution and Maintenance*" with respect to a particular Patent, means (a) all activities associated with the preparation, filing, prosecution and maintenance of such Patent, and (b) all Patent Defense Matters with respect to such Patent.
 - 1.144"*Prosecuting Party*" has the meaning set forth in Section 10.2.2.
 - 1.145"*Receiving Party*" has the meaning set forth in Section 13.1.2.
 - 1.146"*Reference Rate*" is the [***].
 - 1.147" *Registration Rights Agreement*" has the meaning set forth in the Asset Purchase Agreement.
- 1.148"*Regulatory Approvals*" means, collectively, any and all approvals (including supplements, amendments, pre- and post-approvals, Pricing and Reimbursement Approvals), licenses, registrations, permits, notifications, and authorizations (including marketing and labeling authorizations) or waivers of any Regulatory Authority that are necessary for the testing or Exploitation of a pharmaceutical product (including any Licensed Product) in any country or jurisdiction, including Pricing and Reimbursement Approval, as applicable.
- 1.149"*Regulatory Authority*" means any Governmental Authority that has responsibility in its applicable jurisdiction over the Exploitation of pharmaceutical products (including any Licensed Product) in any country or jurisdiction. For countries or jurisdictions where governmental approval is required for pricing or reimbursement for a pharmaceutical product (including any Licensed Product) to be reimbursed by national health insurance (or its local equivalent), Regulatory Authority includes any Governmental Authority whose review or approval of pricing or reimbursement of such product is required.
- 1.150"*Regulatory Filings*" means, collectively, any and all applications, filings, submissions, approvals (including supplements, amendments, pre- and post-approvals, pricing and reimbursement approvals), licenses, registrations, notifications, and authorizations (including marketing and labeling authorizations), non-clinical and clinical study authorization applications or notifications (including all supporting files, writings, data, studies and reports) or waivers with respect to the Commercialization of a pharmaceutical product (including Licensed Products) made to or received from any Regulatory Authority or research ethics committee in a given country or jurisdiction, including INDs and BLAs.
- 1.151"*Relevant Factors*" means all factors that are relevant to the Development, manufacture or Commercialization of a product, including its safety and efficacy, product profile, cost to develop, cost and availability of supply, the time required to complete Development, the

competitiveness of the marketplace (including the proprietary position and anticipated market share of the product), the patent position with respect to such product (including the ability to obtain or enforce, or have obtained or enforced, such patent rights), the third-party patent landscape relevant to the product, the regulatory structure involved, the likelihood of regulatory approval, the anticipated or actual profitability of the applicable product and other technical, commercial, legal, scientific, regulatory and medical considerations.

- 1.152"*Representatives*" has the meaning set forth in Section 10.1.6.
- 1.153"*Research*" means, with respect to a Licensed Product, or other product or therapy, any and all activities directed to the discovery, identification, screening, testing, assessment and optimization of such Licensed Product, or other product or therapy.
- 1.154"*Right of Reference*" means the authority to rely upon, and otherwise use, an investigation for the purpose of filing, and conducting a Clinical Trial under, an IND, or obtaining approval of a Marketing Authorization or other Regulatory Approval, including the ability to make available the underlying raw data from the investigation for audit by the applicable Regulatory Authority in such country or other jurisdiction, if necessary.
 - 1.155" Royalty" has the meaning set forth in Section 9.3.2.
 - 1.156" *Royalty Term*" has the meaning set forth in Section 9.3.1.
 - 1.157[***]
 - 1.158"*Selling Party*" means Imugene, its Affiliates or its or their Sublicensees.
- 1.159"*Sublicensee*" means a Third Party that is granted a license or sublicense to Develop, Manufacture or Commercialize a Licensed Product in the Licensed Field in the Territory, beyond the mere right to purchase such Licensed Product from Imugene and its Affiliates, and excludes Imugene's and its Affiliates' Distributors.
- 1.160"*Target*" means a single unique molecular species that (a) is chemically distinct from other molecules, (b) is an antigenic protein or antigenic glycolipid protein complex that is expressed on or in a human cell (including all epitopes of such antigenic protein or antigenic glycolipid protein complex), and (c) wherein a binding entity derives recognized therapeutic value from binding such molecular species. Notwithstanding the foregoing, "Target" does not include any human gene that would otherwise be included in the foregoing definition, genetic variations or mutations in which cause or contribute to a human disease, wherein a therapeutic effect with respect to such disease may be achieved by delivery of an *in vivo* gene editing product intended to permanently add to, subtract or modify such gene in a patient's cells.
 - 1.161"*Term*" has the meaning set forth in Section 14.1.
- 1.162"*Terminated Existing Product*" means an Existing Product that has become a Terminated Product in accordance with Article 14.
 - 1.163"*Terminated Product*" has the meaning set forth in Section 14.3.

- 1.164"*Territory*" means worldwide.
- 1.165" *Third Party*" means any Person other than Imugene or Precision (or their respective Affiliates).
- 1.166"*Trading Day*" means a day on which the ASX is open for trading, provided that if no closing price or daily trading volume is reported in respect of the relevant shares on the ASX for one or more consecutive trading days, such day or days will be disregarded in any relevant calculation and shall be deemed not to have existed when ascertaining any period of trading days.
 - 1.167"*Transaction Documents*" has the meaning set forth in the Asset Purchase Agreement.
 - 1.168"*Unavailable Target*" means: (a) [***] or (b) an Excluded Target.
 - 1.169"*U.S.*" means the United States of America and its territories and possessions.
 - 1.170" *U.S. GAAP*" has the meaning set forth in the definition of "Net Sales".
- 1.171"Valid Claim" means, with respect to a given Licensed Product, a claim that Covers (i) [***], (ii) [***] or (iii) [***], in each case (i) (iii) contained in (a) an issued and unexpired Patent that has not been revoked or held unenforceable, unpatentable or invalid by a decision of a court or other governmental agency of competent jurisdiction that is not appealable or has not been appealed within the time allowed for appeal; or (b) a pending patent application that has not been cancelled, withdrawn or abandoned or finally rejected by an administrative agency action from which no appeal can be taken and that has been pending for no longer than [***].
- 1.172"*VWAP*" means the arithmetic average of the daily volume weighted average sale price of the relevant shares sold on the ASX during the Trading Day (subject to excluding certain special trades, crossings, overseas trades and trades pursuant to exercise of options, and appropriate adjustments being made in respect of reconstructions, consolidations, divisions or reclassifications of such shares into a lesser or greater number of securities, other than a buyback or capital reduction, during the relevant period, and in respect of certain other market circumstances to adjust for market anomalies, such as suspensions of trading).
 - 1.173" Withholding Tax Action" has the meaning set forth in Section 9.9.4.

ARTICLE 2

GOVERNANCE AND JOINT STEERING COMMITTEE

2.1**Relationship Managers**. No later than [***] after the Effective Date, each Party will appoint an individual to act as its relationship manager under this Agreement as soon as practicable after the Effective Date (each a "*Relationship Manager*"). The Relationship Managers will: (a) serve as the primary points of contact between the Parties for the purpose of providing the other Party with information on the progress of a Party's activities under this Agreement; (b) be responsible for facilitating the flow of information and otherwise promoting communication, coordination, and collaboration between the Parties; (c) facilitate the prompt resolution of any

disputes; and (d) attend JSC meetings, in each case, as a non-voting member. A Relationship Manager may also bring any matter to the attention of the JSC if such Relationship Manager reasonably believes that such matter warrants such attention. Each Party will use reasonable efforts to keep an appropriate level of continuity but may replace its Relationship Manager at any time upon written notice to the other Party.

2.2Joint Steering Committee.

- 2.2.1 **Establishment; Purpose of JSC**. No later than [***] after the Effective Date, the Parties will establish a joint steering committee (the "*JSC*") to monitor the Exploitation of the Licensed Products in the Licensed Field in the Territory. The JSC will be composed of an equal number of representatives from each Party and a minimum of three (3) representatives of each Party and who have the appropriate and direct knowledge and expertise and requisite decision-making authority. Each Party may replace any of its representatives on the JSC and appoint a person to fill the vacancy arising from each such replacement. A Party that replaces a representative will notify the other Party at least [***] prior to the next scheduled meeting of the JSC. Both Parties will use reasonable efforts to keep an appropriate level of continuity in representation. Representatives may be represented at any meeting by another person designated by the absent representative. Each Party's representatives on the JSC will inform and coordinate within their respective organization to enable each Party to fulfill its obligations as agreed upon between the Parties under this Agreement, including within the time frames set forth hereunder.
- 2.2.2 **Meeting Agendas**. Each Party will disclose to the other Party the proposed agenda items along with appropriate information at least [***] in advance of each meeting of the JSC; *provided* that under exigent circumstances requiring JSC input, a Party may provide its agenda items to the other Party within a shorter period of time in advance of a meeting, or may propose that there not be a specific agenda for a particular meeting, so long as such other Party consents to such later addition of such agenda items or the absence of a specific agenda for such JSC meeting.
- 2.2.3 **Meetings**. The JSC will hold meetings at such times as it elects to do so, but will meet no less frequently than quarterly, unless otherwise agreed by the Parties. The JSC may meet in person or by means of teleconference, Internet conference, videoconference, or other similar communication method; *provided* that the Parties will use reasonable efforts for at least one meeting each Calendar Year to be conducted in person at a location selected alternatively by Precision and Imugene or such other location as the Parties may agree. Each Party will be responsible for all of its own costs and expenses of participating in any JSC meeting. The Relationship Managers will jointly prepare and circulate minutes for each JSC meeting within [***] after each such meeting and will ensure that such minutes are reviewed and approved by their respective companies within [***] thereafter.

2.2.4 **JSC Responsibilities**. The responsibilities of the JSC will be to:

(a) provide a forum for the discussion of the Parties' activities and the flow of information contemplated under this Agreement;

(b) review and discuss the Development of each Existing Product, including clinical trial protocols, monitoring plans, and data disclosure plans included with each such protocol;

- (c) [***] approve any clinical trial protocols, monitoring plans, and data disclosure plans included with each such protocol with respect to such Existing Product;
 - (d) monitor progress of the Expansion Cohort;
- (e) consider and discuss the results of the Expansion Cohort upon its completion (or earlier termination, if required by the FDA);
- (f) monitor progress of the Development of the Licensed Products in the Licensed Field in the Territory;
- (g) review and discuss matters that may have a material adverse impact upon the regulatory status of any Existing Product, as described in Section 5.1.2(f);
- (h) review and discuss Medical Affairs Plans and any updates thereto for any Existing Product, as described in Section 5.6;
 - (i) review and discuss the Commercialization of the Existing Product;
- (j) oversee the implementation of activities to be performed under, any Technology Transfer Plan and any other written agreement between the Parties with respect to the subject matter hereof; and
- (k) perform such other functions as expressly set forth in this Agreement or allocated to the JSC by the Parties' written agreement.
- 2.3Non-Member Attendance. Each Party may from time to time invite a reasonable number of participants, in addition to its representatives (which may include legal counsel), to attend a meeting of the JSC (in a non-voting capacity), if such participants have expertise that is relevant to the planned agenda for such JSC meeting; *provided* that if either Party intends to have any Third Party (including any consultant) attend such a meeting, then such Party will provide prior written notice to the other Party reasonably in advance of such meeting and will ensure that such Third Party is bound by obligations of confidentiality and non-use at least as stringent as those set forth in Article 13.

2.4Decision-Making.

2.4.1 **General Process**. The JSC will only have the advisory powers expressly assigned to it in this Article 2 and elsewhere in this Agreement and will not have the authority to: (a) modify or amend the terms of this Agreement; or (b) waive either Party's compliance with the terms of this Agreement. [***] No action taken at any meeting of the JSC will be effective unless there is a quorum at such meeting, and at all such meetings, a quorum will be reached if two voting representatives of each Party are present or participating in such meeting. Except as otherwise expressly set forth in this Agreement, the phrases "determine," "designate," "confirm," "approve," or "determine whether to approve" by the JSC and similar phrases used in this Agreement will mean approval in accordance with this Section 2.4, including the escalation and tie-breaking provisions herein. For the avoidance of doubt, matters that are specified in Section 2.2.4 to be reviewed and discussed (as opposed to reviewed, discussed, and approved) do not require any agreement or decision by either Party and are not subject to the voting and decision-making procedures set forth in this Section 2.4.

2.4.2 **Decisions of the JSC**. The JSC will use good faith efforts, in compliance with this Section 2.4.2, to promptly resolve any such matter for which it has authority. If, after the use of good faith efforts, the JSC is unable to resolve any such matter that is within the scope of the JSC's authority or any other disagreement between the Parties that may be referred to the JSC, in each case, within a period of [***], then a Party may refer such matter for resolution in accordance with Section 2.5.1.

2.5Resolution of JSC Disputes.

- 2.5.1 **Referral to Executive Officers**. If a Party makes an election under Section 2.4.2 to refer a matter on which the JSC cannot reach a consensus decision for resolution by the Executive Officers, then the JSC will submit in writing the respective positions of the Parties to their respective Executive Officers. The Executive Officers will use good faith efforts to resolve any such matter so referred to them as soon as practicable, and any final decision that the Executive Officers agree to in writing will be conclusive and binding on the Parties.
- 2.5.2 **No Change; Status Quo**. If the Executive Officers are unable to reach agreement on any such matter referred to them within [***] after such matter is so referred (or such longer period as the Executive Officers may agree upon), then neither Party will have final decision-making authority over approval of such matter and all such matters must be decided by unanimous agreement of the Parties in order to take any action or adopt any change from the then-current *status quo*, as applicable, provided that Precision will have final decision-making authority with respect to [***].
- 2.5.3 **Limitations on Decision-Making**. Notwithstanding any provision to the contrary set forth in this Agreement, without the other Party's prior written consent, neither Party (in the exercise of a Party's final decision-making authority), the JSC, nor a Party's Executive Officer, in each case, may make a decision that could reasonably be expected to [***].
- 2.6**Discontinuation of JSC**. The JSC will continue to exist until the first to occur of (a) [***], or (b) [***]. Once the JSC is disbanded, the JSC will have no further obligations under this Agreement and, thereafter, the Relationship Managers will be the points of contact for the exchange of information between the Parties under this Agreement.

ARTICLE 3

INCLUDED TARGETS

3.1Nomination of Included Targets. During the period beginning on the Effective Date and ending on the earlier of (a) the fifth (5th) anniversary of the Effective Date and (b) the date that the third (3rd) Target is deemed an Included Target hereunder (the "Target Nomination Period"), Imugene shall have the right, subject to the terms and conditions of this Agreement, to name up to three (3) oncology Targets to be included as an Included Target under this Agreement in accordance with this Section 3.1. Imugene may exercise such right in its sole discretion at any time during the Target Nomination Period by providing written notice to Precision, through the JSC, specifying the identity of the oncology Target that Imugene desires to include as an Included Target under this Agreement (each, a "Nomination Request"), provided that if any such Target is an Unavailable Target at the time Precision receives such Nomination Request, then Precision shall within [***] of receipt of the Nomination Request provide written notice to Imugene that

such Target is an Unavailable Target, and such Nomination Request shall have no further effect. If the Target specified in such Nomination Request is not an Unavailable Target, then such Target shall be deemed an Included Target upon receipt of such Nomination Request by Precision for purposes of this Agreement. [***]

3.2**Reservation of Rights**. Precision will be free to grant rights for any Target that is not an Included Target (including any Unavailable Target) to any Third Party at any time.

ARTICLE 4

DEVELOPMENT MATTERS

4.1Existing Product.

- 4.1.1 **Conduct of the Parties**. The Parties' mutual objective is to permit Imugene, pursuant to and in accordance with the terms of this Agreement, to Develop the Existing Product(s) in the Licensed Field while not taking any action that would be reasonably likely to materially adversely affect Development of any Existing Product outside the Licensed Field. Each Party shall conduct itself and its activities hereunder consistent with that understanding, consistent with sound and ethical business and scientific practices. In all matters related to such activities, the Parties shall strive to balance, as best as reasonably possible, their respective legitimate interests and concerns and to realize the economic potential of the Existing Product(s) in the Licensed Field and outside the Licensed Field.
- 4.1.2 **Ongoing Precision Trial**. Promptly following the Effective Date, to the extent permitted by Applicable Laws, Precision will transfer to Imugene the Ongoing Precision Trial and furnish Imugene with reasonable cooperation to transition to Imugene the management and continued performance of such Ongoing Clinical Trial in a manner in compliance with Applicable Laws and ethical guidelines.

4.1.3 Expansion Cohort.

(a) As of the Effective Date, the Parties acknowledge that Precision has submitted the Expansion Cohort Modification to the FDA. If the FDA objects to the Expansion Cohort Modification during the applicable review period, then the Parties will make adjustments to the Expansion Cohort Modification based on the feedback received from the FDA and submit such adjustments to the FDA for review, as soon as possible after receipt of such FDA feedback, all in accordance with the terms of this Agreement.

(b) [***]

4.1.4 **Development Responsibility; Diligence Obligations**. Subject to the terms of this Agreement, Imugene shall be responsible for, at its sole cost and expense, all Development of the Existing Product in Licensed Field in the Territory, including all Clinical Trials and activities that are necessary for or otherwise support obtaining and maintaining Regulatory Approvals in the Licensed Field in the Territory. Imugene shall use Commercially Reasonable Efforts to Develop and seek and obtain Regulatory Approval for each Existing Product in the Licensed Field throughout the Territory in accordance with all Applicable Laws. Without limiting the foregoing,

- (a) Imugene (i) will initiate the first dosing of a human subject in the Expansion Cohort with the Existing Product that is the subject of the Expansion Cohort Modification (or any adjustment thereof submitted in accordance with Section 4.1.3(a)) by the later of (A) [***], or (B) [***] after the FDA does not object to the Expansion Cohort Modification (or any adjustment thereof submitted in accordance with Section 4.1.3(a)) during the applicable review period therefor, and (ii) will complete the Expansion Cohort in accordance with the study design set forth in the applicable protocol, unless otherwise required by the FDA; and (b) [***].
- 4.1.5 **Standard of Conduct**. Imugene will perform, and will cause its Affiliates, Sublicensees, and subcontractors to perform, all Development activities for any Existing Product in a timely, good scientific manner, in accordance with GLP, cGMP, and cGCP, as applicable, and in compliance with Applicable Laws. In addition, Imugene will conduct its obligations with respect to any Clinical Trial with the study design set forth in the applicable protocol, each as may be amended from time to time.
- 4.1.6 **Transfer of IND**. Precision will use its Commercially Reasonable Efforts to assign and transfer within [***] all of its rights, title and interests in and to any INDs and orphan drug designations for the Licensed Products. Imugene will cooperate in connection therewith and shall perform all duties under such INDs and orphan drug designations from and after assignment. Subject to the foregoing, the Parties will reasonably cooperate to ensure an orderly transition of duties under such INDs and orphan drug designations and to fulfill applicable filing obligations with Regulatory Authorities. Imugene will not transfer or otherwise assign to an Affiliate or Third Party any such Regulatory Filing that was in existence as of the Effective Date and was transferred to Imugene pursuant to this Section 4.1.6 without obtaining a Right of Reference to such Regulatory Filing for Precision, its Affiliates and any of its licensees consistent with Section 5.3; any purported assignment or transfer that is not in compliance with the foregoing shall be null and void.

4.2 Development Records.

- 4.2.1 **Generally**. Imugene will, and will cause its Affiliates, Sublicensees, and subcontractors to, maintain reasonably complete, current, and accurate records of all Development activities conducted by or on behalf of Imugene, and its Affiliates, Sublicensees, and subcontractors, respectively, pursuant to this Agreement and all data and other information resulting from such activities consistent with its usual practices, in validated computer systems that are compliant with 21 C.F.R. §11 and in accordance with Applicable Laws ("*Development Records*"). Such Development Records will fully and properly reflect all work done and results achieved in the performance of the Development activities for the Licensed Products in good scientific manner appropriate for regulatory and patent purposes.
- 4.2.2 **Additional Requirements for Existing Product**. Imugene will maintain all Development Records related to any Existing Product for a period of [***] after the end of the Term. Imugene will document all non-clinical and preclinical studies and Clinical Trials of any Existing Product in formal written study reports in accordance with GLP, cGMP, and cGCP, as applicable, and in compliance with Applicable Law.

(a) Upon Precision's reasonable request, not more frequently than [***] during which Imugene or its Affiliates, Sublicensees, or subcontractors are performing or having performed Development activities for any Existing Product, Imugene will, and will cause its Affiliates, Sublicensees, and subcontractors to, allow Precision to access, review, and copy such records (including access to relevant databases). Precision will have the right to use the data and results generated by or on behalf of Imugene and its Affiliates, Sublicensees, and subcontractors hereunder to Exploit any Existing Product outside of the Licensed Field in the Territory. Imugene will ensure that all records or other documents that it transmits to Precision electronically under this Agreement are transmitted over secure systems that include adequate encryption safeguards to prevent unauthorized access and maintain data security.

(b) Upon Imugene's reasonable request during the Term, not more frequently than [***] during which Precision or its Affiliates, Sublicensees, or subcontractors are performing or having performed development activities for any Existing Product, Precision will allow Imugene to access, review, and copy all data and other information resulting from such activities (including access to relevant databases) that are Controlled by Precision. Imugene will have the right to use such data and information hereunder to Exploit any Existing Product in the Licensed Field in the Territory. Precision will ensure that all data and other information that it transmits to Imugene electronically under this Agreement are transmitted over secure systems that include adequate encryption safeguards to prevent unauthorized access and maintain data security.

4.3**Data Exchange and Use.** In addition to its adverse event and safety data reporting obligations set forth in Section 5.4, each Party will promptly provide the other Party, through the JSC, with copies of all data and results and all supporting documentation (*e.g.*, protocols, Investigator's Brochures, case report forms, analysis plans, and all in English language) Controlled by such Party or its Affiliates (or Sublicensees, in the case of Imugene) that are generated by or on behalf of such Party or its Affiliates (or Sublicensees, as applicable) in the Development of each Existing Product. Imugene will have the right to use and reference such data and results provided by Precision for the purpose of performing Development activities under this Agreement, obtaining, supporting, and maintaining Regulatory Approvals and any Reimbursement Approval, as applicable, of Licensed Products in the Licensed Field in the Territory, without additional consideration. Precision and its designees will have the right to use and reference such data and results provided by Imugene for the purpose of Developing any Existing Product or any other products based on ARCUS Technology, and obtaining, supporting, and maintaining Regulatory Approvals or any Reimbursement Approvals, as applicable, of any such product, in each case in all fields in the Territory, without additional consideration.

4.4**Development Reports**. On an annual basis, during any period in which Imugene is performing, or having performed, Development activities for any Licensed Product, Imugene will provide Precision, at Imugene's sole cost and expense, with reasonably detailed written reports summarizing the Development activities performed during the period since the preceding report, the Development activities in process, and the future activities that Imugene or its Sublicensees or subcontractors expect to initiate. Without limiting the foregoing, such reports will contain sufficient detail to enable Precision to assess Imugene's compliance with its Development diligence obligations set forth in this Article 4 (including, for clarity, to assess the expenditures of Imugene, Parent, and any of their respective Affiliates to assess compliance with Section 4.1.4(b)). Imugene will promptly respond to Precision's reasonable requests from time to time for additional

information regarding significant Development activities for any Licensed Product performed by or on behalf of Imugene or its Affiliates, Sublicensees, or subcontractors.

ARTICLE 5

REGULATORY MATTERS; MEDICAL AFFAIRS

5.1Regulatory Responsibilities.

5.1.1 **Existing Product Outside the Licensed Field**. As between Precision and Imugene, Precision shall be solely responsible for any and all regulatory activities with respect to any Existing Product outside the Licensed Field, including filing of all Regulatory Filings for any Existing Product, maintenance of all Regulatory Approvals, any reports or submissions required to be made to any non-governmental Third Party payors, and any and all regulatory matters arising after obtaining Regulatory Approval, including post-marketing inquiries and safety surveillance activities. Precision shall keep Imugene reasonably apprised of such activities.

5.1.2 Existing Product in the Licensed Field.

- (a) As between Precision and Imugene, subject to the terms of this Agreement, Imugene shall be responsible for regulatory activities with respect to any Existing Product in the Licensed Field in the Territory, and shall use Commercially Reasonable Efforts to prepare any and all Regulatory Filings for all indications in the Licensed Field. Imugene shall submit to Precision each such proposed Regulatory Filing prepared for Precision's comments and review, such review and comments to be transmitted to Imugene as soon as reasonably possible. In addition, the JSC shall establish mechanisms or guidelines for the interactions of the Parties with respect to the submission of Regulatory Filings for any Existing Product in the Licensed Field in the Territory, including the submission of information as a follow up to any such submitted Regulatory Filing, which mechanisms or guidelines shall be consistent with the Parties' obligations described in this Section 5.1.
- (b) To the extent possible, and as soon as reasonably possible, each Party shall provide to the JSC reasonable written notice of all meetings and conference telephone calls with any Regulatory Authority in which matters that would be expected to relate to an Existing Product will be discussed. Precision shall have the right to have reasonable representation present at all such meetings and have reasonable representation attend each such conference telephone call with any Regulatory Authority in which matters that would be expected to relate to an Existing Product will be discussed, in each case to the extent permitted by such relevant Regulatory Authority.
- (c) Each Party shall notify the JSC within [***] after it receives information about the initiation of any investigation or inquiry by any Regulatory Authority concerning the Development, Manufacture or Commercialization of an Existing Product to the extent such investigation or inquiry would be reasonably likely to adversely affect the other Party.
- (d) If a Regulatory Authority desires to conduct an inspection or audit with regard to an Existing Product or Imugene's facility or a facility under contract with Imugene or its Affiliate with respect to the activities relevant to this Agreement, Imugene shall permit and cooperate with such inspection or audit, and shall cause the contract facility to permit and

cooperate with such Regulatory Authority during such inspection or audit. Imugene shall conform its activities under this Agreement to any commitments made in such a response, except to the extent that Imugene believes in good faith that such commitments violate Applicable Laws.

- (e) If any Regulatory Authority takes or gives notice of its intent to take any regulatory action with respect to any activity of either Party or its Affilaite relating to any Existing Product, then such Party will notify the JSC of such contact, inspection, or notice or action within [***] after receipt of such notice (or, if action is taken without notice, within [***] of Imugene becoming aware of such action). Such Party will have the final decision-making authority with respect to [***]. The costs and expenses of any such regulatory action will be borne by such Party.
- (f) If either Party believes that the other Party or its Affiliate is taking or intends to take any action with respect to an Existing Product that could have a material adverse impact upon the regulatory status of any Existing Product, [***].
- 5.1.3 **Additional Products**. Subject to the terms of this Agreement, as between the Parties, Imugene shall have sole responsibility for and control of the preparation, submission, and maintenance of all Regulatory Filings and obtaining and maintaining all Regulatory Approvals (including the preparation and submission of the IND filing and for seeking IND approval) with respect to Additional Products in the Licensed Field, and shall have sole control over all related interactions with the applicable Regulatory Authority, including all correspondence to or with the applicable Regulatory Authority. Imugene will own all right, title and interest in and to any and all Regulatory Filings and Regulatory Approvals for Additional Products in the Licensed Field and, as between the Parties, all such Regulatory Filings and Regulatory Approvals will be held in the name of Imugene.
- 5.1.4 **ARCUS Nuclease Matters**. Notwithstanding Imugene's sole responsibility with respect to regulatory matters involving Additional Products, Precision shall have the right, prior to BLA approval for each Additional Product, to have its employees attend each INTERACT meeting or pre-IND submission meeting, the end of the Phase II Clinical Trial meeting for such Additional Product, and any other meeting with the FDA or EMA if such other meeting has any item on the agenda directed to the manufacturing, quality, safety (including non-clinical safety related to production of ARCUS Nucleases) or delivery of ARCUS Nucleases or ARCUS Technology (collectively, "*ARCUS Regulatory Matters*"). Prior to BLA approval for each Additional Product, Imugene will provide drafts of its communications with the FDA and EMA (including with respect to CMC-related matters) to the extent they relate to ARCUS Regulatory Matters to Precision for review and comment, and will consider Precision's comments in good faith and not unreasonably reject any such comments, before submitting such communications to the FDA or EMA. Following BLA approval for each Additional Product, Imugene shall provide Precision notice regarding any communications from Regulatory Authorities regarding ARCUS Regulatory Matters.
- 5.2**Regulatory Costs**. Imugene shall bear all costs and expenses it incurs to conduct all regulatory activities under this Agreement.
- 5.3**Right of Reference**. Each Party hereby grants, and shall cause its Affiliates (and Sublicensees, in the case of Imugene) to grant, at no cost, to the other Party, its Affiliates and any of their respective licensees (in the case of Precision) or Sublicensees (in the case of Imugene) a Right of Reference to any data and Regulatory Filings Controlled by the granting Party or its

Affiliates (or Sublicensees, in the case of Imugene) that relates to any Existing Product that the other Party reasonably believes may be necessary or useful to the Development, Manufacture or Commercialization of any Existing Product in such other Party's respective field (i.e., in the Licensed Field, in the case of Imugene, or outside the Licensed Field, in the case of Precision), and the granting Party will provide, and shall cause its Affiliates (and Sublicensees, in the case of Imugene) to provide, a signed statement to the foregoing effect, as reasonably requested by the other Party.

5.4Adverse Event Reporting; Pharmacovigilance Agreement.

5.4.1 **Generally**. As between the Parties, Imugene shall be responsible for the timely reporting of all relevant adverse drug reactions/experiences, product quality, product complaints and safety data relating to Licensed Products in the Licensed Field to the appropriate Regulatory Authorities in the Territory, in each case in accordance with Applicable Laws of the relevant countries and Regulatory Authorities. Imugene shall be solely responsible for all costs and expenses it incurs to conduct its pharmacovigilance responsibilities. At the reasonable request of either Party, the Parties shall enter into a pharmacovigilance agreement on terms that comply with ICH guidelines, including: (i) providing detailed procedures regarding the maintenance of core safety information and the exchange of safety data relating to any Existing Product or the Licensed ARCUS Nuclease worldwide within appropriate timeframes and in an appropriate format to enable each Party to meet both expedited and periodic regulatory reporting requirements; and (ii) ensuring compliance with the reporting requirements of all applicable Regulatory Authorities on a worldwide basis for the reporting of safety data in accordance with standards stipulated in the ICH guidelines, and all applicable regulatory and legal requirements regarding the management of safety data.

5.4.2 **Right to Audit for Existing Product**. Each Party shall have the right to perform audits of the other Party's pharmacovigilance activities relating to the Parties' activities in relation to any Existing Product under the terms of this Agreement including compliance by the other Party with Applicable Laws. The frequency of such audits will be no more than [***] in any [***] period during the Term; provided that such audits may be more frequent if, in the auditing Party's sole discretion, more frequent audits are necessary by a risk-based approach, and except in 'for cause' situations where, in the event of a serious or potentially serious issue, additional audits may be conducted. The notification of one Party's intent to conduct such an audit will be provided in writing to the other Party within a reasonable time period in advance, based upon the particular circumstances of the situation.

5.5**Product Withdrawals and Recalls**. In the event that (a) an event, incident, or circumstance has occurred which may result in the need for a recall or other removal of an Existing Product or any lot or lots thereof from the market in the Territory; (b) any Regulatory Authority in the Territory threatens or initiates any action to remove an Existing Product from the market in the Territory; or (c) any Regulatory Authority in the Territory requires distribution of a "Dear Doctor" letter or its equivalent, regarding use of the Existing Product in the Territory, Imugene shall promptly advise Precision in writing with respect thereto, and shall provide to Precision copies of all relevant correspondence, notices, and any other related documents. Unless otherwise agreed by the Parties, Imugene shall be responsible for conducting a recall of any Existing Product in the Licensed Field. No recall shall be commenced in the Territory without Precision's prior written

consent, such consent not to be unreasonably withheld, conditioned or delayed. Each Party will cooperate with the other Party in the performance of any recall or withdrawal.

5.6Medical Affairs No later than [***] prior to the anticipated date of performance of Medical Affairs activities for an Existing Product in the Territory, Imugene will prepare an initial draft of each Medical Affairs Plan for the applicable Existing Product(s) and provide such initial draft to the JSC to review and discuss. The Medical Affairs Plan(s) will contain a high-level summary of the major Medical Affairs activities to be undertaken by Imugene for the applicable Existing Product(s) in the Licensed Field in the Territory and the estimated timelines for performing such activities. Thereafter, from time to time, but at least annually, Imugene will propose updates to the Medical Affairs Plan(s) for the applicable Existing Product(s) in the Licensed Field in the Territory to reflect changes in such plan(s), including to account for relevant factors that may influence such plan and the Medical Affairs activities set forth therein and provide each such update to the JSC to review and discuss. For each Calendar Quarter in which any Medical Affairs are conducted by or on behalf of Imugene or its Affiliates or Sublicensees for any Existing Product in the Licensed Field in the Territory, Imugene will provide updates on Medical Affairs activities at each meeting of the JSC. The Parties recognize that each Party may benefit from the coordination of certain Medical Affairs activities for the Existing Product(s) inside and outside of the Licensed Field. Accordingly, the Parties will coordinate such activities through the JSC where appropriate.

ARTICLE 6

COMMERCIALIZATION

6.1Existing Product.

- 6.1.1 **Principles of Commercialization**. The Parties intend for Imugene to Commercialize the Existing Product in the Licensed Field in the Territory, following Regulatory Approval thereof, as set forth in this Section 6.1. Each Party shall appoint a representative to be such Party's single point of contact to facilitate information flow between the Parties relating to each Party's experience and relationships in the Licensed Field (in the case of Imugene) and outside the Licensed Field (in the case of Precision). Each Party shall first address any communications relating to Commercialization by the other Party to such representatives unless otherwise agreed to by the Parties on a case-by-case basis. Such representatives shall, without limitation, coordinate direct involvement or meetings with subject matter experts within each Party's internal organization and/or its field account management organization. Notwithstanding the foregoing, Precision's representative shall not be required to provide details relating to any customer specific transaction or agreement.
- 6.1.2 **Commercialization Activities**. Imugene shall (i) use Commercially Reasonable Efforts to Commercialize [***] Existing Product in the Licensed Field following Regulatory Approval thereof in the Licensed Field throughout the Territory; and (ii) use Commercially Reasonable Efforts to perform other activities not otherwise identified herein but which are required by Regulatory Authorities to Commercialize [***] Existing Product in any indication in the Licensed Field for which Regulatory Approval has been obtained in the Territory.
 - 6.1.3 Advertising and Promotional Materials.
 - (a) **Imagene Promotional Materials**. Imagene will be responsible for

development of all advertising and promotional materials, programs and initiatives related to the use of any Existing Product in the Licensed Field in the Territory, including medical education, symposia, opinion leader development, peer-to-peer development, publications, journal ads, and all other written communications that describe the features or benefits of the Product, in each case in the Licensed Field in the Territory (the "*Imugene Promotional Materials*"). All Imugene Promotional Materials shall be prepared in accordance with Applicable Law, Imugene's policies for compliance with Applicable Laws, industry guidelines relating to promotional and advertising materials, any requirements of the FDA imposed as a condition of any Regulatory Approval, industry marketing codes such as the PhRMA code, and implementation guidelines to be mutually agreed upon by the Parties. Imugene shall implement appropriate policies and procedures relating to safety reporting, approval of Imugene Promotional Materials, sales force training and similar matters.

- (b) **Imugene's Compliance Policies**. Imugene, on Precision's request, shall provide Precision copies of and access to Imugene's policies for compliance with Applicable Law relating to promotional and advertising materials, and Imugene's procedures relating to the approval of promotional materials, sales force compliance training, and related matters. Precision shall have the right to audit Imugene's materials and procedures, no more than [***].
- 6.1.4 **Complaints and Inquiries**. The Parties shall mutually develop a protocol for responding to any and all complaints, medical questions, or other inquiries relating to any Existing Product in the Licensed Field in the Territory, which are directed to such Parties' respective sales representatives. Imugene shall be responsible for responding to complaints, medical questions, or other inquiries relating to the Imugene Commercialization Activities and Precision or its designee shall be responsible for responding to all other complaints, medical questions, or other inquiries. Imugene shall notify Precision of, and provide to Precision all pertinent information in Imugene's possession relating to, any and all suspected or actual tampering, counterfeiting, or contamination or other similar problems with respect to any Existing Product in the Licensed Field in the Territory.

6.2Additional Products.

- 6.2.1 **Commercialization of Additional Products**. Imugene shall have the sole right and responsibility for, and shall bear all costs and expenses associated with, the Commercialization of Additional Products, including Manufacturing, distribution, marketing, and sales activities, in the [***]. Subject to the terms of this Agreement, all decisions concerning Commercialization of Additional Products in the [***], including the marketing and sales of Additional Products, and the design, price, and promotion of Additional Products, shall be within Imugene's sole discretion.
- 6.2.2 **Diligence Efforts for Additional Products**. After receipt of Regulatory Approval, Imugene shall use Commercially Reasonable Efforts to Commercialize each Additional Product for which it obtains Regulatory Approval in the United States and each Major Foreign Market.
- 6.3**Reports**. On an annual basis, Imugene will be obligated to deliver to Precision a report describing the status of Imugene's and its Affiliates and Sublicensees' Commercialization efforts with respect to Licensed Products in the Licensed Field in the Territory. In addition, Precision may from time to time provide Imugene with written requests describing specific types

of information Precision requires in order to comply with Precision's reporting and disclosure obligations under any Applicable Laws, and Imugene shall include such information in such reports.

6.4Compensation for Sales Outside the Licensed Field. If Precision believes that there are material sales of any Existing Product outside the Licensed Field in the Territory, Precision shall be permitted to implement and conduct procedures under which material sales and purchases of any Existing Product in the Territory and other related market research data shall be audited and monitored, using for example IQVIA data and information, and Imugene agrees to cooperate with Precision in the implementation and conduct of such procedures. In the event that such an audit and monitoring procedure determines that material sales of any Existing Product outside the Licensed Field in the Territory have been or are being made by or for the benefit of Imugene, [***]. Notwithstanding anything to the contrary, the remedy to Precision set forth in the previous sentence will not be the exclusive remedy available to Precision for such sales.

ARTICLE 7

MANUFACTURING

7.1 **Licensed Products**. Subject to the terms of this Agreement, Imugene shall be solely responsible, at its sole cost and expense, for all Manufacturing (or having Manufactured through a CMO), including development of any Chemistry, Manufacturing and Controls sections of any Regulatory Materials or Regulatory Approval, for all Licensed Products for Imugene's, its Affiliates' and Sublicensees' pre-clinical and clinical Development and Commercialization in the Licensed Field in the Territory under this Agreement.

7.2 Manufacturing Technology Transfer.

- 7.2.1 Upon written request by Precision, Imugene shall cooperate with Precision or its designee (at Precision's expense) to conduct a manufacturing technology transfer sufficient to enable Precision, its Affiliate or its designee (or a CMO engaged by any of them) to Manufacture (or have Manufactured by such CMO) each Existing Product, or any CAR-T product that is not an Additional Product ("*Manufacturing Technology Transfer*") for Development and Commercialization pursuant to the applicable Technology Transfer Plan for such product.
- 7.2.2 In connection with the Manufacturing Technology Transfer following the applicable request by Precision under Section 7.2.1, the Parties shall promptly agree to a technology transfer plan with reasonable access to Imugene personnel (including reasonable caps on hours of access) and facilities, for the Manufacturing Technology Transfer for the applicable product to Precision, its Affiliate or its designee (or a CMO engaged by any of them), in a form reasonably acceptable to the Parties (the "*Technology Transfer Plan*"). Any such technology transfers shall be overseen by a working group of the JSC established for such purposes.
- 7.3**Clinical Supply**. At Precision's request, [***], Imugene will enter into one or more agreement(s) with Precision or its designee (at Precision's sole discretion) on commercially reasonable terms providing for the supply of [***], together with an appropriate quality agreement setting forth Imugene's quality and compliance obligations with respect to the manufacture and supply of the applicable product (collectively, a "*Clinical Supply Agreement*"). [***]

ARTICLE 8

LICENSE RIGHTS

8.1License Grants to Imugene.

- 8.1.1 **Exclusive License**. Subject to the terms and conditions of this Agreement, Precision (on behalf of itself and its Affiliates) hereby grants to Imugene an exclusive (even as to Precision and its Affiliates), royalty-bearing (as set forth in Section 9.3), license, with the right to grant sublicenses (through multiple tiers, as provided in Section 8.4), under the Precision Product IP, to Exploit Licensed Products in the Licensed Field in the Territory; provided, however that the foregoing license shall be non-exclusive with respect to Manufacture of any Existing Product. Notwithstanding the foregoing, Precision or its designee may conduct Research and other Development activities with any Existing Product, including in connection with Exploitation of such Existing Product outside the Licensed Field; provided that Precision shall not have the right to conduct any clinical study of an Existing Product in the Licensed Field in the Territory, other than to transition the Ongoing Precision Trial to Imugene in accordance with the other Transaction Documents, without the prior written consent of Imugene.
- 8.1.2 **Non-Exclusive Licenses**. Subject to the terms and conditions of this Agreement, Precision (on behalf of itself and its Affiliates) hereby grants to Imugene a non-exclusive, royalty-bearing (as set forth in Section 9.3) license, with the right to grant sublicenses (through multiple tiers, as provided in Section 8.4), under the Precision Platform IP, to Exploit Licensed Products in the Licensed Field in the Territory. The license set forth in this Section 8.1.2 under Precision Platform IP is intended to provide Imugene a "freedom to operate" license with respect to the Precision Platform IP solely for the Exploitation of Licensed Products in the Licensed Field, and not for Imugene's independent use of the Precision Platform IP. Imugene acknowledges and agrees that Imugene will not have any right to (a) access or receive any ARCUS Technology, except as expressly set forth with respect to ARCUS Research Tools in Section 8.2, (b) design, create, select, or optimize any ARCUS Nucleases using the ARCUS Technology, or (c) otherwise use the ARCUS Technology as a genome engineering tool; in the case of (a) and (c), except to the extent that the ARCUS Technology is embodied in the Licensed ARCUS Nuclease or any Licensed Product and utilized solely in Imugene's practice of the licenses granted in Section 8.1.1. The Parties agree that ARCUS Technology will not be transferred to Imugene or its designee under this Agreement except as expressly set forth with respect to ARCUS Research Tools in Section 8.2.
- 8.1.3 **Restrictions on Licensed ARCUS Nuclease**. Imugene acknowledges and agrees that the foregoing license does not include any right to, and Imugene shall not, and shall not permit any of its Affiliates or its or their Sublicensees to (a) modify the Licensed ARCUS Nuclease, or (b) [***], in each case (a) and (b), without Precision's prior written consent.
- 8.2**ARCUS Research Tools Transfer**. Within [***] following the Effective Date, Precision will make available to Imugene Precision Background IP reasonably sufficient to enable the use of the ARCUS Research Tools under the licenses granted to Imugene in Section 8.1.

8.3License Grant to Precision. Subject to the terms and conditions of this Agreement, Imugene agrees to grant and hereby grants (on behalf of itself and its Affiliates) to Precision (a) a perpetual, fully-paid, royalty-free, non-exclusive license, with right to grant sublicenses through multiple tiers, under all Imugene Arising IP and any Imugene Background IP that is necessary or reasonably useful for the applicable Existing Product, or its use or manufacture, to Exploit any Existing Product in all fields in the Territory, and (b) a perpetual, fully-paid, royalty-free, non-exclusive license, with right to grant sublicenses through multiple tiers, under all Imugene Arising IP to Exploit all other products, other than the Licensed Products, in all fields in the Territory. Notwithstanding the non-exclusive nature of the foregoing license, Imugene shall not Research, Develop (including conduct of any Clinical Trial) or otherwise Exploit any Licensed Products outside the Licensed Field. Precision shall not: (y) practice the license granted to it under clause (a) in the Licensed Field during the Term unless the Existing Product has become a Terminated Product in accordance with Article 14, and (z) practice the license granted to it under clause (b) in the Licensed Field during the Term.

8.4Third Party Sublicenses.

8.4.1 **Generally**. Imugene and Precision may grant one or more sublicenses under the rights and licenses granted to it under Section 8.1 (in the case of Imugene) or Section 8.3 (in the case of Precision), in full or in part, to Third Parties (with the right to sublicense through multiple tiers); *provided*, that: (a) any such permitted sublicense is consistent with and subject to the terms and conditions of this Agreement, including the confidentiality provisions of Article 13 and the intellectual property provisions of Article 10 (in the case of Imugene); and (b) the Party granting such sublicense shall remain responsible for performance of such Party's obligations under this Agreement and shall be responsible for all actions of each such sublicensee as if such sublicensee were the Party hereunder.

8.4.2 **By Imugene**. Imugene will not grant any sublicense or other right that permits any Research, Development or Commercialization of any Existing Product by any Third Party without Precision's prior written consent, provided that Imugene may grant any sublicense or other right, without Precision's prior written consent, to (a) a sublicensee that is a pharmaceutical or biotechnology company having [***], or (b) a contract distributor, Third Party contractor or service provider, including a CMO, in order to provide services for a fee for the benefit of Imugene. Without limiting the foregoing, any sublicense or other right must include in the written agreement pursuant to which such sublicense or other right is granted provisions ensuring that (x) the Existing Product(s) are Exploited in a manner consistent with the requirements set forth in this Agreement, (y) Precision is an intended third party beneficiary to such agreement and (z) all rights attaching therefrom in relation to any activities contemplated by this Agreement and the right to enforce the provisions of such agreement against the applicable Third Party are vested in Precision. To the extent required by the Cellectis Agreement, each sublicense granted by Imugene under any Patents within Precision Product IP must grant the same scope of rights for all Patents within Precision Platform IP. Any purported sublicense or other right granted by Imugene that is not in compliance with the requirements of this Section 8.4.2 shall be null and void. Imugene shall deliver a copy of each sublicense, or amendment thereto, to Precision promptly following the execution thereof.

8.5Retention of Rights; No Implied Rights. Except as expressly set forth in this Agreement, neither Party shall be granted, by implication, estoppel or otherwise, any license or right to or under any other intellectual property interest, including any trademarks, Know-How, or Patents, of the other Party. The licenses granted by Precision to Imugene hereunder do not include any rights with respect to other products or therapies with which a Licensed Product may be combined or any other products or therapies other than the Licensed Products under this Agreement. Each party covenants that it will not use or practice any of the other Party's intellectual property rights licensed to it under this Agreement except for the purposes expressly permitted in the applicable license grant. Imugene agrees to impose the foregoing covenant in this Section 8.5 on all of its Affiliates and sublicensees.

8.6Existing In-License Agreements.

- 8.6.1 For clarity, the license granted to Imugene in Section 8.1 includes a sublicense under certain Duke IP and Cellectis Patents.
- 8.6.2 **Cellectis Patents.** Imugene acknowledges and agrees that rights under certain Precision Patents are licensed to Precision by Cellectis S.A. (the "*Cellectis Patents*") under that certain Patent Cross-License Agreement between Cellectis S.A. ("*Cellectis S.A.*") and Precision dated January 23, 2014 (the "*Cellectis Agreement*"), and, notwithstanding any exclusive license granted to Imugene under this Agreement, (a) Cellectis S.A. retains rights under the Cellectis Patents and is not restricted from granting rights to Third Parties under the Cellectis Patents, (b) any licenses and rights granted by Precision to Imugene under the Cellectis Patents are granted only within the permissible scope of sublicenses granted under the Cellectis Agreement, and (c) pursuant to the Cellectis Agreement, Cellectis S.A. retains non-exclusive rights under certain Precision Patents identified in the Cellectis Agreement, which may be further sublicensed by Cellectis S.A. without Precision control or consent. Imugene acknowledges and agrees that any exercise of any right by Cellectis S.A., or by any Third Party through Cellectis S.A., under the Cellectis Agreement shall not constitute a breach of this Agreement by Precision. Attached hereto as Exhibit 8.6.2 is that certain list of certain Cellectis Patents that was attached to the Cellectis Agreement on its execution date, which, for clarity, is not the complete list of the Cellectis Patents as of the Effective Date.
- 8.6.3 **Duke IP.** Imugene acknowledges and agrees that any licenses and rights granted by Precision to Imugene under the Duke IP are granted subject to the terms and conditions of the Duke Agreement, including Duke's right to practice under the Duke IP for its own internal, non-commercial, educational, research and clinical purposes, and subject to the rights of the United States Government and applicable limitations under 37 C.F.R. § 401, Public Law 96-517 and Public Law 98-620 resulting from the United States Government's funding of research leading to creation of the Duke IP. Without limiting the foregoing, Imugene agrees to comply with any obligations resulting from such government rights with respect to its practice of the Duke IP (if any) under this Agreement.
- 8.6.4 **Other Third Party IP**. In the event that, after the Effective Date, any Know-How or Patent licensed to Precision by a Third Party (other than the Duke IP or Cellectis Patents) becomes necessary or reasonably useful for the Exploitation of a Licensed Product [***], then the Parties would discuss in good faith the terms pursuant to which Precision would grant a sublicense

to Imugene under such Know-How or Patent, and subject to and effective upon the Parties' mutual written agreement to such terms, such Know-How or Patent would be sublicensed by Precision to Imugene. For the avoidance of doubt, this Section 8.6.4 does not apply to the Duke IP, Cellectis Patents or Existing In-License Agreements.

8.7Existing Product Outside the Licensed Field. [***]

8.8**Consideration**. The Parties acknowledge that each of the licenses and rights granted by Precision in this Agreement and each of the provisions of this Agreement for efforts or assistance by Precision and access to Precision Technology, individually and collectively, constitute good, valuable and sufficient consideration for each and all of the fees and payments called for hereunder and for each and all of the other obligations of Imugene, its Affiliates and its and their Sublicensees; and the Parties further acknowledge that the individual and collective rights under and access to Precision Technology renders the way in which those fees and payments hereunder are determined, their amount (and potential reduction) and their duration, appropriate and desirable as a matter of convenience.

ARTICLE 9

FEES, ROYALTIES, & PAYMENTS

9.1Regulatory and First Commercial Sale Milestones. As partial consideration for the rights granted by Precision to Imugene hereunder with respect to Licensed Products, Imugene shall pay to Precision or its designee the following milestone payments in the corresponding amount set forth in the right-hand column of the table immediately below upon the first achievement of each of the following milestone events in the left-hand column of the table immediately below by Imugene, its Affiliates or Sublicensees. Each such milestone payment based on the occurrence of a First Commercial Sale shall be referred to in this Agreement as a "First Commercial Sale Milestone Payment". The Expansion Cohort Success Milestone and Initiation Milestone are intended to be sequential; achievement of the Initiation Milestone prior to achievement of the Expansion Cohort Success Milestone.

Regulatory or First Commercial Sale Milestone Event	Milestone Payment (USD)
[***] (the "Expansion Cohort Success Milestone")	\$8,000,000 (payable in cash or a mix of cash and Ordinary Shares, provided that at least [***] will be paid in cash)
[***] ("Initiation Milestone")	[***]
[***]	[***]
[***]	[***]
[***]	[***]
[***]	[***]
[***]	[***]
[***]	[***]

[***]	[***]
[***]	[***]

- 9.1.1 Imugene shall notify Precision in writing no later than [***] after the achievement of each Milestone Event set forth in the table above and shall make the corresponding milestone payment within [***] after receipt by Imugene of an invoice from Precision delivered after such achievement; provided, however, that with respect to the milestone payment for the Expansion Cohort Success Milestone, if Imugene elects to pay a portion of such milestone payment in Ordinary Shares, Imugene shall comply with, and the payment for the Expansion Cohort Success Milestone shall be subject to, Section 9.1.2; and provided, further, however, that for the first achievement of the [***] Milestone Event above, Imugene may make the corresponding milestone payment within [***] after receipt of the applicable invoice from Precision.
- 9.1.2 Upon achievement of the Expansion Cohort Success Milestone, Imugene shall, in each case within [***] after receipt by Imugene of an invoice from Precision delivered after such achievement (the date of payment, "Expansion Cohort Success Milestone Payment Date"), either, at Imugene's discretion, (i) pay Precision \$8,000,000 in cash or (ii) pay Precision and issue (or cause to be issued) to Precision a combination of cash (with the amount of cash not being less than [***]) and Ordinary Shares that equal, in the aggregate, \$8,000,000 (the "Expansion Cohort Success Milestone Consideration," with any such Ordinary Shares issued the "Expansion Cohort Success Milestone Shares" and any such cash paid the "Expansion Cohort Success Milestone Cash"). For purposes of determining the number of Expansion Cohort Success Milestone Shares to be issued, the value of such shares shall be based on the ten (10) Trading Day VWAP of Ordinary Shares immediately prior to the date [***] converted into Dollars using the Reference Rate. Notwithstanding the foregoing, in the event that Imugene elects to pay any portion of the Expansion Cohort Success Milestone Consideration in the form of Expansion Cohort Success Milestone Shares, Imugene shall satisfy and comply with each of the following conditions, and if any of the following conditions are not satisfied on the Expansion Cohort Success Milestone Payment Date (other than the requirement in Section 9.1.2(f) to issue the cleansing notice/prospectus, which condition must be satisfied within [***] following the Expansion Cohort Success Milestone Payment Date), Imugene shall be required to pay the Expansion Cohort Success Milestone Consideration entirely as Expansion Cohort Success Milestone Cash:
- (a) the representations and warranties of Imugene and Parent set forth in this Agreement and in the other Transaction Documents shall be true, correct and complete in all material respects as of the Expansion Cohort Success Milestone Payment Date;
 - (b) Parent shall be in compliance in all material respects with the Registration Rights Agreement;
- (c) Parent shall have sufficient placement capacity under ASX Listing Rule 7.1 to issue the Expansion Cohort Success Milestone Shares;
- (d) Imugene shall have, on or before the issuance of the Expansion Cohort Success Milestone Shares, taken all steps necessary to procure that the Expansion Cohort Success Milestone Shares issued on the Issue Date are quoted on the financial market operated by the ASX (including making an application for quotation within [***] of issue of the Expansion Cohort

Success Milestone Shares) and cause to be issued to Precision by Parent a certificate or holding statement for the Expansion Cohort Success Milestone Shares:

(e) in connection with and upon any issuance of Expansion Cohort Success Milestone Shares, Parent shall have (i) lodged a notice with ASX that complies with sections 708A(5) and 708A(6) of the Australian Corporations Act or a prospectus that complies with section 713 of the Australian Corporations Act on the Issue Date; and (ii) ensured that each Expansion Cohort Success Milestone Share issued on the Issue Date is freely transferable (including being free of any restriction under section 707(3) of the Australian Corporations Act) without any requirement for disclosure to investors under Part 6D.2 of the Australian Corporations Act on and from the Issue Date; and

(f) the Expansion Cohort Success Milestone Shares, when issued and delivered in accordance with the terms of this Agreement, will (i) be duly and validly issued, fully paid and nonassessable; (ii) be free of any liens, options, encumbrances, proxies, adverse claims or restrictions; (iii) rank equally in all respects with the then existing issued Ordinary Shares unless otherwise approved in writing by Precision; (iv) be issued in compliance with all U.S., Australian and other applicable securities laws, regulations and listing requirements and (v) be immediately and freely tradable by Precision, without restriction, on the ASX and, if Parent has listed on Nasdag, on the Nasdag, and Parent shall have taken all actions necessary in connection therewith (including, if requested by Precision, depositing the Expansion Cohort Success Milestone Shares with the Depositary (as defined in the Registration Rights Agreement) and causing the issuance of ADSs to Precision in respect of the Expansion Cohort Success Milestone Shares), all at the sole cost and expense of Imugene; provided that the Parties acknowledge that Parent will be required to file a cleansing notice/prospectus in order for the shares to be freely tradable which Parent will do as soon as practicable and by no later than [***] following the Expansion Cohort Success Milestone Payment Date.

On allotment and issue of any Expansion Cohort Success Milestone Shares, Precision authorizes Parent to enter Precision's name in Parent's register of shareholders and agrees to be bound by the terms of Parent's constitution.

9.2 Commercial Milestones. As partial consideration for the rights granted by Precision to Imugene hereunder with respect to Licensed Products, Imugene shall pay to Precision the following milestone payments in the corresponding amount set forth in the right-hand column of the table immediately below (each, a "Commercial Milestone Payment") upon the first achievement of each of the following milestone events in the left-hand column of the table immediately below by Imugene, its Affiliates or Sublicensees. For purposes of determining whether the Net Sales thresholds in the table below have been achieved for a Licensed Product, all Net Sales of such Licensed Product shall be aggregated globally for all sales made by Imugene or any of its Affiliates or its or their Sublicensees of such Licensed Product, in any and all preparations, formulations, dosages, packaging or methods of administration thereof.

Commercial Milestone Event	Milestone Payment (USD)
[***]	[***]

[***]	[***]
[***]	[***]
[***]	[***]
[***]	[***]
[***]	[***]

Imugene shall promptly notify Precision in writing of the achievement of each Milestone Event set forth in the table above within [***] after the end of the Calendar Year in which such milestone has been achieved and shall make the corresponding milestone payment within [***] after receipt by Imugene of an invoice from Precision delivered after such achievement. Achievement of each Milestone Event measured by Net Sales with respect to a Licensed Product shall result in achievement of all Milestone Events measured by a lower amount of Net Sales of such Licensed Product.

9.3Royalties.

9.3.1 **Royalty Term**. Imugene shall pay Precision royalties as set forth in this Section 9.3 on a Licensed Product-by-Licensed Product and country-by-country basis in the Territory during the period of time beginning on the date of the First Commercial Sale of such Licensed Product in such country and continuing until the latest to occur of: (a) the expiration of the last-to-expire Valid Claim in such country Covering such Licensed Product; (b) the expiration of any period of data, regulatory, or market exclusivity, or supplemental protection certificates (other than Patents) covering the Licensed Product in such country; and (c) ten (10) years after the First Commercial Sale of such Licensed Product in such country (the "*Royalty Term*").

9.3.2 **Royalty Rates**. On a Licensed Product-by-Licensed Product and country-by-country basis, during the Royalty Term, Imugene shall pay to Precision a royalty equal to the percentages of aggregate annual global Net Sales of such Licensed Product, as set forth below (the "*Royalty*"), calculated by multiplying the applicable royalty rate percentage by the corresponding portion of aggregate global Net Sales for such Licensed Product in such Calendar Year. For purposes of determining whether the Net Sales thresholds in the table below have been achieved for a Licensed Product, all Net Sales of such Licensed Product shall be aggregated globally for all sales made by Imugene or any of its Affiliates or its or their Sublicensees of such Licensed Product, in any and all preparations, formulations, dosages, packaging or methods of administration thereof.

Annual Net Sales of the Applicable Licensed Product	Royalty Rate
Aggregate annual global Net Sales of an Existing Product less than [***]	[***]
Aggregate annual global Net Sales of an Existing Product equal to or greater than [***] but less than [***]	[***]
Aggregate annual global Net Sales of an Existing Product equal to or greater than [***]	[***]
Aggregate annual global Net Sales of an Additional Product less than [***]	[***]
Aggregate annual global Net Sales of an Additional Product equal to or greater than [***] but less than [***]	[***]
Aggregate annual global Net Sales of an Additional Product equal to or greater than [***]	[***]

- 9.3.3 **Valid Claim**. If, at the time a Licensed Product is sold in a country during the Royalty Term for such Licensed Product, there is no longer a Valid Claim that Covers such Licensed Product in such country, the Royalty rates provided in Section 9.3.2 above for the sale of such Licensed Product in such country will be reduced in such country by [***].
- 9.4**Acquisition Fee**. Without limiting any other payments due to Precision, Imugene would make the following payments to Precision upon the occurrence of the following events ("*Acquisition Fees*"):
- 9.4.1 **Change of Control**. Upon any Change of Control of Imugene, Parent or any of their respective Affiliates in which [***], if, as of the date of such Change of Control, [***], then [***] Imugene shall pay to Precision [***] of all Change of Control Consideration.
- 9.4.2 **Product Partnering Transaction**. Imugene shall pay to Precision [***] of all Proceeds from each Product Partnering Transaction.
- 9.5**Payment; Reports**. Royalty payments due by Imugene to Precision under Section 9.3 and Acquisition Fees payable by Imugene to Precision under Section 9.4 shall be: (a) calculated and reported for each Calendar Quarter; (b) paid within [***] after the end of each Calendar Quarter; and (c) accompanied by a report setting forth, with respect to each Calendar Quarter, on a Licensed Product-by-Licensed Product and country-by-country basis: (i) Net Sales of the Licensed Product by the applicable Selling Party(ies) in the Territory, (ii) a calculation of the Royalty due by Imugene to Precision on such Net Sales, and (iii) a calculation of Acquisition Fees payable by Imugene to Precision.
- 9.6**Method of Payment; Currency Conversion**. Unless otherwise agreed by the Parties, all payments due under this Agreement shall be paid in Dollars by wire transfer or electronic funds transfer of immediately available funds to an account designated by the payee;

provided however, that a Party shall only be required to disburse funds to the payee's jurisdiction of incorporation or to a jurisdiction in which the payee has a significant business presence. When conversion of payments from any currency other than Dollars is required, such Party's then-current standard exchange rate methodology will be employed for the translation of foreign currency sales into Dollars; provided, that this methodology is used by such Party in the translation of its foreign currency operating results, is consistent with U.S. GAAP or IFRS, as applicable, is audited by such Party's independent certified public accountants in connection with the audit of the consolidated financial statements of such Party, and is used for external reporting of foreign currency operating results.

9.7**Records and Audits**. Imagene shall maintain complete and accurate records in sufficient detail to permit Precision to confirm the accuracy of Commercial Milestone Payments, Royalty payments and Acquisition Fees payable under this Agreement. Upon reasonable prior notice, such records shall be open during regular business hours for a period of [***] from the creation of individual records, for examination at Precision's expense, and not more often than [***], by an independent certified public accountant selected by Precision and reasonably acceptable to Imugene for the sole purpose of verifying for Precision the accuracy of the financial statements or reports furnished by Imugene pursuant to this Agreement or of any payments made, or required to be made, by Imugene to Precision pursuant to this Agreement. No Calendar Quarter shall be subject to audit more than one time. Any such auditor shall not disclose Imugene's Confidential Proprietary Information to Precision, except to the extent such disclosure is necessary to verify the accuracy of the financial reports furnished by Imugene or the amount of payments due by Imugene under this Agreement. Any amounts shown to be owed but unpaid, or overpaid and in need of reimbursement, shall be paid or refunded (as the case may be) within [***] after the accountant's report, plus interest (as set forth in Section 9.8) from the original due date (unless challenged in good faith by Imugene, in which case any undisputed portion shall be paid in accordance with the foregoing timetable, any dispute with respect to such challenge shall be resolved in accordance with Section 15.2, and any remaining disputed portion shall be paid within [***] after resolution of the dispute). Precision shall bear the full cost of such audit unless such audit reveals an underpayment by Imugene during the applicable audit period, which underpayment was more than [***] of the amount set forth in such report, in which case Imugene shall bear the full cost of such audit.

9.8**Late Payments**. If any payment properly due under this Agreement and not subject to a good faith dispute is not paid when due in accordance with the applicable provisions of this Agreement, the payment shall accrue interest from the date due at [***]. The payment of such interest shall not limit Precision from exercising any other rights it may have as a consequence of the lateness of any payment.

9.9**Taxes**.

- 9.9.1 **Taxes on Income.** Each Party shall be solely responsible for the payment of all taxes imposed on its share of income arising directly or indirectly from the collaborative efforts of the Parties under this Agreement.
- 9.9.2 **Tax Cooperation.** The Parties agree to cooperate with one another and use reasonable efforts to avoid or reduce tax withholding or similar obligations in respect of any

payments made by Imugene to Precision under this Agreement. Without limiting the generality of the foregoing, Precision shall provide Imugene any tax forms and other information that may be reasonably necessary in order for Imugene to not withhold tax or to withhold tax at a reduced rate under an applicable bilateral income tax treaty. Each Party shall provide the other with reasonable assistance to enable the recovery, as permitted by Applicable Law, of withholding taxes, value added taxes, or similar obligations resulting from payments made under this Agreement, such recovery to be for the benefit of the Party bearing such withholding tax or value added tax.

- 9.9.3 **Payment of Tax.** To the extent Imugene is required by Applicable Law to deduct and withhold taxes on any payment to Precision, Imugene shall pay the amounts of such taxes to the proper Governmental Authority in a timely manner and promptly transmit to Precision an official tax certificate or other evidence of such withholding sufficient to enable Precision to claim such payment of taxes.
- 9.9.4 **Treatment of Certain Withholding Taxes.** If Imugene is required to deduct and withhold taxes on any payment to Precision and such withholding obligation arises as a result of any action by Imugene that has the effect of modifying the tax treatment of the Parties hereto (including any assignment or sublicense, or any failure on the part of the paying Party to comply with Applicable Law or filing or record retention requirements) (a "*Withholding Tax Action*"), then the sum payable by Imugene (in respect of which such deduction or withholding is required to be made) shall be increased to the extent necessary to ensure that Precision actually receives, as appropriate, a sum equal to the sum that it would have received had no such Withholding Tax Action occurred.
- 9.9.5 **Blocked Currency.** In each country where the local currency is blocked and cannot be removed from the country, royalties accrued on Net Sales in that country shall be paid in the equivalent amount in Dollars.
- 9.9.6 **Manner and Place of Payment.** All payments owed under this Agreement shall be made by wire transfer in immediately available funds to a bank and account designated in writing by Precision, unless otherwise specified in writing by Precision.

ARTICLE 10

INTELLECTUAL PROPERTY

$10.1 {\bf Ownership\ of\ Intellectual\ Property}.$

10.1.1**Background IP**. As between the Parties, and subject to the licenses granted under this Agreement (a) Imugene shall solely own (or retain ownership of) all rights, title and interests in and to the Imugene Background IP, and (b) Precision shall solely own (or retain ownership of) all rights, title and interests in and to the Precision Background IP. If any Third Party becomes an Acquirer of a Party after the Effective Date pursuant to a Change of Control of such Party, any Patents and Know-How Controlled by the Acquirer before the relevant Change of Control transaction or thereafter during the Term will not be considered part of the Precision Background IP (where Precision is the acquired Party) or Imugene Background IP (where Imugene

is the acquired Party); *provided*, however, that any Patents or Know-How that would otherwise constitute Precision Background IP or Imugene Background IP, as applicable, and are discovered or created by or on behalf of the Acquirer after the relevant Change of Control transaction in its performance under this Agreement, will be considered part of the Precision Background IP or Imugene Background IP, accordingly.

- 10.1.2**Inventions**. Ownership of Inventions arising under this Agreement shall be as follows:
- (a) Imugene shall solely own (or retain ownership of) all Inventions discovered, created, acquired, conceived or reduced to practice, solely by or on behalf of Imugene or any of its Affiliates in the course of performing activities under this Agreement, except to the extent constituting Precision Sole IP ("*Imugene Sole IP*").
- (b) Precision shall solely own (or retain ownership of) (i) all Inventions discovered, created, acquired, conceived or reduced to practice, solely by or on behalf of Precision or any of its Affiliates in the course of performing activities under this Agreement, and (ii) all Inventions that relate to [***], whether discovered, created, conceived or reduced to practice by or on behalf of Imugene or Precision or any of their respective Affiliates in the course of performing activities under this Agreement ("**Precision Sole IP**"). Imugene agrees to assign and hereby assigns to Precision all of its and its Affiliates' right, title and interests in and to the Precision Sole IP and agrees to execute such documents and perform such other acts as Precision may reasonably request to obtain, perfect and enforce the Precision Sole IP and the assignment thereof.
- (c) Except to the extent constituting Precision Sole IP, any Invention discovered, created, conceived, reduced to practice or acquired, jointly by or on behalf of the Parties in the course of performing activities under this Agreement ("*Joint IP*"), will be jointly owned by the Parties.
- 10.1.3**Inventorship**. Inventorship as between the Parties will be determined in accordance with U.S. patent laws. All such determinations shall be documented to ensure that the Patent claims in any divisional or continuation patent applications reflect appropriate inventorship.
- 10.1.4**Rights of Joint Owners**. Subject to the licenses granted hereunder and the payment obligations under Article 9, each Party shall have full rights to exploit and license Joint IP (and any Patents therein), without any obligation or requirement of an accounting to the other Party.
- 10.1.5**Independent Development**. Subject to the licenses granted hereunder, nothing in this Agreement shall be construed as limiting either Imugene's or Precision's right to Develop and in-license technology related to the Imugene Background IP (in the case of Imugene) or Precision Background IP (in the case of Precision) outside the scope of this Agreement in its ordinary course of business.
- 10.1.6**Assignment Obligation**. Each Party shall cause all of its Affiliates, directors, officers, employees, agents, independent contractors, Sublicensees, consultants, and others who perform activities for such Party under this Agreement (each, a "*Representative*") to

be under an appropriate obligation of confidentiality and non-use consistent with the provisions of this Agreement and an obligation to assign (or, if such Party is unable to cause such person or entity to agree to such assignment obligation despite such Party using reasonable efforts to negotiate such assignment obligation, provide a license, preferably exclusive, under) to such Party their rights in and to any Inventions and all intellectual property rights therein such that the Party is able to comply with its obligations under this Agreement as if such Invention had been discovered, created, acquired, conceived or reduced to practice by such Party, except where Applicable Law requires otherwise and except in the case of governmental, not-for-profit and public institutions that have standard policies against such an assignment (in which case a Party shall obtain a suitable license, preferably exclusive, or right to obtain such a license). Each Party shall use reasonable efforts to promptly disclose to the other Party in writing all Inventions arising under this Agreement that are owned by the other Party, including any invention disclosures, or other similar documents, submitted to it by its Representatives describing such Inventions, and all information relating to such Inventions to the extent necessary or useful for the preparation, filing and maintenance of any Patent with respect to such Invention.

10.2Patent Prosecution and Maintenance.

10.2.1**Rights to Prosecute and Maintain Patents**. As between the Parties:

- (a) Imagene has the sole right, but not the obligation, to Prosecute and Maintain any Patents constituting or claiming any Imagene Background IP or Imagene Sole IP, at Imagene's sole cost and expense.
- (b) Precision (or Precision's designee, as applicable) has the first right, but not the obligation, to Prosecute and Maintain any Patents constituting or claiming any Precision Background IP or Precision Sole IP, at Precision's (or its designee's, as applicable) sole cost and expense. Precision will give Imugene the opportunity to review (i) the text of any Precision Product-Specific Claim and (ii) responses to office actions related thereto, in each case, before filing of the relevant application or responding to such office action. Precision will reasonably consider any input or feedback from Imugene with respect to the foregoing, provided, that Precision shall have the final authority with respect to any such decisions. In the event that Precision (or Precision's designee, as applicable) elects not to conduct a Patent Defense Matter with respect to a Precision may, in Precision's sole discretion, elect to permit Imugene to conduct such Patent Defense Matter, at Imugene's sole cost and expense. In the event that Precision elects in writing to permit Imugene to conduct a Patent Defense Matter with respect to any Precision Patent, Imugene shall keep Precision reasonably informed of the status of such Patent Defense Matter and shall consider in good faith Precision's comments thereon. Imugene shall provide Precision with drafts of all material papers and statements to be filed in connection with such Patent Defense Matter in sufficient time to allow Precision to review, consider and substantively comment thereon, and shall in good faith consider all reasonable comments thereto by Precision before filing such papers or statements. Precision may, at its own expense, join as a party to such Patent Defense Matter and be represented in any such action by counsel of its own choice.

(c) Imugene has the first right, but not the obligation, to Prosecute and Maintain any Patents constituting or claiming any Joint IP, at Imugene's sole cost and expense,

and Precision shall have the secondary right, at Precision's sole cost and expense, to Prosecute and Maintain any Patents constituting or claiming any Joint IP, subject to and in accordance with Section 10.2.2.

- (d) Imugene acknowledges and agrees that Precision has no rights or responsibility for preparing, filing, Prosecuting or Maintaining the Cellectis Patents. For clarity, Imugene shall have no rights with respect to preparing, filing Prosecuting or Maintaining the Cellectis Patents.
- 10.2.2**Prosecution and Maintenance Procedures for Joint IP**. The Party handling the Prosecution and Maintenance of a Patent claiming or constituting Joint IP under Section 10.2.1(c) (the "*Prosecuting Party*") shall keep the other Party reasonably informed of the status of the applicable Patent and shall promptly provide the other Party with all material correspondence received from any patent authority in connection therewith. In addition, the Prosecuting Party shall promptly provide the other Party with drafts of all proposed material filings and correspondence to any patent authority with respect to the applicable Patent for the other Party's review and comment prior to the submission of such proposed filings and correspondences, and the Prosecuting Party shall consider the other Party's reasonable comments in good faith. The Prosecuting Party shall notify the other Party of its intention to suspend or cease any Prosecution and Maintenance of any such Patent. The Prosecuting Party shall provide such notice at least [***] prior to any filing or payment due date, or any other due date that requires action, in connection with such Patent. In such event, the Prosecuting Party shall permit the other Party, at the other Party's discretion and at its sole expense, to continue Prosecution and Maintenance of such Patent.

10.2.3 Separation of Patent Claims.

- (a) If a Party determines that an application for a Patent filed, or sought to be filed, by the other Party claims [***], then the Parties agree that, to the extent practicable, such application shall be divided into two (2) or more Patent applications, so that each application shall contain claims that cover only [***].
- (b) If the division contemplated in Section 10.2.3(a) is not practicable, or a single claim covers [***], then such Patent application shall be subject to the provisions of this Agreement relating to [***].
- (c) Similarly, an attempt shall be made to divide Patent applications into those that claim Inventions [***].
- 10.2.4**Cooperation of the Parties**. Each Party shall cooperate fully with the other Party in the Prosecution and Maintenance of Patents under this Section 10.2 at [***] cost (except as expressly set forth otherwise in this Article 10), including by: (a) executing all papers and instruments, or requiring its Representatives, to execute such papers and instruments, to enable the other Party to apply for and to Prosecute and Maintain such Patents in any country as permitted by this Section 10.2; and (b) promptly informing the other Party of any matters coming to such Party's attention that may affect the Prosecution and Maintenance of any such Patents. Each Party will use reasonable efforts via good faith consultation with the other to avoid creating potential issues in Prosecution and Maintenance of Patents under this Section 10.2.

10.2.5**Patent Working Group**. Each Party shall designate to the other Party in writing a patent Prosecution and Maintenance representative to liaise with the other Party's patent Prosecution and Maintenance representative with respect to the Prosecution and Maintenance of Patents under this Section 10.2; such representatives will meet no less frequently than quarterly during the Term, by means of teleconference, Internet conference, videoconference, or other similar communication method, to discuss matters relevant to the Prosecution and Maintenance of Patents under this Section 10.2, including timing of planned filings and other upcoming Prosecution and Maintenance actions. Each Party may update its patent Prosecution and Maintenance representative at any time upon written notice to the other Party.

10.3Infringement or Misappropriation by Third Parties.

10.3.1**Notice**. Each Party shall notify the other within [***] of becoming aware of any alleged or threatened infringement by a Third Party of any of the Precision Patents or Joint Patents, in each case in the Licensed Field in the Territory, and any related declaratory judgment, opposition, or similar action alleging the invalidity, unenforceability or non-infringement of any of the Precision Patents or Joint Patents (collectively "*Infringement*").

10.3.2Joint IP and Precision Product Patents.

(a) As between the Parties, Imugene has the first right, but not the obligation, to bring and control any legal action, at [***] cost and expense, in connection with (i) any Infringement of any [***] or (ii) any Infringement of any Joint IP (other than any [***]) that is competitive with an Existing Product. Imugene shall keep Precision reasonably informed of the status of such enforcement efforts for such Joint IP or [***] and shall consider in good faith Precision's comments thereon. Imugene shall provide Precision with drafts of all material papers and statements to be filed with the court in sufficient time to allow Precision to review, consider and substantively comment thereon, and shall in good faith consider all reasonable comments thereto by Precision before filing such papers or statements. Precision may, at [***] expense, join as a party to such claim, suit, or proceeding and be represented in any such action by counsel of its own choice. If Imugene does not bring such legal action within [***] after the notice provided pursuant to Section 10.3.1 (or within such shorter period prior to the next deadline for any action that must be taken in order to bring such legal action), Precision may bring and control any legal action in connection with such Infringement, at [***] cost and expense as it reasonably determines appropriate so long as Imugene does not reasonably object to such action.

(b) As between the Parties, Precision shall have the first right, but not the obligation, to bring and control any legal action, at [***] cost and expense, in connection with any Infringement of any Joint IP (other than any Infringement described in Section 10.3.2(a)). Precision shall keep Imugene reasonably informed of the status of such enforcement efforts for such Joint IP and shall consider in good faith Imugene's comments thereon. Precision shall provide Imugene with drafts of all material papers and statements to be filed with the court in sufficient time to allow Imugene to review, consider and substantively comment thereon, and shall in good faith consider all reasonable comments thereto by Imugene before filing such papers or statements. Imugene may, at its own expense, join as a party to such claim, suit, or proceeding and be represented in any such action by counsel of its own choice. If Precision does not bring such legal action within [***] after the notice provided pursuant to Section 10.3.1, Imugene may bring and

control any legal action in connection with such Infringement, at [***] cost and expense as it reasonably determines appropriate.

10.3.3**Precision Background IP and Precision Sole IP**. Except as set forth in Section 10.3.2(a), as between the Parties, Precision has the sole right to initiate any proceedings or take other appropriate actions against an infringement of any Precision Background IP or Precision Sole IP and to defend against any challenge of any Precision Background IP or Precision Sole IP that are brought by a Third Party in connection with such infringement. Imugene acknowledges and agrees that (a) Precision has no rights or responsibility for enforcing the Cellectis Patents, and therefore all references to Precision Background IP in this Section 10.3 shall be deemed to exclude the Cellectis Patents for all purposes, (b) prior to initiating enforcement actions against a Third Party with respect to certain Precision Patents which are subject to the non-exclusive license granted by Precision to Cellectis S.A. pursuant to the Cellectis Agreement, Precision is required by the Cellectis Agreement to confirm that Cellectis S.A. has not granted a license to such Third Party under such Precision Patents, and Imugene will cooperate with Precision in taking such actions as required by the Cellectis Agreement, and (c) Duke retains discretion as to whether to become a party plaintiff and has certain rights with respect to enforcement of Patents contained within the Duke IP in the event Precision does not enforce such Patents.

10.3.4**Imugene Background IP and Imugene Sole IP**. Imugene has the sole right to initiate any proceedings or take other appropriate actions against an infringement of any Imugene Background IP or Imugene Sole IP and to defend against any challenge of any Imugene Background IP or Imugene Sole IP that are brought by a Third Party in connection with such infringement.

10.3.5**Allocation of Recoveries**. Any recoveries resulting from enforcement action relating to a claim of Infringement shall be [***].

10.3.6**Cooperation**. At the request and expense of the Party bringing an action under this Section 10.3, the other Party shall provide reasonable assistance in connection therewith, including by executing reasonably appropriate documents, cooperating in discovery and joining as a party to the action if required by Applicable Law to pursue such action. In connection with any such enforcement action, the Party bringing the action shall not enter into any settlement admitting the invalidity or non-infringement of, or otherwise impairing the other Party's rights in the applicable Patents without the prior written consent of the other Party.

10.4**Defense and Settlement of Third Party Claims**. Each Party shall promptly notify the other in writing of: (a) any allegation by a Third Party that the activity of either of the Parties pursuant to this Agreement infringes or may infringe the intellectual property rights of such Third Party; or (b) any declaratory judgment action that is brought naming either Party as a defendant and alleging invalidity of any of the Precision Patents, or Joint Patents. Precision has the sole right to control any defense of any such claim described in (a) involving alleged infringement of Third Party rights by Precision's activities at [***] expense and by counsel of its own choice, and Imugene may, at [***] expense, be represented in any such action by counsel of its own choice. Imugene has the sole right to control any defense of any such claim described in (a) involving alleged infringement of Third Party rights by Imugene's activities at [***] expense and by counsel

of its own choice, and Precision may, at [***] expense, be represented in any such action by counsel of its own choice. Neither Party may settle any patent infringement litigation under this Section 10.4 in a manner that admits the invalidity or unenforceability of the other Party's Patents or a Joint Patent or imposes on the other Party restrictions or obligations or other liabilities, without the written consent of such other Party, which consent shall not be unreasonably withheld, conditioned, or delayed. Nothing in this Section 10.4 will limit any indemnification rights or obligations of a Party under Article 12.

10.5**Patent Extension**. The Parties shall cooperate in determining whether a Joint Patent claiming or covering a Licensed Product should be extended, and thereafter the Parties shall cooperate in obtaining patent term restorations, supplemental protection certificates or their equivalents, and other forms of patent term extensions for a given Licensed Product with respect to any applicable Joint Patent in any country or region where applicable. Precision shall have final decision-making authority with respect to decisions regarding patent term extensions for Precision Patents. Imagene shall have final decision-making authority with respect to decisions regarding patent term extensions for Imagene Patents.

10.6**CREATE** Act. It is the Parties' intention that this Agreement is a "joint research agreement" as that phrase is defined in 35 U.S.C. § 102(c) as amended by the Cooperative Research and Technology Enhancement (CREATE) Act, including the provisions of 35 U.S.C. § 102(b)(2)(c). The Parties agree to cooperate and to take reasonable actions to maximize the protections available for the Licensed ARCUS Nuclease and Licensed Products under such safe harbor provisions.

10.7Licenses to Third Party Intellectual Property Rights. If (a) a Party becomes aware of any Patent of a Third Party that (i) claims or embodies the Licensed ARCUS Nuclease or ARCUS Technology as a composition of matter, or a method of making or using the Licensed ARCUS Nuclease or ARCUS Technology and (ii) is not the subject of an agreement with a Party as of or prior to the Effective Date; then (b) such Party shall notify the other in writing, identifying the relevant Patent. Precision shall have the first right (but not the obligation) to negotiate and obtain a license from such Third Party under such Patent described under a notice described in the foregoing (a) for a period of [***] following the date of such notice.

10.8 Licensed Product Trademarks. Imugene shall have the right to select, and will be free, in its sole discretion, to use and to register in any trademark office in the Territory any trademark for use with a Licensed Product (the "Licensed Product Trademarks"); provided that (a) Imugene shall not use, file applications for, or register any trademarks owned by Precision (or its Affiliates), whether stand-alone or in combination with a design element, for the benefit of branding (including co-branding) without the prior written consent of Precision; and (b) Imugene's selection of any Licensed Product Trademark for an Existing Product shall be subject to Precision's prior written approval in Precision's sole discretion. As between the Parties, Imugene shall own all right, title and interest in and to any such Licensed Product Trademarks adopted by Imugene for use with Licensed Product, and is responsible for the registration, filing, maintenance and enforcement thereof.

ARTICLE 11

REPRESENTATIONS, WARRANTIES AND COVENANTS

- 11.1**Mutual Representations and Warranties**. Each of Imugene and Precision represent and warrant that, as of the Effective Date:
- 11.1.1it is duly organized and validly existing under in the Applicable Laws of the jurisdiction of its incorporation or formation, as applicable, has full corporate, limited liability company or other power and authority, as applicable, to enter into this Agreement and to carry out the provisions hereof, and has sufficient facilities, experienced personnel or other capabilities (including via Affiliates and/or Third Parties) to enable it to perform its obligations under this Agreement;
- 11.1.2it is duly authorized to execute and deliver this Agreement and to perform its obligations hereunder, and the individual executing this Agreement on its behalf has been duly authorized to do so by all requisite corporate, limited liability company or other action, as applicable; and
- 11.1.3this Agreement is legally binding upon it and enforceable in accordance with its terms (except as the enforceability thereof may be limited by bankruptcy, bank moratorium or similar laws affecting creditors' rights generally and laws restricting the availability of equitable remedies and may be subject to general principles of equity whether or not such enforceability is considered in a proceeding at law or in equity) and the execution, delivery and performance of this Agreement by it have been duly authorized by all necessary corporate action and do not and will not: (a) conflict with, or constitute a default or result in a breach under, any agreement, instrument or understanding, oral or written, to which it is a party or by which it may be bound, or violate any Applicable Law; or (b) require any consent or approval of its stockholders or similar.
- 11.2**Precision Representations and Warranties**. Precision represents and warrants to Imugene that, as of the Effective Date:
- 11.2.1**No Grants that Conflict with this Agreement**. Precision and its Affiliates have not granted any rights (or other encumbrances) to any Third Party under Precision Technology that conflict with the rights granted to Imugene hereunder.

11.2.2 Existing Patents.

- (a) All Precision Patents Covering any Existing Product or the Licensed ARCUS Nuclease that exist as of the Effective Date, other than the Cellectis Patents, that are issued or subject to a pending application for issuance are listed on Exhibit 11.2.2 (the "Existing Patents").
- (b) The Existing Patents and the Cellectis Patents represent all Patents Controlled by Precision that Cover an Existing Product, the Licensed ARCUS Nuclease, or the Exploitation of any of the foregoing in the Licensed Field.
 - (c) All Existing Patents are: (i) to the extent issued (unless otherwise

45

indicated on Exhibit 11.2.2), subsisting and, to Precision's Knowledge, not invalid or unenforceable, in whole or in part, or to Precision's Knowledge, confer a valid right to claim priority thereto; (ii) solely and exclusively owned or exclusively licensed to Precision, free of any encumbrance, lien or claim of ownership by any Third Party; (iii) in respect of Existing Patents owned by Precision, to the extent subject to a pending application for issuance, being prosecuted in good faith in the respective patent offices in which such applications have been filed in accordance with Applicable Law and, to Precision's Knowledge, all material references, documents and information have been presented to the relevant patent office in respect of such Existing Patents to the extent required by such patent office; and (iv) in respect of Existing Patents owned by Precision, filed and maintained in accordance with applicable Patent office rules, and all applicable fees applicable thereto have been paid on or before any final due date for payment.

11.2.3[***]

11.2.4[***]

- 11.2.5**Other Material Claims and Actions**. There are no claims, actions, or proceedings pending or, to Precision's Knowledge, threatened by any Third Party against Precision or its properties, assets or business, which if adversely decided, would, individually or in the aggregate, have a material adverse effect on, or prevent Precision's ability to grant the licenses or rights granted to Imugene under this Agreement.
- 11.2.6**No Government Funding**. Except with respect to the Duke IP, the Inventions claimed or covered by the Precision Patents: (a) were not conceived, discovered, developed or otherwise made in connection with any research activities funded, in whole or in part, by the federal government of the United States of America or any agency thereof; (b) are not a "subject invention" as that term is described in 35 U.S.C. Section 201(e) and (c) are not otherwise subject to the provisions of the Patent and Trademark Law Amendments Act of 1980, as amended, codified at 35 U.S.C. §§ 200-212, as amended, as well as any regulations promulgated pursuant thereto, including in 37 C.F.R. Part 401 (the "*Bayh-Dole Act*").
- 11.3**Imugene Representations and Warranties**. Imugene represents and warrants to Precision that, as of the Effective Date, Imugene and its Affiliates have not granted any rights (or other encumbrances) to any Third Party under Imugene Arising IP or Imugene Background IP that conflict with the rights granted to Precision hereunder.

11.4Mutual Covenants.

11.4.1**Debarment**. Each Party represents and warrants to the other Party that such Party has not, and its Representatives have not been: (a) debarred or disqualified under the U.S. Federal Food, Drug and Cosmetic Act; (b) listed by any government or regulatory agencies as ineligible to participate in any government healthcare programs or government procurement or non-procurement programs (as that term is defined in 42 U.S.C. § 1320a-7b(f)), or excluded, debarred, suspended or otherwise made ineligible to participate in any such program; or (c) convicted of a criminal offense related to the provision of healthcare items or services, or is subject to any such pending action. Each Party will not during the Term knowingly, employ or use, directly or indirectly, including through Affiliates the services of any such person. In the event that either Party becomes aware of the debarment or disqualification or threatened debarment or

disqualification of any person providing services to such Party, directly or indirectly, including through Affiliates or Sublicensees, which directly or indirectly relate to activities contemplated by this Agreement, such Party shall promptly notify the other Party in writing and such Party shall cease employing, contracting with, or retaining any such person to perform any such services.

- 11.4.2**Protection of Information**. Each Party agrees that during the Term of this Agreement, and without limiting its obligations hereunder, such Party shall implement technical and organizational measures to protect all information under the Agreement that are appropriate and that provide no less protection than both (i) good industry practice (i.e., in accordance with ISO 27001 and/or similar industry standards) and (ii) such Party's measures to protect its own information of a similar nature or importance.
- 11.5**Precision Covenant**. Precision covenants and agrees that during the Term: (1) it shall satisfy all of its obligations under (including making all payments), and take all steps to maintain in full force and effect, the Existing In-License Agreements; (2) it will not assign (except an assignment to a party to which this Agreement has been assigned as permitted under Section 16.7), amend, restate, amend and restate, terminate in whole or in part, or otherwise modify an Existing In-License Agreement in any manner that limits Imugene's exercise of the rights granted in this Agreement without the prior written consent of Imugene; and (3) it will provide Imugene with prompt notice of any claim of a breach under an Existing In-License Agreement made by either Precision or Duke or Cellectis S.A., as applicable.

11.6Compliance.

- 11.6.1**Compliance with Applicable Laws**. Each Party covenants to the other that in the performance of its obligations under this Agreement, such Party shall comply with, and shall cause its Affiliates and its and its Affiliates' employees and contractors to comply with, all Applicable Laws. No Party shall, or shall be required to, undertake any activity under or in connection with this Agreement which violates, or which it believes, in good faith, may violate, any Applicable Laws.
- 11.6.2**Compliance with Internal Compliance Codes**. All Internal Compliance Codes shall apply only to the Party to which they relate. The Parties agree to cooperate with each other to help ensure that each Party is able to comply with the substance of its respective Internal Compliance Codes and, to the extent practicable, each Party shall operate in a manner consistent with its Internal Compliance Codes applicable to its performance under this Agreement. "*Internal Compliance Codes*," as used in this Section 11.6.2, means a Party's internal policies and procedures intended to ensure that a Party complies with Applicable Laws and such Party's internal ethical, medical and similar standards.
- 11.6.3 **Compliance with Anti-Corruption Laws**. In connection with this Agreement, the Parties shall comply with all applicable local, national, and international laws, regulations, and industry codes dealing with government procurement, conflicts of interest, corruption or bribery, including, if applicable, the U.S. Foreign Corrupt Practices Act of 1977, as amended, and any laws enacted to implement the Organization of Economic Cooperation and Development Convention on Combating Bribery of Foreign Officials in International Business Transactions.

11.6.4**Prohibited Conduct.** Without limiting the other obligations of the Parties set forth in this Section 11.6, each Party covenants to the other that, as of the Effective Date and in the performance of its obligations under this Agreement through the expiration and termination of this Agreement, such Party and, to its knowledge, its Affiliates and its and its Affiliates' Representatives, in connection with the performance of their respective obligations under this Agreement, have not made, offered, given, promised to give, or authorized, and will not make, offer, give, promise to give, or authorize, any bribe, kickback, payment or transfer of anything of value, directly or indirectly through Third Parties, to any Government Official for the purpose of: (a) improperly influencing any act or decision of the Person or Government Official; (b) inducing the Person or Government Official to do or omit to do an act in violation of a lawful or otherwise required duty; (c) securing any improper advantage; or (d) inducing the Person or Government Official to improperly influence the act or decision of any organization, including any government or government instrumentality, to assist any Party in obtaining or retaining business. For the purpose of this Section "Government Official" means: (x) any officer, employee (including physicians, hospital administrators, or other healthcare professionals), agent, representative, department, agency, de facto official, representative, corporate entity, instrumentality or subdivision of any government, military or international organization, including any ministry or department of health or any state-owned or affiliated company or hospital; (y) any candidate for political office, any political party or any official of a political party, in each case for the purpose of obtaining or retaining business for or with, or directing business to, any Person, including either Party; or (z) any Person acting in an official capacity on behalf of any of the foregoing.

11.7Disclaimer, EXCEPT AS OTHERWISE EXPRESSLY SET FORTH IN THIS Article 11, NEITHER PARTY MAKES ANY REPRESENTATIONS OR EXTENDS ANY WARRANTIES OF ANY KIND, EITHER EXPRESS OR IMPLIED, INCLUDING WARRANTIES OF MERCHANTABILITY, QUALITY, FITNESS FOR A PARTICULAR PURPOSE, NONINFRINGEMENT, OR VALIDITY OF PATENT CLAIMS OR THE AVAILABILITY OF ANY LICENSES WITH RESPECT TO INTELLECTUAL PROPERTY RIGHTS OF THIRD PARTIES. NOTHING IN THIS AGREEMENT SHALL BE CONSTRUED AS A REPRESENTATION MADE OR WARRANTY GIVEN BY EITHER PARTY THAT EITHER PARTY WILL BE SUCCESSFUL IN OBTAINING ANY PATENTS OR THAT ANY PATENTS WILL ISSUE BASED ON A PENDING APPLICATION. WITHOUT LIMITING THE RESPECTIVE RIGHTS AND OBLIGATIONS OF THE PARTIES EXPRESSLY SET FORTH HEREIN, EACH PARTY SPECIFICALLY DISCLAIMS ANY GUARANTEE THAT ANY LICENSED PRODUCTS, **INCLUDING** THE RESEARCH, MANUFACTURE, **DEVELOPMENT** OR COMMERCIALIZATION THEREOF, WILL BE SUCCESSFUL, IN WHOLE OR IN PART.

ARTICLE 12

INDEMNIFICATION

12.1Indemnity.

12.1.1**By Precision**. Precision shall defend, indemnify and hold harmless Imugene and its Affiliates, and their respective Representatives (each, a "*Imugene Indemnitee*")

from and against any and all costs, fees, expenses, losses, liabilities, and damages, including reasonable legal expenses and attorneys' fees (collectively, "*Losses*") to which any Imugene Indemnitee may become subject as a result of any claim, demand, action or other proceeding by any Third Party (a "*Claim*") to the extent such Losses arise out of: (a) the gross negligence or willful misconduct of Precision or its Affiliates in connection with its activities under this Agreement; or (b) the breach of this Agreement or the representations, warranties, and covenants made hereunder by Precision; except, in each case, to the extent such Losses result from matters subject to clause (a), (b) or (c) of Section 12.1.2.

12.1.2**By Imugene**. Imugene shall defend, indemnify and hold harmless Precision, its Affiliates, Duke, and its and their respective Representatives (each, a "*Precision Indemnitee*") from and against any and all Losses to which any Precision Indemnitee may become subject as a result of any Claim to the extent such Losses arise out of: (a) the gross negligence or willful misconduct of Imugene, its Affiliates, or its or their respective Sublicensees in connection with its activities under this Agreement; (b) the breach of this Agreement or the representations, warranties and covenants made hereunder by Imugene; or (c) [***]; except, in each case, to the extent such Losses result from matters subject to clause (a) or (b) of Section 12.1.1.

12.1.3**Procedure.** A Party that intends to claim indemnification under this Article 12 (the "*Indemnitee*") shall promptly notify the Indemnitor (the "Indemnitor") in writing of any Claim in respect of which the Indemnitee intends to claim such indemnification. The failure to deliver written notice to the Indemnitor within a reasonable time after the commencement of any action with respect to a Claim shall only relieve the Indemnitor of its indemnification obligations under this Article 12 if and to the extent the Indemnitor is actually and materially prejudiced thereby. The Indemnitor has sole control of the defense or settlement thereof. The Indemnitee shall cooperate fully with the Indemnitor and its legal representatives in the investigation of any action with respect to a Claim covered by this indemnification. The Indemnitee may participate at its expense in the Indemnitor's defense of and settlement negotiations for any Claim with counsel of the Indemnitee's own selection. The Indemnitor shall not settle any Claim in a manner that admits liability of Indemnitee or requires Indemnitee to perform any material obligations (other than payment of money which will be fully satisfied by Indemnitor) without the prior written consent of the Indemnitee, not to be unreasonably withheld, conditioned or delayed. So long as the Indemnitor is actively engaged in activities relating to defending or settling the Claim in good faith, the Indemnitee shall not settle or compromise any such Claim without the prior written consent of the Indemnitor. If the Indemnitor does assume activities in furtherance of the defense and settlement of a Claim as provided above within [***] after written notice from Indemnitee stating intent of Indemnity to undertake such activities if Indemnitor does not: (a) the Indemnitee may defend against, consent to the entry of any judgment, or enter into any settlement with respect to such Claim in any manner the Indemnitee may deem reasonably appropriate (and the Indemnitee need not consult with, or obtain any consent from, the Indemnitor in connection therewith); and (b) the Indemnitor shall remain responsible to indemnify the Indemnitee as provided in this Article 12.

12.2**Insurance**. Each Party, at its own expense, shall maintain product liability and other appropriate insurance (or self-insure) in an amount consistent with sound business practice and reasonable in light of its obligations under this Agreement during the Term and for a period of [***] thereafter or for otherwise longer as may be required by Applicable Law; but in any event,

and without limiting the foregoing, no later than Initiation of the first Clinical Trial for a Licensed Product, Imugene shall procure and maintain product liability insurance in an amount not less than [***] per occurrence and in the annual aggregate. Each Party shall provide a certificate of insurance (or evidence of self-insurance) evidencing such coverage to the other Party upon request. The Parties agree that such insurance shall not be construed to create a limit of either Party's liability with respect to its indemnification obligations under this Article 12 or other obligations under this Agreement.

ARTICLE 13

CONFIDENTIALITY

13.1 Confidential Proprietary Information.

13.1.1Confidential Proprietary Information. In connection with this Agreement, each Party may disclose technical, business or other confidential information in connection with this Agreement, whether prior to, on, or after the Effective Date, including (a) any unpublished Patents, and (b) any information regarding the scientific, regulatory or business affairs or other activities of either Party; in each case ((a) and (b)) that is marked or identified at the time of disclosure as confidential or proprietary or is of such a nature that would be understood by a reasonable person to be confidential or proprietary (such confidential information, "Confidential Proprietary Information"). Without limiting the foregoing, the terms of this Agreement and all Joint IP are the Confidential Proprietary Information of both Parties and shall be treated confidentially by each of the Parties, subject to the exceptions set forth in this Section 13.1. [***] Information exchanged by the Parties pursuant to the Confidentiality Agreement shall be treated as Confidential Proprietary Information under this Agreement and governed by the terms of this Agreement.

13.1.2**Restrictions**. A Party (the "*Receiving Party*") that receives Confidential Proprietary Information from the other Party (the "*Disclosing Party*") shall keep all the Disclosing Party's Confidential Proprietary Information in confidence with the same degree of care with which the Receiving Party holds its own confidential information (but in no event less than a commercially reasonable degree of care), and will not disclose such Confidential Proprietary Information to any Person except as permitted under Section 13.1.4. A Receiving Party shall not use the Disclosing Party's Confidential Proprietary Information except in connection with the performance of its obligations and exercise of its rights under this Agreement.

13.1.3Exceptions. The obligations of confidentiality and restriction on use of Confidential Proprietary Information under Section 13.1.2 do not apply to any information that the Receiving Party can prove by competent written evidence: (a) is now, or hereafter becomes, through no act or failure to act on the part of the Receiving Party, generally known or available to the public; (b) is known by the Receiving Party at the time of receiving such information, other than by previous disclosure of the Disclosing Party or its Affiliates or Representatives; (c) is hereafter furnished to the Receiving Party without restriction by a Third Party who has no obligation of confidentiality or limitations on use with respect thereto, as a matter of right; or (d) is independently discovered or developed by the Receiving Party without the use of or reference to Confidential Proprietary Information belonging to the Disclosing Party. Specific information

shall not be deemed to be within any of the foregoing exclusions merely because it is embraced by more general information falling within those exclusions. Further, any combination of Confidential Proprietary Information shall not be deemed to be generally known, available to the public or known by the Receiving Party merely because individual elements of such Confidential Proprietary Information are subject to such exclusions unless the combination and its principles are subject to such exclusions.

- 13.1.4**Permitted Disclosures**. The Receiving Party may disclose Confidential Proprietary Information belonging to the Disclosing Party as expressly permitted by this Agreement or if and to the extent such disclosure is reasonably necessary in the following instances:
- (a) made by or on behalf of the Receiving Party to a Patent authority as may be reasonably necessary or useful for purposes of Prosecution and Maintenance of Patents as permitted by this Agreement; *provided*, that neither Party shall file a patent application that discloses Imugene Technology (for disclosures by Precision) or Precision Technology (for disclosures by Imugene) without the prior written consent of the other Party (such consent not to be unreasonably withheld, conditioned or delayed);
- (b) made by or on behalf of the Receiving Party to Regulatory Authorities as necessary or reasonably useful in connection with any Regulatory Filings for a product that such Party has a license or right to develop in a given country or jurisdiction;
- (c) made by or on behalf of the Receiving Party as may be necessary ore reasonably useful for prosecuting or defending litigation as permitted by this Agreement;
- (d) made by or on behalf of the Receiving Party for the purpose of complying with a valid order of a court of competent jurisdiction or other Governmental Authority of competent jurisdiction or, if in the opinion of the Receiving Party's legal counsel, such disclosure is otherwise required by Applicable Law;
- (e) made by or on behalf of the Receiving Party where such disclosure is required by a Regulatory Authority (including in filings with the Securities and Exchange Commission or other agency) of certain material developments or material information generated under this Agreement; *provided* that, to the extent permitted, the Party seeking such disclosure first provides the other Party a copy of the proposed disclosure; and *provided*, further, that the receiving Party shall afford to the other Party an opportunity to review and comment, which period shall be no less than [***] (*provided* that if the applicable disclosure is required to be made within fewer than [***], then the receiving Party shall afford to the other Party a reasonable opportunity to review and comment consistent with such disclosure requirement), and the Receiving Party shall accept any reasonable comments so provided;
- (f) made by or on behalf of Precision to Duke solely as and to the extent necessary to fulfill Precision's reporting obligations under the Duke Agreement as of the Effective Date so long as such information is disclosed subject to the confidentiality provisions of the Duke Agreement as of the Effective Date;

(g) made by or on behalf of the Receiving Party in response to a valid request by a U.S., state, foreign, provincial, or local tax authority, in which case either Party may disclose, a copy of this Agreement (including any Exhibits, Appendices, ancillary agreements, and amendments hereto);

(h) made by the Receiving Party to its and its Affiliates' Representatives, subcontractors, and to Sublicensees (in the case of Imugene) or licensees (in the case of Precision), in each case on a need-to-know basis (as reasonably determined by the Receiving Party) in connection with the Exploitation of an Existing Product (in the case of both Parties) or other Licensed Products (in the case of Imugene) in the Territory, in each case under written obligations of confidentiality and non-use substantially consistent with those herein; and

(i) made by the Receiving Party to potential and actual investors, acquirers, licensees and other financial or commercial partners solely for the purpose of evaluating or carrying out an actual or potential investment, acquisition, or collaboration, in each case under written obligations of confidentiality and non-use at least as stringent as those herein; *provided*, however, that with respect to disclosure to actual or bona fide potential investors, such disclosure is under a written obligation of confidentiality that is consistent with market terms, including a shorter period of time during which such information must be held confidential.

[***]

13.1.5**Disclosure of Agreement**. Notwithstanding the foregoing in this Article 13, either Party or its Affiliates may disclose the relevant terms of this Agreement: (a) to the extent required or advisable to comply with the rules and regulations promulgated by the U.S. Securities and Exchange Commission or any equivalent governmental agency in any country in the Territory, provided that such Party shall submit a confidential treatment request in connection with such disclosure and shall submit with such confidential treatment request only such redacted form of this Agreement, which redacted form of this Agreement shall be provided to the other Party for review and comment and which comments shall be considered in good faith by the disclosing Party; (b) upon request from a Governmental Authority (such as a tax authority), *provided* that the disclosing Party uses reasonable efforts to ensure the Governmental Authority maintains such terms as confidential; (c) to applicable licensors, to the extent necessary to comply with the terms of any Third Party license agreement, the rights under which are sublicensed to the other Party under this Agreement; and (d) to the extent necessary to perform obligations or exercise rights under this Agreement, to any sublicensee, collaborator or potential sublicensee or potential collaborator of such Party, *provided* that any sublicensee, collaborator or potential sublicensee or collaborator agree in writing to be bound by obligations of confidentiality and non-use no less protective of the Disclosing Party than those set forth in this Agreement.

13.1.6**Survival**. Each Party's obligations under this Section 13.1 shall apply during the Term and continue for [***].

13.2**Publicity**. Neither Party shall issue any press release or public statement disclosing information relating to this Agreement or the transactions contemplated hereby or the terms hereof without the prior written consent of the other Party, not to be unreasonably withheld, conditioned, or delayed; *provided* however, that (a) neither Party will be prevented from complying with any

duty of disclosure it may have pursuant to Applicable Laws or pursuant to the rules or regulations of any applicable Governmental Authority, national securities exchange or quotation system, subject to the restrictions set forth in Sections 13.1.4 and 13.1.5; and (b) Precision will not be prevented from disclosing publicly the achievement of any Milestone Event and the receipt (and the amount) of any corresponding payment, provided that Imugene shall have at least [***] to review and provide edits and comments to any public disclosure proposed by Precision under this Section 13.2(b) and (ii) Precision shall reasonably incorporate any edits and address any comments provided by Imugene in such proposed public disclosure. If either Party desires to issue a press release or other public statement disclosing information relating to this Agreement or the transactions contemplated hereby or the terms hereof, the issuing Party will provide the other Party with a copy of the proposed press release or public statement. The issuing Party shall specify with each such proposed press release or public statement, taking into account the urgency of the matter being disclosed, a reasonable period of time within which the Receiving Party may provide any comments on such proposed press release or public statement. If the reviewing Party provides any comments, the Parties shall consult with one another on such proposed press release or public statement and work in good faith to prepare a mutually acceptable press release or public statement. Each Party may repeat any information relating to this Agreement that has already been publicly disclosed in accordance with this Section 13.2 or another Transaction Document, provided that such information continues as of such time to be accurate.

13.3 Publication. At least [***] before Imagene or its Affiliate makes any public disclosure (whether by oral presentation, poster, manuscript or abstract) or submits for publication of a proposed publication (such applicable period, the "Review Period") relating to any Clinical Trial data, non-clinical or preclinical data, or any associated results or conclusions specific to any Existing Product or the Licensed ARCUS Nuclease that have not been previously publicly disclosed (collectively, a "Publication"), Imugene shall deliver a complete copy of the applicable proposed Publication to Precision. Imugene will provide Precision with a copy of such proposed Publication at least [***] prior to the earlier of its presentation or intended submission for publication. Imagene agrees that it will not submit or present any Publication until (a) Precision has provided written comments during such Review Period on the material in such Publication, or (b) the applicable Review Period has elapsed without written comments from Precision, in which case Imugene may proceed and the Publication will be considered approved in its entirety. If Imugene receives written comments from Precision on any Publication during the applicable Review Period, then it will consider Precision's comments in good faith and incorporate such comments where appropriate. Notwithstanding any provision to the contrary set forth in this Agreement, Imugene will (i) delete any Confidential Proprietary Information of Precision that Precision identifies for deletion, and (ii) delay such Publication for a period of up to an additional [***] after the end of the applicable Review Period to enable Precision to draft and file one or more patent applications with respect to any subject matter to be made public in such Publication. Imagene will provide Precision a copy of the Publication at the time of the submission or presentation thereof. Imagene agrees to acknowledge the contributions of Precision and the employees of Precision, in each case, in all Publications as scientifically appropriate. Imagene will require its Affiliates and Sublicensees to comply with the obligations of this Section 13.3 as if they were Imugene, and Imugene will be liable for any noncompliance of such Persons.

ARTICLE 14

TERM & TERMINATION

14.1**Term**. This Agreement shall commence on the Effective Date and, unless terminated earlier as provided in this Article 14 or by mutual written agreement of the Parties, shall continue until the expiration of the last Royalty Term (the "*Term*"). Upon expiration (but not termination of this Agreement) of the Royalty Term with respect to any Licensed Product in any country within the Territory, the licenses under Section 8.1.1 and Section 8.1.2 with respect to such Licensed Product in such country will become perpetual, fully paid-up and royalty-free.

14.2Termination.

14.2.1**Termination for Material Breach of Agreement**. Either Party may terminate this Agreement upon written notice to the other Party if such other Party materially breaches its obligations under this Agreement and, after receiving written notice from the non-breaching Party identifying such material breach in reasonable detail, fails to cure such material breach within [***] from the date of such notice [***].

14.2.2Termination by Precision.

- (a) Without limiting Section 14.2.1, Precision may terminate this Agreement upon written notice to Imugene upon the occurrence of any breach in any material respect of any representation, warranty, covenant or agreement of Imugene or any Affiliate of Imugene in this Agreement or any other Transaction Document.
- (b) Without limiting Section 14.2.1, Precision may terminate this Agreement with respect to the Existing Product (but, for clarity, not the Additional Products) upon written notice to Imugene upon the occurrence of (i) Imugene fails to initiate the first dosing of a human subject in the Expansion Cohort with the Existing Product that is the subject of the Expansion Cohort Modification (or any adjustment thereof submitted in accordance with Section 4.1.3(a)) by the later of (A) [***], and (B) [***] after the FDA does not object to the Expansion Cohort Modification (or any adjustment thereof submitted in accordance with Section 4.1.3(a)) during the applicable review period therefor, (ii) [***] Imugene and its Affiliates and Sublicensees have suspended or do not have an active and ongoing Development program with respect to any Existing Product for [***], or (iii) [***] Imugene, Parent, and their respective Affiliates, on an annual basis, fails to collectively expend on the Development of the Existing Product [***].
- 14.2.3**Termination for Insolvency**. Subject to any applicable Australian Ipso Facto Stay and in the event that either Party or Parent (a) makes an assignment for the benefit of creditors, (b) appoints or suffers appointment of a receiver or trustee over any or substantially all of its property, where the receiver or trustee appointment is not discharged within [***] after such filing, (c) proposes a written agreement of composition with its creditors, (d) resolves to enter into, or enters into, a scheme of arrangement or a deed of company arrangement, (e) proposes or is a party to any dissolution or liquidation, (f) appoints or suffers the appointment of an administrator, (g) files a petition under any bankruptcy or insolvency act or has any such petition filed against it that is not discharged within [***] of the filing thereof, or (h) admits in writing its inability generally to meet its obligations as they fall due in the general course or is otherwise

insolvent within the meaning given in the Australian Corporations Act, then Precision (if Imugene or Parent becomes subject to a relevant Insolvency Event) or Imugene (if Precision becomes subject to a relevant Insolvency Event) may terminate this Agreement in its entirety effective immediately upon written notice to the other Party.

14.2.4**Termination for Patent Challenges**. To the extent permitted under Applicable Law, Precision shall have the right to terminate this Agreement upon written notice to Imugene if Imugene or any of its Affiliates or Sublicensees, directly, or indirectly through any Third Party challenges the validity of any Patents Controlled by Precision, including commencing any interference or opposition proceeding with respect to, challenging the patentability, validity or enforceability of, or opposing any extension of or the grant of a Patent Term Adjustment or Extension or supplementary protection certificate with respect to, any Licensed Product in the Territory. Notwithstanding the forgoing, Precision will not have any right to terminate this Agreement pursuant to this Section 14.2.4 on the basis of that act if, within [***] after Imugene's receipt of written notice from Precision, (a) the challenging party permanently withdraws its challenge with respect to any challenge made by a Sublicensee or (b) Imugene terminates the applicable sublicense agreement.

14.3Effects of Termination. Upon any termination of this Agreement, the following provisions will apply, and all Licensed Products will be deemed "*Terminated Products*." Upon any termination of this Agreement with respect to the Existing Product pursuant to Section 14.2.2(b), the following provisions will apply, but only the Existing Product will be deemed a "*Terminated Product*".

14.3.1**Termination of Licenses from Precision**. All licenses for Terminated Products granted by Precision under Article 8 terminate automatically as of the termination effective date and all such rights shall revert to Precision; provided that, if Imugene (or its Affiliates or Sublicensees) has inventory of usable Terminated Product(s) as of the effective date of termination, then Imugene (and its Affiliates and Sublicensees) may continue to sell off such inventory of Terminated Products in the Licensed Field in the Territory (and fulfill customer orders therefor) until the earlier to occur of [***] after the effective date of termination and the date on which Imugene (or its Affiliates or Sublicensees) no longer has such inventory of Terminated Product(s) and shall pay Precision any applicable Royalties and Proceeds due (and Commercial Milestone Payments for Commercial Milestone Events achieved, as applicable) based on such sales. Any permitted sublicense granted by Imugene or its Affiliate to a Sublicensee under the licenses granted to Imugene under this Agreement shall survive the termination of this Agreement upon written request by the applicable Sublicensee and Imugene shall assign such sublicense to Precision such that such sublicense becomes a direct license between Precision and the Sublicensee on the same terms and conditions as those set forth in this Agreement to the extent applicable to the rights granted by Imugene to such Sublicensee, provided that, such sublicense was granted in accordance with the terms of Section 8.4 and in the case where termination of this Agreement was for Imugene's uncured material breach pursuant to Section 14.2, such Sublicensee did not cause such uncured material breach and such Sublicensee is, at the time of such termination, otherwise in compliance with the sublicense granted by Imugene to such Sublicensee and the applicable terms and conditions of this Agreement.

14.3.2**Destruction of Confidential Proprietary Information**. Subject to the potential transfer of any data and information covered below in Section 14.4, each Receiving Party shall destroy (at the Disclosing Party's written request) all such Confidential Proprietary Information of the Receiving Party in its possession as of the effective date of expiration or termination (with the exception of one (1) copy of such Confidential Proprietary Information, which may be retained by the legal department of the Receiving Party to confirm compliance with the non-use and non-disclosure provisions of this Agreement), and any Confidential Proprietary Information of the Disclosing Party contained in its laboratory notebooks or databases, *provided* that each Receiving Party may retain and continue to use such Confidential Proprietary Information of the Disclosing Party only to the extent necessary to exercise any surviving rights, licenses or obligations under this Agreement. Notwithstanding the foregoing, a Receiving Party shall not be required to destroy any computer files created during automatic system back up that are subsequently stored securely by it and not readily accessible to its Representatives who received the Disclosing Party's Confidential Proprietary Information under this Agreement, and neither Party shall be required to destroy any Joint IP.

14.4Terminated Existing Product Reversion.

14.4.1In the event of any termination of this Agreement or termination of this Agreement with respect to the Existing Product pursuant to Section 14.2.2(b), upon Precision's request, Imugene shall perform the following obligations, and Precision shall reimburse Imugene for the actual, reasonable costs associated with the performance of such obligations:

(a) to the extent permitted by Applicable Laws or the terms of any applicable Third Party agreements (including Third Party agreements under which Imugene or any of its Affiliates are granted a license related to the Exploitation of any Terminated Existing Product), (i) assign to Precision (A) Imugene's and its Affiliates' entire right, title and interest in and to all materials, preclinical and clinical data, safety data and all other supporting data, in each case, relating to such Terminated Existing Product that is in Imugene's or its Affiliates' Control, and (B) Imugene's and its Affiliates' entire right, title and interest in and to all such Third Party agreements that are freely assignable and relate to the Exploitation of any applicable Terminated Existing Product and for which such Third Party agrees to release Imugene for obligations and liabilities arising from and after such assignment, *provided*, that Imugene will retain the right to use any of the assigned materials or data as necessary for legal or compliance purposes, (ii) with respect to any Third Party agreements that are not assigned under (i) and under which Imugene or any of its Affiliates are granted a license related to Exploitation of any Terminated Existing Product and pursuant to which Imugene or its Affiliates have a right or ability to grant sublicenses to Precision, grant a sublicense to Precision of all license rights granted to Imugene thereunder, on and subject to the same terms and conditions (including financial terms) set forth in the applicable Third Party agreement solely to Exploit such Terminated Existing Product in all fields in the Territory, and (iii) deliver to Precision a copy of all relevant Know-How, in each case that relates to, and to the extent necessary or reasonably useful for, Precision to continue the Exploitation of such Terminated Existing Product;

(b) to the extent permitted by Applicable Laws and the terms of any applicable Third Party agreements, transfer to Precision ongoing Clinical Trials or other studies being conducted by or under authority of Imugene related to such Terminated Existing Product as of the date of the applicable termination notice and furnish Precision with reasonable cooperation

to transition to Precision the management and continued performance of such Clinical Trials or other studies or, if requested by Precision, terminate such Clinical Trials or other studies, in each case in a manner in compliance with Applicable Laws and ethical guidelines;

(c) to the extent permitted by Applicable Laws and the terms of any applicable Third Party agreements, transfer to Precision any and all Regulatory Filings and related regulatory data (including pharmacovigilance databases, adverse drug experience reports and associated documents) and nonclinical, clinical and other data contained or referenced in or supporting any Regulatory Filings and related Know-How, manufacturing records, Regulatory Approvals, Marketing Authorizations and all other correspondence (including minutes and official contact reports relating to any communications with any Regulatory Authority), filings and submissions with and to Regulatory Authorities with respect to such Terminated Existing Product; and, to this end, Imugene shall file for transfer with the relevant Regulatory Authorities and to give all other notifications and approvals necessary under Applicable Laws for the transfer of such Regulatory Filings and related regulatory data and Know-How, Regulatory Approvals, Marketing Authorizations and such other filings and submissions;

(d) after fulfillment of Imugene's existing commitments to its customers (including its Distributors) (which fulfillment period shall not in any event exceed [***] following termination of this Agreement as set forth in Section 14.3.1), sell to Precision Imugene's then-existing inventory of such Terminated Existing Product, at Imugene's cost of goods sold for such Terminated Existing Product as calculated in accordance with U.S. GAAP without mark-up; *provided* that Precision shall not be obligated to purchase such inventory;

(e) if an application seeking Marketing Authorization for a given Terminated Existing Product has been filed as of the effective date of termination of this Agreement, assign to Precision all right, title and interest in and to the Licensed Product Trademarks that have been used in commerce solely with such Terminated Existing Product, together with all goodwill relevant thereto, throughout the Territory; provided, however, that such obligation to assign will not extend to (i) any corporate name or logo of Imugene or any of its Affiliates, or (ii) any trademarks used by Imugene or any of its Affiliates on products that are not a Terminated Existing Product;

(f) Imugene shall not withdraw or cancel any such Terminated Existing Product's Regulatory Approval or Marketing Authorization or application for either, unless expressly instructed so by Precision in writing or required by Applicable Laws or any Regulatory Authority; *provided* that Precision shall be responsible for all costs and expenses for the maintenance of all Regulatory Approvals and Marketing Authorizations following receipt of notice of termination;

(g) Imugene shall thereafter refrain from making any statement, public or otherwise, regarding any Terminated Existing Product unless Imugene is required to make such statement pursuant to Applicable Law or requirements of any Regulatory Authority and such statement is limited to the fact that Imugene is no longer Developing or Commercializing such Terminated Existing Product or Precision shall have approved any such statement in writing; and

(h) following written request by Precision, Imugene shall take such other actions and execute such other instruments, assignments and documents that are reasonably necessary to effect the transfers and grants of rights under this Section 14.4.1 to Precision.

Following the foregoing assignments and transfer, all information and Know-How so assigned or transferred that was previously Confidential Proprietary Information of Imugene shall thereafter be deemed the Confidential Proprietary Information of Precision under Article 13.

14.4.2**Other Rights and Obligations**. Upon any termination of this Agreement, all other rights granted under this Agreement and all obligations of the Parties will automatically terminate except as expressly set forth in Section 14.3, this Section 14.4 or Section 14.5.

14.5Survival. Expiration or termination of this Agreement shall not relieve the Parties of any obligation or right accruing prior to such expiration or termination. Except as set forth below or elsewhere in this Agreement, the obligations and rights of the Parties under the following provisions of this Agreement shall survive expiration or termination of this Agreement: Articles 1 (to the extent such definitions are used in surviving provisions), 5 (only upon expiration, and not termination, of this Agreement) and 15 and Sections 4.1.6 (last sentence only), 4.2.2, 6.1.4 (only upon expiration, and not termination, of this Agreement), 6.4, 8.5 (first sentence only), 9.1 (with respect to Milestone Events achieved prior to the effective date of such termination or expiration), 9.2 (with respect to Milestone Events achieved prior to the effective date of such termination or expiration), 9.3 (with respect to sales of Licensed Products made before the effective date of such termination or expiration or pursuant to Section 14.3.1), 9.4 (with respect to any Change of Control or Product Partnering Transaction that occurred prior to the effective date of such termination or expiration), 9.5, 9.6, 9.7, 9.8, 9.9, 10.1.1, 10.1.2, 10.1.3, 10.2.1, 10.2.2, 10.3.2 (with respect to any and all Infringements of Joint IP), 10.3.5 (with respect to actions brought before the effective date of such termination or expiration, or brought with respect to Joint IP after the effective date of such termination or expiration), 10.3.6 (with respect to actions brought with respect to Joint IP), 10.4 (with respect to Joint IP), 10.6, 10.8 (final sentence only), 11.7, 12.1 (with respect to claims for which the cause of action arose prior to the effective date of termination or expiration), 13.1 (to the extent and as described in Section 13.1.6), 14.3, 14.4, 14.5, 14.6, 16.1, 16.2, 16.4, 16.5, 16.6, 16.8, 16.10, 16.14, 16.15, and 16.16.

14.6Exercise of Rights to Terminate; Damages; Relief. The valid use by either Party of a termination right provided for under this Agreement shall not give rise to the payment of damages or any other form of compensation or relief to the other Party with respect thereto; *provided*, however, that termination of this Agreement shall not preclude either Party from claiming any other damages, compensation or relief that it may be entitled to upon termination.

14.7**Bankruptcy Code**. If this Agreement is rejected by a Party as a debtor under Section 365 of the United States Bankruptcy Code or similar provision in the bankruptcy laws of another jurisdiction (the "*Code*"), then, notwithstanding anything else in this Agreement to the contrary, all licenses and rights to licenses granted under or pursuant to this Agreement by the Party in bankruptcy to the other Party are, and shall otherwise be deemed to be, for purposes of Section 365(n) of the Code (or similar provision in the bankruptcy laws of the jurisdiction), licenses of rights to "intellectual property" as defined under Section 101(35A) of the Code (or similar provision in the bankruptcy laws of another applicable jurisdiction). The Parties agree that

a Party that is a licensee of rights under this Agreement shall retain and may fully exercise all of its rights and elections under the Code, and that upon commencement of a bankruptcy proceeding by or against a Party under the Code, the other Party shall be entitled to a complete duplicate of, or complete access to (as such other Party deems appropriate), any such intellectual property to which such other Party is otherwise entitled to have access under this Agreement and all embodiments of such intellectual property, if not already in such other Party's possession, shall be promptly delivered to such other Party: (a) upon any such commencement of a bankruptcy proceeding upon written request therefor by such other Party, unless the bankrupt Party elects to continue to perform all of its obligations under this Agreement; or (b) if not delivered under the foregoing subclause (a), upon the rejection of this Agreement by or on behalf of the bankrupt Party upon written request therefor by the other Party. The foregoing provisions of this Section 14.7 are without prejudice to any rights a Party may have arising under the Code.

ARTICLE 15

GOVERNING LAW; DISPUTE RESOLUTION

15.1**Governing Law**. This Agreement shall be interpreted and construed in accordance with the laws of the State of New York. Any and all claims, controversies, and causes of action arising out of or relating to this Agreement, whether sounding in contract, tort, or statute, shall be governed by the laws of the State of New York, including its statutes of limitations, without giving effect to any conflict-of-laws or other rule that would result in the application of the laws of a different jurisdiction. Notwithstanding the foregoing, any issue relating to the interpretation, construction, validity, enforceability or infringement of Patents shall be determined according to the patent laws of the country (or countries) in which the relevant Patent (or Patents) issued. The United Nations Convention of International Contracts on the Sale of Goods (the Vienna Convention) does not apply to this Agreement.

15.2**Disputes**. The Parties recognize that controversies or claims arising out of, relating to, or in connection with this Agreement may arise from time to time. It is the objective of the Parties to establish procedures to facilitate the resolution of disputes in an expedient manner by mutual cooperation prior to resort to litigation. To accomplish this objective, the Parties shall follow the procedures set forth in this Article 15 to resolve any dispute, claim or controversy of any nature arising out of or relating to this Agreement, including any action or claim based on tort, contract or statute, or concerning the interpretation, effect, termination, validity, performance or breach of this Agreement (each, a "*Dispute*"). For the avoidance of doubt, Disputes within the purview of the JSC shall be resolved pursuant to Section 2.5, including through the exercise by a Party of its final decision-making authority in accordance therewith and including the escalation procedures set forth therein; *provided* that Disputes regarding whether a decision is subject to Precision's JSC representatives having final decision-making authority or to Imugene's JSC representatives having final decision-making authority pursuant to Section 2.5 shall be resolved pursuant to the procedures set forth in this Article 15.

15.3**Executive Officers**. If a Dispute arises between the Parties, either Party may refer the Dispute to Executive Officers of each Party for resolution within [***] of a written request by either Party to the other Party. Each Party, within [***] after a Party has received such written request from the other Party to so refer such Dispute, shall notify the other Party in writing of the

Executive Officer to whom such Dispute is referred. If, after an additional [***] after the notice of Dispute, such Executive Officers have not succeeded in negotiating a resolution of the Dispute, and a Party wishes to pursue the matter, the Parties may seek to resolve the Dispute in accordance with Section 15.4.

15.4**Submission to Jurisdiction**. Each Party hereby (i) submits to the exclusive jurisdiction of the United States District Court for the Southern District of New York or, if such court does not have jurisdiction, any state court sitting in the City of New York, New York in any action or proceeding arising out of or relating to this Agreement, (ii) agrees that all claims in respect of such action or proceeding may be heard and determined only in any such court, and (iii) agrees not to bring any action or proceeding arising out of or relating to this Agreement in any other court. Each Party waives any defense of inconvenient forum to the maintenance of any action or proceeding so brought and waives any bond, surety or other security that might be required of the other Party with respect thereto. Either Party may make service on the other Party by sending or delivering a copy of the process to the Party to be served at the address and in the manner provided for the giving of notices in Section 16.4. Nothing in this Section 15.4, however, shall affect the right of either Party to serve legal process in any other manner permitted by law.

15.5 Waiver of Jury Trial. TO THE EXTENT PERMITTED BY APPLICABLE LAW, EACH PARTY HEREBY IRREVOCABLY WAIVES ALL RIGHTS TO TRIAL BY JURY IN ANY ACTION, PROCEEDING OR COUNTERCLAIM (WHETHER BASED ON CONTRACT, TORT OR OTHERWISE) ARISING OUT OF OR RELATING TO THIS AGREEMENT OR THE TRANSACTIONS OR THE ACTIONS OF EITHER PARTY IN THE NEGOTIATION, ADMINISTRATION, PERFORMANCE AND ENFORCEMENT OF THIS AGREEMENT.

15.6**Equitable Relief**. Either Party may, at any time and without waiving any remedy under this Agreement, seek from any court having jurisdiction any temporary injunctive or provisional relief necessary to protect the rights or property of that Party. Any final judgment resolving a Dispute may be enforced by either Party in any court having appropriate jurisdiction.

ARTICLE 16

MISCELLANEOUS

16.1**Entire Agreement; Amendment**. This Agreement, including the Exhibits and Schedules hereto, together with the other Transaction Documents, sets forth the complete, final and exclusive agreement and all the covenants, promises, agreements, warranties, representations, conditions and understandings between the Parties hereto with respect to the subject matter hereof and supersedes, as of the Effective Date, all prior and contemporaneous agreements and understandings between the Parties with respect to the subject matter hereof, including the Confidentiality Agreement. The foregoing may not be interpreted as a waiver of any remedies available to either Party as a result of any breach, prior to the Effective Date, by the other Party of its obligations under the Confidentiality Agreement. No subsequent alteration, amendment, change or addition to this Agreement shall be binding upon the Parties unless reduced to writing and signed by an authorized officer of each Party.

16.2 Limitation of Liability. NEITHER PARTY MAY RECOVER FROM THE OTHER PARTY ANY SPECIAL, INCIDENTAL, INDIRECT, CONSEQUENTIAL OR PUNITIVE DAMAGES IN CONNECTION WITH THIS AGREEMENT, REGARDLESS OF ANY NOTICE OF THE POSSIBILITY OF SUCH DAMAGES; PROVIDED, HOWEVER, THAT THIS SECTION 16.2 SHALL NOT BE CONSTRUED TO LIMIT EITHER PARTY'S INDEMNIFICATION OBLIGATIONS UNDER ARTICLE 12, EITHER PARTY'S LIABILITY FOR BREACH OF ITS CONFIDENTIALITY OBLIGATIONS UNDER ARTICLE 13 OR LIABILITY OF A PARTY FOR ITS INFRINGEMENT OR MISAPPROPRIATION OF ANY INTELLECTUAL PROPERTY RIGHTS OR FOR A PARTY'S GROSS NEGLIGENCE, WILLFUL MISCONDUCT OR FRAUD.

16.3**Independent Contractors**. The relationship between Imugene and Precision created by this Agreement is solely that of independent contractors. This Agreement does not create any agency, distributorship, employee-employer, partnership, joint venture or similar business relationship between the Parties. Neither Party is a legal representative of the other Party, and neither Party can assume or create any obligation, representation, warranty, or guarantee, express or implied, on behalf of the other Party.

16.4**Notice**. Any notice required or permitted to be given by this Agreement must be in writing, in English. Any and all notices or other communications or deliveries required or permitted to be provided hereunder must be in writing and will be deemed given and effective if: (a) delivered by hand or by overnight courier with tracking capabilities; (b) mailed postage prepaid by first class, registered, or certified mail; or (c) delivered by facsimile or electronic mail followed by delivery via either of the methods set forth in clauses (a) and (b) of this Section 16.4, in each case, addressed as set forth below unless changed by notice so given:

If to Precision: Precision BioSciences, Inc.

302 East Pettigrew Street, Suite A-100 Durham, NC 27701, U.S.A. Attn: [***], Chief Business Officer E-mail: [***]

with a copy (which shall not constitute notice) to:

Smith, Anderson, Blount, Dorsett, Mitchell & Jernigan, LLP 150 Fayetteville Street, Suite 2300 Raleigh, NC 27601, U.S.A. Attention: John Therien

If to Imugene: Imugene (USA) Inc.

Suite 200 701 South Carson St Carson City, NV 89701, U.S.A. Attention: [***], CEO & MD Email: [***] with a copy (which shall not constitute notice) to:

Locke Lord LLP 2800 Financial Plaza Providence, RI 02903 Attention: Douglas G. Gray, Esq.

Email: [***]

16.5**Severability**. If, for any reason, any part of this Agreement is adjudicated invalid, unenforceable, or illegal by a court of competent jurisdiction, (a) such adjudication shall not, to the extent feasible, affect or impair, in whole or in part, the validity, enforceability, or legality of any remaining portions of this Agreement, (b) this Agreement shall be construed and enforced as if such invalid, unenforceable or illegal provision had never comprised a part hereof, (c) all remaining portions will remain in full force and effect and shall not be affected by the invalid, unenforceable or illegal provision or by its severance herefrom, and (d) in lieu of such invalid, unenforceable or illegal provision, the Parties shall use reasonable efforts to seek and agree on an alternative valid and enforceable provision that preserves the original purpose and intent of this Agreement.

16.6Non-Use of Names. Except as permitted pursuant to Section 13.2, Precision shall not use the name, trademark, logo, or physical likeness of Imugene or its respective officers, directors or employees, or any adaptation of any of them, in any advertising, promotional or sales literature, without Imugene's prior written consent; *provided* that Precision shall have the right to use the name and logo of Imugene on its website solely for the purpose of referring to Imugene as a partner of Precision. Precision shall require its Affiliates to comply with the foregoing. Except as permitted pursuant to Section 13.2, Imugene shall not use the name, trademark, logo, or physical likeness of Precision or its officers, directors or employees, or any adaptation of any of them, in any advertising, promotional or sales literature, without Precision's prior written consent; *provided* that Imugene shall have the right to use the name and logo of Precision on its website and in presentation materials solely for the purpose of referring to Precision as licensor of technology used by Imugene. Imugene shall require its Affiliates and Sublicensees to comply with the obligations set forth in this Section 16.6.

16.7**Assignment**. Neither Party may assign or transfer this Agreement or any rights or obligations hereunder without the prior written consent of the other, except that a Party may make such an assignment or transfer, without the other Party's consent to: (a) its Affiliate *provided* that (i) such Affiliate has sufficient resources to perform under this Agreement and (ii) such Party shall remain primarily liable for any acts or omissions of such Affiliate; or (b) to an Acquirer in connection with a Change of Control of such Party. For the avoidance of doubt, nothing in this Agreement shall be construed as consent by Precision to assignment of this Agreement by Imugene in the context of a bankruptcy proceeding. Any permitted assignee shall, in writing reasonably satisfactory to the non-assigning party and as a condition to the effectiveness of such assignment, expressly assume performance of such assignment of this Agreement by Imugene, Imugene also shall require the permitted assignee, in writing reasonably satisfactory to Precision and as a condition to the effectiveness of such assignment, to expressly assume performance of Imugene's and Parent's rights and obligations under the other Transaction

Documents and unconditionally agree to the terms of the other Transaction Documents. Any permitted assignment or transfer is binding on the successors of the assigning or transferring Party and shall inure to their benefit. Any assignment or transfer or attempted or purported assignment or transfer by either Party in violation of the terms of this Section 16.7 is null, void and of no legal effect.

16.8 **Waivers**. The failure of a Party to insist upon strict performance of any provision of this Agreement or to exercise any right arising out of this Agreement shall neither impair that provision or right nor constitute a waiver of that provision or right, in whole or in part, in that instance or in any other instance. Any waiver by a Party of a particular provision or right shall be in writing, shall be as to a particular matter and, if applicable, for a particular period of time and shall be signed by such Party.

16.9Force Majeure. Neither Party shall be responsible to the other for any failure or delay in performing any of its obligations under this Agreement or for other nonperformance hereunder (excluding, in each case, the obligation to make payments when due) if such delay or nonperformance is caused by strike, fire, flood, earthquake, accident, war, act of terrorism, epidemics, pandemics, the spread of infectious diseases, quarantines, act of God or of the government of any country or of any local government, or by any other cause unavoidable or beyond the control of any Party hereto. In such event, such affected Party shall use Commercially Reasonable Efforts to resume performance of its obligations and will keep the other Party informed of actions related thereto.

16.10**Interpretation**. The captions and headings to this Agreement are for convenience only, and are to be of no force or effect in construing or interpreting any of the provisions of this Agreement. Unless specified to the contrary, references to Articles, Sections, Appendices or Exhibits mean the particular Articles, Sections, Appendices or Exhibits to this Agreement and references to this Agreement include all Exhibits hereto. In the event of any conflict between the main body of this Agreement and any Exhibit hereto, the main body of this Agreement shall prevail. Unless context otherwise clearly requires, whenever used in this Agreement: (a) the words "include" or "including" shall be construed as incorporating, also, "but not limited to" or "without limitation"; (b) the word "day" or "year" means a calendar day or Calendar Year unless otherwise specified; (c) the word "notice" means notice in writing (whether or not specifically stated) and shall include notices, consents, approvals and other written communications contemplated under this Agreement; (d) the words "hereof," "herein," "hereby" and derivative or similar words refer to this Agreement as a whole and not merely to the particular provision in which such words appear; (e) the words "shall" and "will" have interchangeable meanings for purposes of this Agreement; (f) provisions that require that a Party, the Parties or a committee hereunder "agree," "consent" or "approve" or the like shall require that such agreement, consent or approval be specific and in writing, whether by written agreement, letter, approved minutes or otherwise; (g) words of any gender include the other gender; (h) words using the singular or plural number also include the plural or singular number, respectively; (i) references to any specific law, rule or regulation, or article, section or other division thereof, shall be deemed to include the then-current amendments thereto or any replacement law, rule or regulation thereof; (j) the phrase "non-refundable" shall not prohibit, limit or restrict either Party's right to obtain damages in connection with a breach of this Agreement; (k) neither Party shall be deemed to be acting on behalf of the

other Party; and (l) the words "gene editing" and "genome editing" have interchangeable meanings for purposes of this Agreement and do not include gene therapy activities (other than gene editing).

- 16.11**Counterparts; Electronic Signatures**. This Agreement may be executed in any number of counterparts, each of which is deemed an original, but all of which together constitute one instrument. This Agreement may be executed and delivered electronically and upon such delivery such electronic signature will be deemed to have the same effect as if the original signature had been delivered to the other Party.
- 16.12**Expenses**. Each Party shall pay its own costs, charges and expenses incurred in connection with the negotiation, preparation and execution of this Agreement.
- 16.13**Further Assurances**. Imugene and Precision hereby covenant and agree without the necessity of any further consideration, to execute, acknowledge and deliver any and all documents and take any action as may be reasonably necessary to carry out the intent and purposes of this Agreement.
- 16.14**No Third Party Beneficiary Rights**. This Agreement is not intended to and shall not be construed to give any Third Party any interest or rights (including any Third Party beneficiary rights) with respect to or in connection with any agreement or provision contained herein or contemplated hereby.
- 16.15**Construction**. The Parties hereto acknowledge and agree that: (a) each Party and its counsel reviewed and negotiated the terms and provisions of this Agreement and have contributed to its revision; (b) the rule of construction to the effect that any ambiguities are resolved against the drafting Party shall not be employed in the interpretation of this Agreement; and (c) the terms and provisions of this Agreement shall be construed fairly as to all Parties hereto and not in a favor of or against any Party, regardless of which Party was generally responsible for the preparation of this Agreement.
- 16.16**Cumulative Remedies**. No remedy referred to in this Agreement is intended to be exclusive unless explicitly stated to be so, but each shall be cumulative and in addition to any other remedy referred to in this Agreement or otherwise available under law.
- 16.17**Extension to Affiliates**. Except as expressly set forth otherwise in this Agreement, each Party shall have the right to extend the rights and immunities granted in this Agreement to one or more of its Affiliates. All applicable terms and provisions of this Agreement, except this right to extend, shall apply to any such Affiliate to which this Agreement has been extended to the same extent as such terms and provisions apply to the Party extending such rights and immunities. For clarity, the Party extending the rights and immunities granted hereunder shall remain primarily liable for any acts or omissions of its Affiliates.

[signature page follows]

In Witness Whereof, the Parties have caused this Agreement to be executed as of the Effective Date by their duly authorized representatives.
Precision BioSciences, Inc.
Dru /a/ Michael Amorese
By: <u>/s/ Michael Amoroso</u> Name: Michael Amoroso

IMUGENE (USA) INC.

Title: Chief Executive Officer

By: <u>/s/ Paul Hopper</u>
Name: Paul Hopper
Title: President

[Signature Page to License Agreement]

Exhibit 1.100

Licensed ARCUS Nuclease

[Omitted]

Exhibit 1.100 - 1

Exhibit 8.6.2

Exhibit to Cellectis Agreement

[Omitted]

Exhibit 8.6.2 - 1

Exhibit 11.2.2

Existing Patents

[Omitted]

Exhibit 11.2.2 - 1

Schedule 1.10

ARCUS Research Tool

[Omitted]

Schedule 1.10 - 1

Schedule 1.52

Patents within Duke IP as of the Effective Date

[Omitted]

Schedule 1.52 - 1

Schedule 1.114

Ongoing Precision Trial

[Omitted] Schedule 1.114 - 1

Unaudited Pro Forma Condensed Consolidated Financial Statements

On August 15, 2023, Precision BioSciences, Inc. (the "Company") entered into an Asset Purchase Agreement (the "Purchase Agreement") with Imugene Limited, an Australian corporation ("Imugene Limited"), and its wholly owned subsidiary Imugene (USA) Inc. ("Imugene US"), a Nevada corporation (collectively, "Imugene"). Pursuant to and simultaneously with the execution of the Purchase Agreement, on August 15, 2023 (the "Closing Date"), Imugene US acquired the manufacturing infrastructure to the Company's CAR T program that uses azer-cel for the potential treatment of relapsed or refractory non-Hodgkin lymphoma, which includes the Company's MCAT facility, certain contracts of the Company with respect to the MCAT facility and the Company's azer-cel program, including the lease to the MCAT facility, and related equipment, supplies, azer-cel clinical trial inventory and other assets related to the Company's CAR T cell therapy platform (the "Acquisition").

In consideration for the Acquisition, Imugene US assumed certain liabilities of the Company, paid the Company \$8 million in cash, and issued to the Company convertible notes in an aggregate principal amount of \$13 million (the "Convertible Notes"). The Convertible Notes are non-interest bearing and mature on the first anniversary of the Closing Date. On the maturity date, the Convertible Notes generally must be redeemed with cash, converted into ordinary shares of Imugene Limited at a conversion price based on the volume weighted average price of Imugene Limited's ordinary shares prior to the date of conversion, or partially redeemed with cash and partially converted into shares.

In connection with the Purchase Agreement, on the Closing Date, the Company and Imugene US entered into a License Agreement (the "License Agreement"), pursuant to which the Company granted Imugene US certain exclusive and non-exclusive license rights to develop, manufacture, and commercialize oncological applications of the Company's allogeneic CAR T therapy azer-cel and up to three additional research product candidates directed to targets that Imugene US may nominate prior to the fifth anniversary of the effective date of the License Agreement, pursuant to the terms of the License Agreement.

In connection with the Purchase Agreement and the License Agreement, the Company and Imugene entered into other related agreements and documents, including a registration rights agreement, a transition services agreement, a sublease for laboratory space at the Company's headquarters and a parent company guaranty from Imugene Limited.

The following unaudited pro forma condensed consolidated financial statements are based on the Company's historical consolidated financial statements and are presented to illustrate:

- the estimated effects of the Acquisition; and
- the unaudited pro forma condensed consolidated statements of operations of the Company reflecting the Company's results of operations as if the Acquisition had occurred on January 1, 2021. The unaudited pro forma condensed balance sheet of the Company reflects the Company's financial position as if the Acquisition had occurred on June 30, 2023. Beginning in the third quarter of 2023, the Company's CAR T cell therapy historical financial results for periods prior to the Acquisition will be reflected in the Company's consolidated financial statements as discontinued operations.

The unaudited pro forma condensed consolidated financial statements and the accompanying notes should be read in conjunction with:

• the audited consolidated financial statements and accompanying notes and "Management's Discussion and Analysis of Financial Condition and Results of Operations" included in the Company's Form 10-K for the year ended December 31, 2022; and

• the unaudited condensed financial statements and accompanying notes and "Management's Discussion and Analysis of Financial Condition and Results of Operations" included in the Company's Form 10-Q for the six months ended June 30, 2023.

The "Precision BioSciences as Reported" column in the unaudited pro forma condensed consolidated financial statements reflects the Company's historical financial statements for the periods presented and does not reflect any adjustments related to the Acquisition.

The information in the "Discontinued Operations" column in the unaudited pro forma condensed consolidated balance sheet was derived from the Company's consolidated financial statements and the related accounting records as of June 30, 2023, adjusted to include certain assets and liabilities that have been transferred to Imugene US pursuant to the Purchase Agreement.

The Company believes that the adjustments included within the "Discontinued Operations" column are consistent with the guidance for discontinued operations under accounting principles generally accepted in the United States. The Company's current estimates on a discontinued operations basis are preliminary and could change as it finalizes discontinued operations accounting to be reported in the Quarterly Report on Form 10-Q for the nine months ending September 30, 2023.

The unaudited pro forma condensed consolidated financial statements have been prepared by management in accordance with Article 11, *Pro Forma Financial Information*, under Regulation S-X and are for illustrative and informational purposes only. The pro forma financial statements are based on various adjustments and assumptions and are not necessarily indicative of what the Company's consolidated financial statements actually would have been or will be for any future periods had the Acquisition been completed as of the dates indicated. These estimates and assumptions are based on information currently available to management and do not reflect the final determinations by management of the impact of the Acquisition on the Company's consolidated financial statements.

The pro forma condensed consolidated financial statements do not purport to project the future financial position or operating results of the Company following the completion of the Acquisition. The pro forma condensed consolidated financial statements do not include adjustments to reflect any potential synergies or dis-synergies that may result from the Acquisition. The actual effects of the Acquisition will differ from the pro forma adjustments.

Unaudited Pro Forma Condensed Statement of Operations Six Months Ended June 30, 2023 (in thousands, except per share data)

	Precision BioSciences, Inc. As Reported	Discontinued Operations (A)	Subtotal	Transaction Accounting Adjustments	Notes	Pro Forma
Revenue	\$ 28,569	_	\$ 28,569	_		\$ 28,569
Operating expenses				_		
Research and development	44,104	(15,012)	29,092	_		29,092
General and administrative	20,916	(137)	20,779	_		20,779
Total operating expenses	65,020	(15,149)	49,871	_		49,871
Operating loss	(36,451)	15,149	(21,302)	_		(21,302)
Other income (expense):						
Loss from equity method investment	(2,710)	_	(2,710)	_		(2,710)
Changes in fair value	(769)	_	(769)	_		(769)
Interest expense	(1,075)	_	(1,075)	_		(1,075)
Interest income	3,989	_	3,989	_		3,989
Gain on disposal of assets	65	_	65	_		65
Total other expense	(500)	-	(500)	-		(500)
Net loss	\$ (36,951)	\$ 15,149	\$ (21,802)	\$ -		\$ (21,802)
Net loss per share - basic and diluted	(0.33)					(0.19)
Weighted-average number of common shares used in computing net loss per share - basic and diluted	112,708,231					 112,708,231

Unaudited Pro Forma Condensed Consolidated Statement of Operations Twelve Months Ended December 31, 2022 (in thousands, except per share data)

	Precision BioSciences, Inc. As Reported	Discontinued Operations (A)		Subtotal	Transaction Accounting Adjustments	Notes]	Pro Forma
Revenue	\$ 25,098		\$	25,098			\$	25,098
Operating expenses								
Research and development	83,939	(29,907))	54,032	_	(B)		54,032
General and administrative	41,525	(376)		41,149	300	(C)		41,449
Total operating expenses	125,464	(30,283)		95,181	300			95,481
Operating loss	(100,366)	30,283		(70,083)	(300)			(70,383)
Other income (expense):								
Impairment charges	(11,438)	<u> </u>		(11,438)	_			(11,438)
Loss on disposal of assets	(106)	—		(106)	_			(106)
Changes in fair value	(510)	<u> </u>		(510)				(510)
Loss from equity method investment	(1,579)	—		(1,579)	_			(1,579)
Interest expense	(1,111)	—		(1,111)				(1,111)
Interest income	3,473			3,473	703	(D)		4,176
Total other expense	(11,271)	—		(11,271)	703			(10,568)
Net loss	\$ (111,637)	\$ 30,283	\$	(81,354)	\$ 403		\$	(80,951)
Net loss per share - basic and diluted	(1.27))						(0.92)
Weighted-average number of common shares used in computing net loss per share - basic and diluted	87,898,498							87,898,498

Unaudited Pro Forma Condensed Consolidated Statement of Operations Twelve Months Ended December 31, 2021 (in thousands, except per share data)

	Prec	rision BioSciences, Inc. As Reported	Discontinued Operations (A)	Pro Forma
Revenue	\$	115,529	(72,851)	\$ 42,678
Operating expenses				
Research and development		115,238	(27,244)	87,994
General and administrative		39,667	(21)	 39,646
Total operating expenses		154,905	(27,265)	127,640
Operating loss		(39,376)	(45,586)	(84,962)
Other income (expense):				
Loss on disposal of assets		(26)	_	(26)
Changes in fair value		2,555	_	2,555
Gain on deconsolidation of subsidiary		5,985		5,985
Income from equity method investment		184	_	184
Interest expense		(132)	_	(132)
Interest income		208		208
Total other income		8,774		 8,774
Net loss	\$	(30,602)	\$ (45,586)	\$ (76,188)
Net loss per share - basic and diluted		(0.52)		 (1.30)
Weighted-average number of common shares used in computing net loss per share - basic and diluted		58,688,102		58,688,102

Unaudited Pro Forma Condensed Balance Sheet As of June 30, 2023 (in thousands)

	Precision BioSciences, Inc. As Reported		Discontinued Operations (A)	Subtotal		Transaction Accounting Adjustments	Notes	Pro Forma	
Assets									
Current assets:									
Cash and cash equivalents	\$	137,794	_	\$	137,794	\$ 8,304	(E)	\$ 146,098	
Accounts receivable		652	_		652	_		652	
Prepaid expenses		8,915	(569)		8,346	_		8,346	
Other current assets		503	_		503	_		503	
Total current assets		147,864	(569)		147,295	8,304		155,599	
Property, equipment, and software—net		18,249	(11,750)		6,499	_		6,499	
Intangible assets—net		1,552	(258)		1,294	_		1,294	
Right-of-use assets—net		4,721	(3,341)		1,380	_		1,380	
Investments in equity securities		1,807	_		1,807	_		1,807	
Equity method investment		_	_		_	_		_	
Convertible note receivable		_	_		_	12,043	(F)	12,043	
Note receivable—net		6,885	_		6,885	_		6,885	
Other assets		619	(304)		315	_		315	
Total assets	\$	181,697	\$ (16,222)	\$	165,475	\$ 20,347		\$ 185,822	

Unaudited Pro Forma Condensed Balance Sheet As of June 30, 2023 (in thousands)

	Precision BioSciences, Inc. As Reported	Discontinued Operations (A)	Subtotal	Transaction Accounting Adjustments	Notes	Pro Forma
Liabilities and Stockholders' Equity						
Current liabilities:						
Accounts payable	\$ 781	_	\$ 781	_	S	781
Accrued compensation	3,500	(718)	2,782	_		2,782
Accrued clinical and research and development						
expenses	2,280	(1,434)	846	_		846
Deferred revenue	23,811	_	23,811	_		23,811
Lease liabilities	2,499	(727)	1,772	_		1,772
Loan payable—net	22,317	_	22,317	_		22,317
Other current liabilities	1,211	_	1,211	_		1,211
Total current liabilities	56,399	(2,879)	53,520	_	_	53,520
Deferred revenue	78,084	_	78,084	_	<u>-</u>	78,084
Lease liabilities	3,173	(3,026)	147	_		147
Contract liabilities	10,000	_	10,000	_		10,000
Total liabilities	147,656	(5,905)	141,751		_	141,751
Stockholders' equity:				_	_	
Preferred Stock	_	_	_	_		_
Common Stock	1	_	1	_		1
Additional paid-in capital	500,255	_	500,255	_		500,255
Accumulated deficit	(465,263)	(10,317)	(475,580)	20,347	(G)	(455,233)
Treasury stock	(952)	_	(952)	_		(952)
Total stockholders' equity	34,041	(10,317)	23,724	20,347		44,071
Total liabilities and stockholders' equity	\$ 181,697	\$ (16,222)	\$ 165,475	\$ 20,347	9	185,822

- (A) Reflects the discontinued operations of the Company's CAR T cell therapy platform transferred to Imugene US, including the associated assets, liabilities, equity and results of operations. The Company's current estimates on the discontinued operations basis are preliminary and could change as the Company finalizes the accounting for the discontinued operations to be reported in its Quarterly Report on Form 10-Q for the nine months ending September 30, 2023.
- (B) Certain payments are due to other third parties as a result of the Acquisition. These amounts have not been reported as transaction accounting adjustments in the unaudited pro forma condensed consolidated financial statements as the determination of these amounts is not practical at the time of filing of these unaudited pro forma condensed consolidated financial statements.
- (C) Reflects estimated legal, tax, and accounting fees incurred in connection with the Acquisition.
- (D) Reflects imputed interest income on the Convertible Notes during the twelve months ended December 31, 2022.
- (E) Reflects cash proceeds received from Imagene in connection with the Acquisition.
- (F) Reflects the pro forma fair value of the Convertible Notes as of June 30, 2023.
- (G) This adjustment represents the estimated gain of \$10 million arising from the Acquisition. This estimated gain has not been reflected in the unaudited pro forma unaudited condensed consolidated statements of operations as it relates to discontinued operations.